**Department for Work and Pensions** 

**Research Report No 569** 

# The impact of financial incentives in welfare systems on family structure

**Bruce Stafford and Simon Roberts** 

A report of research carried out by the International Centre for Public and Social Policy, The University of Nottingham on behalf of the Department for Work and Pensions

© Crown Copyright 2009. Published for the Department for Work and Pensions under licence from the Controller of Her Majesty's Stationery Office.

Application for reproduction should be made in writing to The Copyright Unit, Her Majesty's Stationery Office, St Clements House, 2-16 Colegate, Norwich NR3 1BQ.

First Published 2009.

ISBN 978 1 84712 511 8

Views expressed in this report are not necessarily those of the Department for Work and Pensions or any other Government Department.

# Contents

Αc	:knowl	edgemer	nts	vii	
Th	e Auth	ors		viii	
ΑŁ	brevia	tions		ix	
Su	mmary	/		1	
1	Introduction				
	1.1	Introduction			
	1.2	Review's objectives and scope			
	1.3	Conduct of the review			
	1.4	Background: UK and US family structures and welfare systems			
		1.4.1	Demographic changes affecting UK families	9	
		1.4.2	UK welfare system	14	
		1.4.3	Demographic changes affecting US families	15	
		1.4.4	US welfare system	17	
	1.5	Theoretical perspectives and empirical observations		21	
		1.5.1	Understanding welfare systems and family structure	22	
		1.5.2	Marriage and welfare systems	24	
		1.5.3	Cohabitation and welfare systems	28	
		1.5.4	Union dissolution and welfare systems	30	
		1.5.5	Fertility and single motherhood and welfare systems	32	

	1.6	Methodological issues			
		1.6.1	Non-experimental methods	35	
		1.6.2	Experimental methods	38	
		1.6.3	Quasi-experimental methods	40	
		1.6.4	Other methodological issues	40	
	1.7	Structu	re of report	41	
2	Partnering				
	2.1	Introduction4			
		2.1.1	Union/marriage bonuses and penalties	45	
	2.2	Research evidence on welfare systems and partnering48			
		2.2.1	UK evidence on union formation and the welfare system	m48	
		2.2.2	Cross-country comparative studies on union formation.	50	
		2.2.3	US evidence on union formation and the welfare system	m51	
		2.2.4	Canadian evidence on union formation and the welfare		
			system	67	
	2.3	Overvie	w	71	
3	Unior	n dissolut	tion	73	
	3.1	Introdu	ction	74	
	3.2	Researc	ch evidence on union dissolution	74	
		3.2.1	UK evidence on union dissolution and the welfare syste	em .74	
		3.2.2	US evidence on union dissolution and the welfare syste	m76	
		3.2.3	Australia evidence on union dissolution and the welfare	5	
			system	78	
	3.3	Overview			
4	Childbearing and single parents			81	
	4.1	Introduction			
	4.2	Research evidence			
		4.2.1	UK evidence on welfare systems and childbearing and		
			single parenthood	83	

		4	2.2	Cross-national evidence on welfare systems and	
				childbearing and single parenthood	85
		4	2.3	US evidence on welfare systems and childbearing and	
				single parenthood	89
	4.3	О١	erviev/	V	113
5	Implications for the UK welfare system			115	
	5.1	Int	troduc	tion	116
	5.2	Th	Theoretical and empirical ambiguity		
	5.3	Ро	Policy transfer		
	5.4	W	elfare	policy and the family	119
		5.4	4.1	Is there a case for marriage penalties?	119
		5.4	4.2	Reducing marriage penalties	121
		5.4	4.3	Policy options for reducing (non-marital) childbearing	123
		5.4	4.4	Welfare systems and modern living arrangements	125
	5.5	О١	erviev/	V	126
6	Conclusions				127
	6.1	Introduction			127
	6.2 F		Research implications		129
		6	2.1	Further research	130
		6	2.2	Undertaking research on welfare systems and family structure	130
	6.3	Fir	nal tho	oughts	
Ар	pendix	А	Non-	experimental methods of analysis	135
Ар	pendix			JK studies modelling the amount and distribution of union	
·	•			ses and penalties	139
Appendix				mary of Moffitt's (1998b) review of US	4.40
				experimental studies	143
Ар	pendix	D		udies modelling the amount and distribution of union ses and penalties	147
Re <sup>-</sup>	ference	·s			151

List of tak	oles	
Table C.1	Number of estimates of welfare effects on marriage and fertility review by Moffitt (1998b)	
List of fig	ures	
Figure 1.1	Heuristic model for how the welfare system might influence family structure	23

# Acknowledgements

The authors are grateful for the advice and support received from (past and current) colleagues within Department for Work and Pensions (DWP) notably Gillian Burgess, Bola Akinwale and Stephen Morris of the Strategy Unit.

Within the School of Sociology and Social Policy we would like to thank Emily Frisby.

Nonetheless, the views expressed in this report are the authors' own and do not necessarily reflect those of the DWP.

# The Authors

**Bruce Stafford** is Professor of Public Policy at the International Centre for Public and Social Policy, School of Sociology and Social Policy, University of Nottingham. Before joining the School in February 2006, he was a social and policy researcher in local government and at the universities of Birmingham, Loughborough and Salford.

His main academic interests are focused on policy evaluations, social security in particular welfare to work, disability issues, and the implementation of policies and the delivery of welfare services.

In recent years, the majority of his research has been for government departments, notably the Department for Work and Pensions (DWP).

**Simon Roberts** is a Special Lecturer at the International Centre for Public and Social Policy, School of Sociology and Social Policy, University of Nottingham.

Before joining the University of Nottingham he was a social and policy researcher at the universities of Brunel and Loughborough. His main research interests include United Kingdom (UK), European and international social security, disability and discrimination, and Housing Benefit reform. He is the UK expert and member of the think tank on the European Commission's Training and Reporting on European Social Security network.

# **Abbreviations**

AFDC Aid to Families with Dependent Children

AFDC-UP Aid to Families with Dependent Children-

**Unemployed Parent** 

BHPS British Household Panel Survey

**CPS** Current Population Survey

CTC Child Tax Credit

**EITC** Earned Income Tax Credit

**FES** Family Expenditure Survey

FTBA Family Tax Benefit Part A

JOBS Job Opportunities and Basic Skills

MFIP Minnesota Family Investment Program

NLSY National Longitudinal Survey of Youth

**OECD** Organisation for Economic Co-operation and

Development

PAYG Pay As You Go

PRWORA Personal Responsibility and Work

Reconciliation Act 1996

**PSID** Panel Study of Income Dynamics

SSP Self-Sufficiency Project

**TANF** Temporary Assistance for Needy Families

UK United Kingdom

US United States

WFTC Working Families Tax Credit

WTC Working Tax Credit

# Summary

#### Introduction

There are academic and policy debates about whether financial incentives in welfare (that is, benefits and tax credit) systems affect demographic behaviour. For some they are seen as leading to less marriage, and more cohabitation, non-marital births and single parents. This report reviews the evidence on the influence of financial incentives in the welfare system on union formation and dissolution and on childbearing. The review also highlights the implications of reported findings for the United Kingdom (UK).

Many countries, including the UK, have witnessed substantial changes to family lives and structures since the Second World War. In general, there has been less marriage with those marrying tending to do so later in life, more cohabitation, more same sex relationships, more divorce, childbearing has been delayed to later in life and there are more births outside of marriage. As a consequence, there has been a rapid increase in single parenthood.

The UK literature, which is very limited in quantity, tends to focus on the impact of the introduction of Working Families Tax Credit (WFTC) on family structure. Most of the literature is from the United States (US), where the focus is on the welfare effects of two means-tested programmes, Aid to Families with Dependent Children (AFDC) and its replacement, Temporary Assistance for Needy Families (TANF), as well as Earned Income Tax Credit (EITC).

Much of the literature is informed by economic theory, which sees the decision to marry, cohabit or have children as being influenced by financial incentives and disincentives. Benefits, tax credits and employment programmes can be conceived as influencing demographic behaviour indirectly through their effect on employment and family income. In addition, there are other (intervening) factors that affect family structure, such as, social norms and attitudes, race, educational attainment, self-esteem, age and so on. As a consequence, it can be difficult to predict whether the welfare system has a positive or a negative effect on family structure. For instance, work incentives in benefits and tax credits may lead to an increase in union formation as gaining work improves a woman's self-confidence,

self-esteem or widens her social network which then includes more potential partners. However, it may mean she has less time to search for a partner or to socialise and so leads to decline in marriage/cohabitation. Similarly, higher female earnings (from the welfare system and/or employment) may mean that she is more attractive as a potential partner or that she has acquired the financial independence to live on her own. The factors underpinning demographic behaviour are complex and the underlying mechanisms are not fully understood.

There are three broad methodological approaches used by analysts:

- non-experimental;
- experimental, and
- quasi-experimental.

In terms of the volume of evidence there are considerably more published analyses using non-experimental methods. Over time, researchers using non-experimental approaches have increased their degree of control over unobserved factors in their analyses, especially for factors operating at State and individual levels. With increasing levels of control for unobserved variables, the size of any welfare effect on demographic behaviour tends to diminish; indeed, what could be reported as statistically significant impacts can become insignificant. Social experiments help to address the problem of unobserved factors, but there are problems with being able to generalise beyond the locality where the demonstration project was conducted, and there are concerns about possible contamination of control groups in some studies. The main quasi-experimental method used is difference-in-difference, which involves comparing demographic outcomes for a group affected by a policy or programme with one unaffected by the intervention both before and after the introduction of the reform.

## Welfare systems and partnering

Some commentators express the concern that there may be a 'marriage penalty' (as opposed to a 'marriage bonus') in the benefit and tax credit system. A union/ marriage bonus (penalty) occurs when a couple receive more (less) in benefit/tax credit payments than they would if they were single. For union/marriage bonuses or penalties to exist there are two pre-conditions: the unit of assessment is the family (rather than the individual) and different marginal tax rates apply at different income levels. A benefit/tax system cannot be simultaneously neutral towards marriage, be progressive and treat families with the same income equally.

The evidence that welfare systems affect union formation is mixed. Where there is an effect it tends to be small or modest in size (typically, two to five percentage points). Moreover, the impacts appear to vary considerably by sub-group and possibly take some time before they emerge. US and Canadian studies also highlight that findings vary by local context.

In the UK, lone parents claiming Family Credit (FC) were more likely to enter marriage or cohabitation than non-recipients. This implies that FC did not act as a disincentive to forming unions notwithstanding that dual-earner families could lose entitlement to the benefit. However, following the introduction of WFTC, single parents were less likely (by 2.4 percentage points) to form a union than single childless women. This impact was more pronounced for women with younger children (two to three percentage points). This difference in UK findings might be because WFTC was more generous than FC; it reduced the opportunity cost of being a single parent.

The US literature shows that AFDC could have a negative, positive or no effect on marriage and cohabitation. Moreover, the effects varied by race and level of control for unobserved factors for those using non-experimental approaches in their analyses. For example, one study shows that for non-black, never-married mothers a \$100 increase in monthly benefit reduced the probability of marriage by 19.5 per cent, but for black, never-married mothers the probability of marriage increased by 19.8 per cent.

US results for the impact of welfare reforms (AFDC waivers and TANF) are also mixed.<sup>1</sup> In terms of non-experimental approaches, for example, one study finds that AFDC waivers and TANF States reduced marriage rates, but another that AFDC waivers increased marriage but TANF had no significant effects. Similarly, social experiments are inconsistent on whether there is relationship between welfare policies and marriage rates for single parents. However, a meta-analysis reveals that, in general, programmes have tended not to have an effect for single parents on rates of marriage or cohabitation.

There is some indication that welfare systems can have a longer term impact on some people's demographic behaviour. For example, one study shows that for two-parent families their rates of marriage/cohabitation were higher (by seven percentage points) 29 to 41 months after joining an employment programme; another study shows that after seven years long-term benefit recipients on an employment programme were less likely to have divorced than non-participants – although no significant effect was found for more recent recipients.

In general, the effect of EITC on partnering has been small and statistically insignificant.

US and Canadian research also shows that local context can have an important bearing on impact estimates. For instance, the Canadian Self-Sufficiency Project (SSP), which was an earnings supplement paid to single parents in receipt of benefits when they entered full-time employment, was evaluated using experiments in two culturally and geographically distinct provinces: British Columbia and New Brunswick. It decreased the probability of being ever married or cohabitating in British Columbia by 2.5 percentage points (a 16 per cent fall); but increased the probability of being ever married or cohabitating in New Brunswick by 4.3 percentage points (a 22 per cent rise).

The waivers covered introducing family caps, extending income disregards, amending time limits and expanding eligibility for two-parent families.

The review also includes two pension studies: First, a Canadian study finds that the removal of marriage penalties for surviving spouses in the Canadian pension system showed that remarriage rates for females aged under 65 and for males aged 45 to 59 increased significantly. However, there are queries about the quality of the data used. Secondly, an analysis of the impact of social security (payroll) taxes for defined-benefit, Pay As You Go (PAYG) pension schemes on family formation across the Organisation for Ecomonica Cooperation and Development (OECD) finds that social security payroll taxes slightly reduce net marriage rates. However, this analysis includes several simplifying assumptions which limit its applicability.

Tentatively, this review finds that whilst there are some studies showing a welfare effect on marriage/cohabitation typically for sub-groups, there is, overall, no consistent evidence for a large and significant impact on union formation. Indeed, there is no set of studies that encompass analyses of an extensive range of sub-groups, explores longer term effects, has high external validity, and uses robust data.

## Welfare systems and union dissolution

A smaller number of studies assess whether financial incentives in welfare systems are sufficient to lead to increased levels of divorce and separation. A problem with this literature is that it does not always distinguish between the dissolution of marriages and cohabitation, yet different factors may be in operation.

Only one UK study was identified, which uses a quasi-experimental approach to explore whether WFTC had an impact on mothers' decisions to dissolve unions. Overall, it found no impact. However, a sub-group, mothers with male partners on low earnings (that is, working less than 16 hours per week), had higher divorce/separation rates (2.4 percentage points) than married women without children. Moreover, for this sub-group the divorce/separation rate was slightly higher where the mother had a higher level of education or where there were young children present.

US studies, together with an Australian study, provide mixed international evidence on whether there is a welfare effect on union dissolution. They do, however, serve to highlight the role of other factors in dissolutions. For example, husbands with poor economic positions, couples dissatisfied with their marriage, males feeling depressed or anxious are all associated with higher rates of divorce/separation.

Two US studies suggest that AFDC had little or no effect on union dissolutions, but another shows that states with AFDC waivers had lower rates of divorce (5.5 per cent) compared to those states without waivers. So welfare reform, as policy makers intended, could lead to a fall in the number of new divorces.

To the extent that there is an effect it appears to be larger for certain sub-groups; the UK and Australian studies suggest that the effect is greater in families with lower earnings. However, the small number of studies, the mixed findings,

doubts about the robustness of some of the studies and that they do not always distinguish between dissolving marriages and cohabitations mean that there is no overwhelming evidence that welfare systems have had a major impact on union dissolution.

## Welfare systems, childbearing and single parents

The review found relatively few studies on fertility and moves into single parenthood at the European level. González (2005 and 2007) finds a positive and significant association between welfare benefits and the incidence of single motherhood. However, as González' (2007) analysis becomes more sophisticated the size of the impact of benefits on single motherhood and single headship becomes smaller and eventually statistically insignificant.

Of the three UK studies of WFTC reform, two find no significant impacts and the third an increase in birth rate for couples.

Findings from US literature on the impact of AFDC on female headship show only small or no effects. Hoynes (1997) finds 'no evidence that AFDC has a significant effect on female headship decisions'. Lichter et al. (1997) find that many of the variables they analyse only have a relatively small effect on female headship, and argue that cultural changes underpin the observed increase in female headship. More recent studies that take into account unobserved variables at the level of the individual find that the welfare effect is not only small but statistically insignificant.

Early US studies suggest a positive welfare effect on the fertility of unmarried women. But these effects largely disappear once controls for unobserved state characteristics are introduced into analysts' models. Subsequent studies suggest minimal or no effect of welfare on fertility – with the exception of family cap policies. The family cap is a mixed picture – most studies find little or no effect; a small number of studies report either a small positive effect, or counter-intuitively a small negative effect.

## Implications for the UK welfare system

Identifying the policy implications of the reviewed literature for the UK is difficult. The underlying theory can be ambiguous about the direction of any welfare effect. Factors other than the welfare system have a key role in influencing demographic behaviour. Empirical studies provide mixed findings. Moreover, differences in policies and national contexts mean that findings from overseas about the direction, magnitude and significance of welfare effects cannot simply be applied to the UK.

Nevertheless, there are policy options in the literature to address union/marriage penalties and reduce (non-marital) childbearing.

The former include:

- changing the unit of assessment from the family/household to the individual;
- introducing a transferable allowance for couples, and
- targeting families with children by, for instance, increasing the basic credit in WTC for couples with children.

Notwithstanding these policy options there may be reasons for retaining union/marriage penalties, for example, couples – in contrast to single adults – benefit from household economies of scale, usually higher incomes, and from the time that a second adult can devote to a family. The latter policy options include in the short-term, improving sex education in schools, contraceptive take-up, and access to early abortion. In the longer term, they include increasing opportunities for employment and further education for client groups (both females and males).

The complexity of welfare systems' interactions with demographic behaviour, the role of other factors, and the absence of cost-effectiveness studies on using welfare systems to influence family policy inevitably raises the question of whether other policy instruments, such as family counselling services, would be more efficient and effective at 'nudging' family structure in any desired direction. The review did not examine this other policy literature, but the mixed and small to modest impact that only some studies show for a welfare effect on family structure suggests limited scope for influencing demographic behaviour through the benefit and tax credit systems.

There are also potential tradeoffs in policy objectives. Welfare systems designed to promote certain family arrangements may have to do so at the expense of other policy goals, such as being progressive, treating families with the same total income equally, maximising work incentives and achieving simplicity in administrative systems and for customers.

#### Conclusions

The actual impact that financial incentives have on family structure is contested. Analysts across countries and over time report mixed findings.

The evidence base for the UK is small. However, the international literature does provide some guidance on how to conduct analyses in this area. There is a need for UK research that is more up-to-date in terms of policies/programmes covered, that has the potential to provide estimates of longer-term impacts and has sufficiently large sample sizes to allow extensive sub-group analyses.

Whilst there are studies finding significant impacts, these tend to be small and are countered by studies finding no relationship or the opposite effect. To the extent that some studies provide evidence of a welfare effect on family structure, its magnitude is often smaller than classical economic theory might predict. On balance the reviewed literature shows that there is no consistent and robust evidence to support claims that the welfare system has a significant impact upon family structure.

# 1 Introduction

#### 1.1 Introduction

'All social policies create incentives, and most create at least some that are undesirable in the eyes of policymakers.'

(Ellwood (2000b:1063))

Many social and fiscal policies incorporate financial incentives and disincentives that could affect family structure. Some commentators argue that financial incentives in welfare systems affect demographic behaviour (see, for instance, Draper (2008) and Morgan (2007)). However, whether financial incentives in the welfare system actually affect behaviour is contested (Blundell and Walker, 2001). This report reviews the evidence on the influence of financial incentives in the welfare system on demographic behaviours such as partnering, separating and childbearing. The review also highlights the implications of reported findings for the United Kingdom (UK).

Most of the evidence reviewed is from the United States (US), which reflects longstanding academic and public debates about whether the tax and transfer system impacts upon family structure (Moffitt, 1998a:1). There is a popular notion that '...welfare provides an incentive for women to not marry or remarry, to have children out of wedlock, and to live independently rather than at home with parents...' (Moffitt, 1998a:1). Morgan (2007:121) outlines how this might happen:

'...today's benefits system can be seen as underwriting a decision to have children looked after by a lone parent when the earning potential of the father is relatively weak. Potentially, the benefits system has three effects. First, it can encourage lone parenthood rather than couple formation because of the bias in tax and benefits systems against couples – particularly single-earner couples. Second, it can encourage childbearing as opposed to a decision not to have children. Third, the benefits system can, itself, bring about labour market conditions that are less conducive to couples taking a decision to marry.'

If the welfare system discourages marriage or increases non-marital births there is a concern that this might adversely affect child outcomes given evidence (for example, for educational attainment, teen parenthood, poverty and health) that children appear to benefit from living in two-parent families (provided they have a *'low-conflict relationship'*) (Roberts, 2008:1; see also Ermisch and Francesconi (2001), Rodgers and Pryor (1998), and McLanahan and Sandefur (1994)); although poor outcomes for children may be more due to income poverty than family structure (see, for instance, Joshi *et al.*, 1999 and Geronimus, 1997).<sup>2</sup>

In general, this review finds that the effect that welfare systems have on family structure is mixed; some studies report a positive effect, others a negative effect and some no significant effect. Where there is a welfare effect, it tends to be relatively small sized; and less than might be expected on the basis of 'classical' economic theory. Moreover, the method used to analyse the impact of benefits and tax credits on demographic behaviour has a bearing on the magnitude and significance of any effect. Hence how the studies were conducted is an important part of this review.

The remainder of this chapter outlines the review's objectives, see Section 1.2, how it was conducted, see Section 1.3, provides background information on UK and US demographics and welfare systems, see Section 1.4, and outlines theoretical, see Section 1.5 and analytical, see Section 1.6, issues relevant to the review.

# 1.2 Review's objectives and scope

The review is focused on people's behavioural responses to financial incentives in welfare systems. The review evaluates evidence on the influence of financial incentives in the tax credit and benefit systems on:

- partnering (marriage and cohabitation);
- separation (divorce and separation), and
- childbearing.

It outlines the mechanisms that may influence demographic behaviour. In addition, the review considers the methodological challenges underpinning the studies, gaps in the evidence base and the extent to which international evidence is applicable to the UK.

The scope of the review has had to be limited because there is an extensive literature, especially from the US, on welfare systems and family structure. Thus the review is focused on studies:

- published after 1997; and
- conducted in English-speaking countries that have tax credits or benefits which are similar in design to those in the UK.

There is also evidence that teen children of single parents living in threegeneration households in the US have outcomes that are as good as those living in two-parent families (see Bitler *et al.*, 2006:3).

Excluded from the review are:

- any influence the wider tax system might have on family structure (see, for instance, Buffeteau and Échevin (2008));
- the identification and evaluation of work incentives;
- the outcomes for children of financial incentives in the welfare system (see, for example, Sherman (2001));
- welfare programmes where there is no direct transfer of cash, such as Food Stamps and Medicaid which are specific US programmes that have no real UK equivalent;
- the literature on child support and maintenance payments (see, for instance, Aizer and McLanahan (2005), Plotnick *et al.* (2006), and Walker and Zhu (2006)).

#### 1.3 Conduct of the review

This review involves the synthesis of a relatively wide range of published sources. The search strategy was purposive involving 'informed', extensive sweeps rather than systematic searches (Gough, 2007). Searches were conducted using academic online databases using combinations of keyword descriptors based on the research objectives, such as 'family formation', 'marriage penalty' and 'marriage bonus'. The names of certain key authors were also used as search terms, not only to identify other sources by them but also citations of their work by other writers. In addition, a few experts in the field were contacted and asked to identify further studies and researchers.

# 1.4 Background: UK and US family structures and welfare systems

As most of the studies reviewed in subsequent chapters come from the UK and US, this section outlines recent demographic changes affecting families in the UK and US, as well as selected components of their welfare systems.

## 1.4.1 Demographic changes affecting UK families

In many countries, including the UK and US, there have been substantial changes to family lives and structures since the Second World War (Williams, 2005:18). In general, there has been less marriage with those marrying tending to doing so later in life, more cohabitation, more same sex partnerships, more divorce, childbearing has been delayed to later in life and there are more births outside of marriage (McConnell and Wilson, 2007:12). A small but increasing proportion of the population also form 'reconstituted families' as families dissolve and reform. These trends reflect a complex interplay of cultural, economic, legal, political, religious and social factors.

#### Families – overview

Although the motif for the discussion below is the changing nature of the family, there is a high degree of continuity in familial relationships. Most children live in two-parent families and most marriages do not end in divorce. Although cohabitations typically have a short duration (see ahead), between 2001 and 2003 most family relationships (94 per cent of couples and 83 per cent of single parents) remained unchanged (Barnes *et al.*, 2005:26-27).

Nevertheless, families in the UK have been changing since the 1980s. The overall number of families in the UK is growing; in 2006 there were 17.1 million families which was over two million more than in 1971 (McConnell and Wilson, 2007:2). Whilst the majority, 71 per cent (over 12 million), of the families in 2006 were married couple families, over the previous ten years the proportion of married couples has fallen by over four per cent. This decline has been accompanied by an increase of over 60 per cent in cohabitating couples (to 2.3 million) and an increase of eight per cent (to 2.6 million) in single parent families.

Most dependent children live with two parents, but this proportion has been falling. In the UK, in 2006, there were 7.4 million families with dependent children (McConnell and Wilson, 2007:5). Most of these children (65 per cent) lived in married couple families, whilst around a quarter (24 per cent) lived in single parent families and just under an eighth (12 per cent) lived in cohabitating couple families.

In general, cohabitating couple families are younger than married couple families (McConnell and Wilson, 2007:6). This reflects that cohabitation can be a precursor to marriage and that younger generations are increasingly tolerant of cohabitation. Similarly, heads of families with dependent children tend to be younger (in their 30s or early 40s) than those with no children or with non-dependent children.

#### Marriage

There has been a decline in marriage rates in most European countries, including the UK. Marriage rates have declined in the UK since the 1970s (McConnell and Wilson, 2007:13).<sup>3</sup> For instance, the number of single women who got married in 1971 per 1,000 of the single population aged 16 or over was 97 and by 2004 it was 30.8. In 2006, England and Wales had the lowest number of marriages since 1895.

In recent years, people are entering their first marriages later in life. In England and Wales, the median ages of first marriages increased from 24 for males and 21.6 for females in 1961 to 30.4 for males and 28.3 for females in 2004 (McConnell and Wilson, 2007:12). This reflects delays in first marriages and more remarriages (Wilson and Smallwood, 2008:19). The trend is associated with an increase in cohabitation, and most marriages (70 per cent) are preceded by a period of

In England and Wales between 2000 and 2004 the marriage rate increased, but since 2005 the rate has continued its longer term downward trend.

cohabitation (Williams, 2005:14). Thus the 'shift to cohabitation' is the 'main engine' for current trends in marriage (Ermisch and Francesconi, 2000:159).

Male and female rates of remarriage following divorce have also fallen; for example, in 1971 the number of divorced men marrying per 1,000 divorced people aged 16 or over was 227.3, but by 2004 it was 45.7 (McConnell and Wilson, 2007:13). However, whilst the number of divorces has been increasing, the number of remarriages in England and Wales has, since the early 1970s, remained 'fairly' constant. Moreover, divorcees as a proportion of those marrying has been increasing, between 1963 and 2003 it increased by 20 percentage points to over a guarter (Wilson and Smallwood, 2008:18).

#### Cohabitation

As already mentioned previously, in recent years the number of couples cohabitating, either as a precursor or as a substitute to (re)marriage, has been increasing. Thus the proportion of women aged 18 to 49 in cohabiting unions trebled between 1976 (nine per cent) and 2004 (28 per cent) (McConnell and Wilson, 2007:15). In 2005, 39 per cent of single people aged 25 to 34 were cohabiting. McConnell and Wilson (2007:15) observe that the rate of cohabitation for the most recent age cohort (those born 1956-60) is more stable. Consequently, the rate of increase in cohabitation may slow.

In most Western European countries, including the UK, cohabiting unions tend to be short-lived, either dissolving or converting into marriage (Ermisch and Francesconi, 2000:157-158; Kiernan, 1999:32). In the UK during the 1990s, almost a half (45 per cent) of cohabiting unions ended within two years by either converting into marriage or dissolving (Ermisch, 2002:1-2). In 2004, 16 per cent of adults aged 16 to 59 said that they had experienced at least one cohabitating union that did not lead to marriage (McConnell and Wilson, 2007:15). Unions with children were less likely to convert to marriage and were more likely to dissolve than those for childless women (Ermisch, 2002:2), although mothers tend to remain in cohabitations for longer than non-mothers (Ermisch and Francesconi, 2000:168). Thus, cohabitating women giving birth to a child were at a relatively high risk of becoming a single parent. Those cohabiting unions at greater risk of having a baby were those facing poorer financial circumstances, notably those where the male was not in employment and the woman's father had had semiskilled or unskilled manual job (Ermisch and Francesconi, 2000:169). Unions that did convert to marriage tended to be those where the male had higher earnings and so contributed more income to the couple (Ermisch, 2002:2; see also Ermisch and Francesconi, 2000:168).

People leaving cohabitation tend to repartner sooner than those divorcing, mainly because they tend to be younger (Ermisch, 2002:3-4). The median duration for repartnering after dissolution of a cohabiting union was 2.3 years (compared to five years following the dissolution of a marriage) with most (70 per cent) of those leaving the cohabitation repartnering within five years. Those with children

leaving a union tend to take longer to repartner; for instance, the proportion repartnering with children within five years was 15 percentage points lower than for those without children.

#### Union dissolution

The divorce rate in England and Wales has increased in recent years. It increased from 12.9 per 1,000 married population in 1998 to 14 per 1,000 married population in 2004, but fell slightly to 13.1 in 2005. Historically, increases in divorce follow legal reforms, such as the Divorce Reform Act 1969 and the Matrimonial and Family Proceedings Act 1984, which allowed couples to divorce after one year of marriage. Since the early 1970s there has been a rapid rise in the rate of divorce after one to two years of marriage; the rate peaks at three to four years of marriage and then declines over time (Ermisch, 2002:4-5). Nonetheless, the annual divorce rate has stabilised at about six per cent for those born after 1951 (McConnell and Wilson, 2007:13).

Unions may dissolve for a variety of reasons, and the causal mechanisms are complex with how, for instance, economic factors interact with 'marital quality' and well-being unclear (see Bradbury and Norris, 2005:426-427). However, Blekesaune (2008), using longitudinal data from the British Household Survey Panel, shows that in the UK male or female unemployment increases the risk of partnership dissolution (of both married and cohabitating couples). Male unemployment is associated with a 33 per cent larger risk of partnership dissolution and female unemployment with an 83 per cent larger risk of dissolution compared to partnerships with no unemployment in the previous year (Blekesaune, 2008:8; see also Böheim and Ermisch, 1999). Moreover, the increased risk due to male unemployment is associated with low levels of financial satisfaction amongst partnered women and this also helps to explain why unemployment leads to increased union dissolution. Thus low family income would appear to destabilise partnerships for women. The increased risk of female unemployment terminating a union was associated with mature, rather than recent, partnerships. (Indicating some support for an 'independence effect' (see ahead) for women in relationships that have lasted some years.) Blekesaune (2008) did not include receipt of benefits in her models, but presumably many of the unemployed couples would have been in receipt of Jobseeker's Allowance (JSA). Nonetheless, the research suggests that the presence of unemployment, rather than the welfare system per se, is a key factor in union dissolution.

Other research suggests that age at marriage is also an important factor in marital instability, with couples marrying at a younger age more likely to separate (see Bradbury and Norris, 2005:426; Ermisch and Francesconi, 2000:168; and Böheim and Ermisch, 1999). This is possibly because they have spent a shorter period of time looking for a 'good' match. In addition, for couples with dependent children, union dissolution (married and cohabiting) is less likely if the couple enjoys an unexpected improvement in their financial circumstances (Böheim and Ermisch, 1999).

#### Single parents

There has been a rapid increase in single parent households. The proportion of dependent children living in single parent families has increased from 18 per cent in 1981 to 23 per cent in 2001 (Bakeo and Clarke, 2004:7). The growth in single parents in the 1970s and early 1980s was due to increases in divorce and separation. Since the mid-1980s, the increase is mainly due to the increase in non-marital births. In the 2000s the flow from couple families into single parenthood was three per cent per annum (Clarke and McKay, 2008:59-60), whilst the rate at which single parents form new unions (with children) was nine per cent per annum. There are also ethnic differences in single parenthood, with high rates amongst African-Caribbean women and lower rates amongst Pakistani and Bangladeshi households (Williams, 2005:22).

#### Fertility

Since the mid-1960s, and until recently, there has been a rapid fall in the fertility rate. The total fertility rate fell rapidly during the 1960s and 1970s, but increased from 1.63 in 2001 to 1.84 in 2006. The decline in fertility reflects that more women have remained childless, more women have been having children later in their lives and fewer women are having larger sized families (Dixon and Margo, 2006:72-73). Women may be delaying having children in order to maximise their life-long earnings – a 'mid-skilled' women forgoes, on average, £564,000 (at 2005 prices) in lifetime earnings if she has a baby at 24 compared to a childless women (Dixon and Margo, 2006:76). Not surprisingly, increases in women's earnings are associated with postponing childbirth. As a proportion of forgone earnings this 'fertility penalty' is larger for women with lower skills (and hence low-income potential).

In the UK, the number and proportion of non-marital births has been increasing since 1975. In 1999, 39 per cent of births in Great Britain were outside marriage (Ermisch, 2000:3) but by 2006 it was 43.7 per cent.<sup>4</sup> There has been an increase in childbearing in first cohabitations (Ermisch, 2000:5). Unemployment is associated with higher non-marital childbearing (Ermisch, 2000). A one percentage point increase in unemployment increases, on average, the annual probability of a woman having a non-marital birth by 0.4 percentage points (a ten per cent increase). In addition, the risk of non-marital childbearing is lower for women with more educational qualifications (Ermisch, 2000:15).

This increase in non-marital live births suggests that the previously strong association between marriage and fertility has weakened (Buffeteau and Échevin, 2008:2-3). More generally, it signifies a weakening of the link between marriage and parenting.

<sup>&</sup>lt;sup>4</sup> Table 2.2, Population Trends (2008:49).

#### 1.4.2 UK welfare system

The UK research reviewed here mainly involves two components of the welfare system: Income Support (IS) and tax credits. Accordingly, other elements of the UK benefit and transfer system are not discussed here.

#### Income Support

IS is an income-based benefit available to people on low-incomes, mainly lone parents, people who are sick or disabled and carers. To be eligible recipients must not work more than 16 hours per week. The amount of benefit includes a personal allowance that varies by age and household composition. IS included child premia which in 2003, were transferred to the Child Tax Credit (CTC).

Critically, for analysts that investigate the impact of policy changes on family structure, when Working Families Tax Credit (WFTC) was introduced in 1999 (see ahead), IS payments to families with children were increased.

#### Tax credits

The UK studies reviewed in this report tend to focus on WFTC, which replaced Family Credit (FC) (a means-tested in-work benefit) in October 1999.<sup>5</sup> WFTC was designed to enhance the work incentives of parents in low-income households. To be eligible for WFTC (or FC) a family had to have at least one dependent child, at least one adult who worked 16 hours or more per week and a 'low' or 'middle' household income. Incomes below a threshold received the maximum credit which was gradually withdrawn (the taper) as income rose up to the level at which eligibility ceased. In comparison to FC, WFTC was more generous due to an increased credit for children aged up to 11 years; a higher income threshold; a lower taper rate; excluding child maintenance from its definition of income; and a more generous childcare credit.

WFTC, and the associated Disabled Person's Tax Credit and Children's Tax Credit, were replaced by Working Tax Credit (WTC) and CTC in April 2003.<sup>6</sup> For families with children WTC essentially replicates the non-child elements of WFTC; however, it is also payable to singles and couples without dependent children. The credit comprises a basic element, a lone parent/couple element, a childcare element (if applicable) and additional elements for working 30 or more hours per week, having an impairment or being aged 50 years or over. Although WTC and CTC are separate programmes they are subject to the same means-test, which is applied to a couple's joint annual income.

FC was introduced in 1988 and replaced Family Income Supplement (which was introduced in 1971) (see Strickland, 1998).

The Children's Tax Credit was introduced in April 2001 and provided families with extra income tax relief. Effectively, it was replaced by CTC.

CTC separated out and merged the child premia in WFTC and IS/JSA.<sup>7</sup> It is meanstested, but payable irrespective of the employment status of the recipient. It consists of a family element (a higher rate is paid if there is a child aged under one year), premia for each dependent child and, where appropriate, an element for disabled children. The CTC is paid to the adult with the main responsibility for caring for the dependent child(ren).

For the purposes of WTC and CTC couples are defined as men and women who are married or cohabitating (that is, living together as if they are married), or two people of the same sex in a civil partnership, or living together as if they are in a civil partnership. Tax credits in the UK take account of the joint income of couples. There is no difference in the treatment of couples or single adults for families with children. For WTC, an additional element is payable to working childless couples (as against working childless single people), and the eligibility requirements in terms of hours worked are slightly different for couples as against single people. However, apart from the support for childcare in tax credits, households with children with the same income and the same number of children will still receive the same amount of tax credits.

In addition, when CTC and WTC were introduced, the 30 hour element was changed so that couples could combine their usual working hours to qualify for this element of the credit. This might have had some impact on any couple penalty/bonus after 2003 as where couples were both working less than 30 hours individually, but 30 or more hours jointly would benefit.

## 1.4.3 Demographic changes affecting US families

Marriage and cohabitation in the US

Most Americans (about 90 per cent) marry at some point in their lives (Roberts, 2007:2), but there is a wide variation by race – flows into marriage are lower in States with higher proportion of African Americans (Bitler *et al.*, 2004:224). However, since the end of the Second World War rates of marriage have been declining in the US. In 1970, the marriage rate per 1,000 unmarried adult women was 76.5 but by 2004 it had fallen to 39.9 (Burstein, 2007:388). The corollary is an increase in non-marital cohabitation. By the mid-1990s most marriages were preceded by cohabitation (Burstein, 2007:388).

The decline in the rate of marriage varies by sub-group and is higher for African Americans and for those with lower educational qualifications. Amongst never-married women living in poverty, rates of marriage are low and declining (Gassman-Pines and Yoshikawa, 2006:11). Moreover, women with more education are tending to marry later in life (Burstein, 2007:388; Ellwood and Jencks, 2004:13).

Existing families on IS/JSA have been gradually migrated onto CTC, although the process is not yet complete – in April 2008 around 365 thousand families still received CTC-equivalent support via IS/JSA, compared with 1.018 million out of work families receiving CTC.

Most American marriages and remarriages follow a period of cohabitation (Smock et al., 2005:680). Research shows that in the US cohabitation is associated with age, race, educational background, religiosity and family background (Moffitt et al., 2003:267).

As in the UK, cohabiting unions in the US have shorter durations and higher exit rates compared to marital unions (Moffitt *et al.*, 1998:260). Compared to the general population, rates of cohabitation amongst benefit recipients are higher (Moffitt *et al.*, 1998:262).

Cohabiting couples with children in the US have lower household incomes than married couples with children (Acs and Maag, 2005:2). This difference reflects the lower age and educational attainment of the mothers and the lower employment status of fathers in cohabiting couples compared to married couples. Their lower incomes mean that cohabitating couples were more likely to participate in the welfare system than their married counterparts.

#### Divorce

In the US rates of divorce have increased since the Second World War. The rate of divorce per 1,000 married women was 9.2 in 1960, rose to 22.6 in 1980 but has fallen since then; it was 17.7 in 2004 (Burstein, 2007:388). The introduction of no-fault divorce in the US is partly responsible for facilitating the increase in the number of divorces (Morgan, 2007:128-129). Divorce and separation rates vary by race and level of education. Rates of divorce and separation are highest for African Americans and women with less than 12 years of schooling, and lowest for non-Hispanic whites and college educated women.

Divorced men in the US are more likely to remarry than divorced women, and divorced women with children are less likely to remarry than those without children (Bergstrom, 1996:1917).

#### Single mothers

The number of unmarried women having children, including teenage pregnancies, has increased in the US. The proportion of children aged under 18 living in single parent families was about ten per cent in 1965 and this rose to 29 per cent in 1997 but fell back to 27 per cent in 2001 (Ellwood and Jencks, 2004:10). Indeed, between 1997 and 1999 the share of children living in single parent families fell by 2.1 percentage points (Acs and Nelson, 2004:274). The drivers for this have changed over time. In the 1960s and 1970s the rise was due to the increase in divorce rates, whilst in the 1980s and 1990s it was due to non-marital births. The more recent decline in single parenthood may be due to welfare reform and a tight labour market. This trend is also accompanied by an increase in the share of children living in cohabitating families.

The increase in female-headed households with children has been more dramatic for African American women than for white women – for the former the increase

has been from about 30 per cent in 1970 to over 50 per cent in 1993, compared to an increase from eight per cent in 1968 to 17 per cent in 1993 for white women (Bergstrom, 1996:1917). The reasons for this difference are poorly understood (Ellwood and Jencks, 2004:10).

However, there is evidence that rates of single parenthood are lower for women with lower levels of educational attainment (and hence earnings potential) (Ellwood and Jencks, 2004:10-11; Hoynes, 1997a:104).

#### Fertility

Births outside of marriage were relatively rare in the US until the 1960s (Ellwood and Jencks, 2004:9), but they have increased from 11 per cent in 1970 to 34 per cent in 2002 (Martin *et al.*, 2003 cited by Ryan *et al.*, 2006:104). In addition, second or higher order non-marital births have risen with half of all non-marital births in 1998 to women who already had at least one child (Terry-Humen *et al.*, 2001 cited by Ryan *et al.*, 2006).

Rates of non-marital births are higher amongst African Americans and increasingly for Latino communities, and are higher for those with less education (Ellwood and Jencks, 2004:11-13). Thus:

'...highly educated women are postponing both marriage and childbearing, while less educated women are postponing marriage but not childbearing. The result has been a rapid rise in the fraction of less educated women who have had children but have not married.'

(Ellwood and Jencks (2004:13))

This increase in non-marital childbirth is mainly due to the fall in the marriage rate, and not because women are more likely to have children – that is, most of the increase in non-marital births has occurred in cohabitating unions (Ellwood and Jencks, 2004:9).

#### 1.4.4 US welfare system

The US literature reviewed in subsequent chapters is focused on three components of the welfare system: Aid to Families with Dependent Children (AFDC), its replacement Temporary Assistance for Needy Families (TANF) and the tax credit, Earned Income Tax Credit (EITC). These three programmes are discussed ahead.

#### **ADFC**

AFDC or AFDC-Basic was introduced in 1935 by the Social Security Act and provided cash benefits to low-income families with dependent children. 'Dependent children' were defined as those without the support or care of a biological parent due to death, disability or absence from the home. In practice, this meant AFDC covered mainly single mother families – indeed, since the 1980s the majority of recipients were unmarried single mothers (Moffitt, 2003:313). The definition of a dependent child also implied that AFDC could be claimed by a parent who

remarried or was cohabitating with someone who was not the natural parent – that is, AFDC was payable as long as any partner was not the biological parent of the woman's children. In addition, the incomes of step-parents (up until 1981) and cohabiters were effectively excluded from the assessment of benefit (Moffitt, 2003:294). Since a Supreme Court ruling in 1968, AFDC could not be denied to cohabiting mothers; although the male's relationship to the child and his degree of economic support must be considered. However, states had some discretion in setting the rules and so cohabitation within AFDC could still be difficult to achieve in certain states.

In 1961 the Federal Government gave states the option of introducing the AFDC-Unemployed Parent (AFDC-UP) programme, which included two parent families where the primary earner was unable to find employment of over 100 hours per month. AFDC-UP was believed to encourage marriage, because a low-income couple might be financially better off if they married and claimed AFDC-UP rather than if they cohabitated and the woman received AFDC-Basic. It operated in approximately a half of States before 1990 and subsequently (as required by the Family Support Act 1988) in all states. The income and employment eligibility conditions for AFDC-UP were more stringent than for AFDC-Basic; for instance, the unemployed parent had to have a history of employment and both parents' income was counted.

Over time, the AFDC caseload has varied; it increased dramatically in the 1970s and gradually declined until 1982 when it stabilised before increasing rapidly again during the early 1990s and then declined (Moffitt, 2003:307-9). Nonetheless, there was a general decline in the real value of AFDC benefits from the late 60s to the early 1990s when there was a modest increase in value (Hoynes, 1997a:100-102; Moffitt, 1998b:60 and 2003:307). For instance, real spending on AFDC fell from \$21.7m in 1984 to \$20.4m in 1996 (Meyer and Rosenbaum, 2000:1036). Yet over the same period the caseload increased from 10.9m to 12.6m; a 15 per cent increase.

The Omnibus Budget Reconciliation Act 1981 allowed states to waiver Federal AFDC rules and experiment with different forms of provision. Between 1993 and 1995 approximately a half of states implemented some sort of waiver programme (Bitler *et al.*, 2004:216). Two policy areas directly relevant to this review are demonstration projects with 'family caps' and AFDC-UP (Maynard *et al.*, 1998:139-140). In 1992, New Jersey was the first state to experiment with not increasing AFDC benefits when recipients had additional children after ten months of claiming benefit (the 'family cap'). Family caps became a feature of several AFDC waiver and TANF programmes (Maynard *et al.*, 1998:141-146). Waivers for AFDC-UP could relax the eligibility criteria on two-parent families, such as the rule that the principle wage earner had to work for less than 100 hours per month (Maynard *et al.*, 1998:146-150). In general, however, State waiver programmes did not concentrate on family formation, but on getting recipients into employment (Fein *et al.*, 2002:1).

All AFDC recipients (and, later on, TANF participants) were eligible for Food Stamps and Medicaid (Moffitt, 2003:295). The Food Stamp programme involves awarding food coupons to low-income households that are redeemable for food items, and Medicaid provides subsidised medical care to low-income families. For those people that are eligible, Food Stamps are paid irrespective of family structure.<sup>8</sup> In calculating the amount of AFDC benefit, Food Stamp benefits as well as EITC (see ahead) and housing subsidies were, generally, excluded.<sup>9</sup>

Critically for the US research reported here AFDC benefits were set at state level, and there was a high variation between states; there was a sixfold difference between the most and least generous (Hoynes, 1997a:90; Moffitt, 2003:293). This variation has been used in studies to consider the impact of AFDC on family structures.

#### **TANF**

TANF, which was introduced by the Personal Responsibility and Work Reconciliation Act (PRWORA), replaced AFDC and AFDC-UP in 1996. Concerns that AFDC might have encouraged single mother families were reflected in the objectives of TANF, which included preventing and reducing non-marital births and teen pregnancies and births, and promoting the formation and maintenance of two-parent families as well as encouraging moves into employment (Burstein, 2007:412; Maynard et al., 1998:134). Implicit in TANF is the assumption that financial incentives can be used to influence attitudes towards family structure and so change actual behaviour (Mauldon et al., 2002:6). Indeed, underpinning the programme is a belief that marriage is an exit route from welfare and poverty (Bitler et al., 2004:213). Although states could impose family caps and extend support to two-parent families as well as require that teenage single mothers in receipt of TANF live with their parents and stay in school (see ahead), there were few provisions in the legislation directly linked to its family-related objectives (Moffitt, 2003:307) and few states have taken substantial steps to change demographic behaviour (Fein et al., 2002:1).

The programme has four key features. First, it is a workfare programme – parents' receipt of benefit is linked to undertaking activities to gain paid work.<sup>10</sup> Secondly, TANF established lifetime time limits (up to five years) for the receipt of Federal

- Eligibility for Food Stamps is based on the 'assistance unit' and a cohabitating couple that shares and prepares food together would, like a married couple, be eligible for benefits (Acs and Maag, 2005:5).
- How EITC interacts with AFDC and TANF has changed over time, at times it has been counted as income, but at other times it has been disregarded (see Meyer and Rosenbaum, 2000:1032). However, since January 1991 it has mainly been disregarded in means-tested programmes. Nonetheless, the Food Stamp programme counted AFDC benefits as income.
- Post-1981 workfare-type demonstration projects were also a feature of some states with AFDC waivers.

funds by families (although 20 per cent of the state caseload was exempt from this provision). Thirdly, aspects of the design of the programme and its financing, via block grants, were devolved to state level. As with AFDC, states were free to set their own benefit levels; however, benefits could be in cash and/or in-kind services (such as childcare) (Moffitt, 2003:311). As a consequence there are substantial differences in welfare programmes across the US (Schoeni and Blank, 2000:8) and the magnitude of any TANF effect on family structure will vary. Fourthly, as States could serve two-parent families based on their income, TANF removed the AFDC disincentives 'against either living with or marrying the children's father, or against marrying rather than living with another man.' (Burstein, 2007:412). The expansion of benefits to two-parent families under TANF (and previously some state waiver reforms) is '...the only aspect of welfare reform that directly affects the incentive to marry.' (Bitler et al., 2004:216). However, for the women to be eligible for benefits the husbands' earnings must be low and this might make the men unattractive as potential spouses. Thus the extension of benefits to two-parent families might not have provided an incentive for women to marry. Although for women who were already married, it might have removed an incentive to divorce and become single in order to be eligible for benefit, something that might have been the case under the previous system.

Eligibility for TANF, like AFDC, does not depend upon marital status, but on family living arrangements. In general, adults unrelated to the TANF unit are assumed not to contribute to it financially, even if they are cohabiting. Related partners, that is, married couples and cohabitating biological fathers, are assumed to contribute to the household. In general, this can generate some complex outcomes (Alm *et al.*, 1999:196):

- a single mother marrying the father of her children who was not previously living with her and who was in low paid work may lose some or all of her benefit, because of her husband's earnings;
- a single mother marrying the father of her children who was not previously living with her and who was unemployed may retain some or all of her benefit, because two-parent families may be eligible for TANF;
- a single mother who marries someone unrelated to her children who is in employment may lose some or all of her benefit regardless of the previous living arrangements, because of her husband's earnings;
- a single mother marrying the father of her children who was previously living with her will not lose any TANF because his earnings would already have been taken into account in the TANF calculation.

The focus of welfare reform in the US was to increase work incentives and the changes have been accompanied by an increase in employment by women (Bitler et al., 2004:216). In real terms, expenditure on TANF has fallen as caseloads have reduced (Moffitt, 2003:307-8).

The Deficit Reduction Act 2005, signed by President Bush in February 2006, reauthorised TANF and also provided funding for marriage and relationship education programmes to promote 'healthy marriages'.

#### **EITC**

EITC, which was introduced in 1975, is an in-work refundable tax credit paid to low-income households.<sup>11</sup> In terms of caseload and expenditure EITC expanded rapidly, especially during the mid-1980s and 1990s.<sup>12</sup> In real terms, the value of EITC increased more than ten-fold between 1984 and 1996 (Meyer and Rosenbaum (2000:1027). Currently, EITC is the largest cash assistance programme in the US (Acs and Maag, 2005:2; Ellwood, 2000a:189). It is a large part of the 'package' that (along with a decline in benefits) has increased work incentives for single parents (Ellwood, 2000a:190-193).

The majority of the credits are received by single parents (Meyer and Rosenbaum, 2000:1030), although it is also available to married taxpayers. Since January 1991, the credit has been larger for taxpayers with two or more children than for those with one child (Hotz and Scholz 2003:146-151). In 1994, a small credit was introduced for childless low-income taxpayers aged between 24 and 65.

To complement the Federal EITC, states can operate their own schemes. The states' EITCs are calculated as a percentage of the Federal EITC and the credit is refundable in most but not all states. EITC is administered through the Federal (and in some cases the state) tax system and is usually paid once a year in the form of a tax refund or adjustment to tax liabilities.

Eligibility is based on the earnings of the 'tax filing unit', and martial status determines who is included in the tax unit. Whilst married couples must file a joint return, unmarried individuals can file either head of household returns (if they have qualifying children) or single returns (if they have no qualifying children). The credit is based on annual family income, not wages.

A non-refundable Federal child tax credit was also introduced in 1998 (Ellwood, 2000b:1064).

## 1.5 Theoretical perspectives and empirical observations

Much of the research reviewed here is informed by economic theory that sees the decision to marry, cohabit or have children as being influenced by financial incentives and disincentives. However, evidence suggests that other factors are influential. For example, Ono (2001, cited in Harknett and Gennetian, 2003:456) argues that the gendered nature of work affects how a woman's economic

Refundable means that if the household's credit is greater than its tax liability the difference is paid to the tax filer by the Treasury.

Ventry (2000) provides a political history of the growth of EITC.

circumstances influence marriage decisions, whilst Wu and Balakrishan (1992) (cited in Harknett and Gennetian, 2003:456) show that attitudes towards marriage have a mediating role on structural variables, such as employment.

This section begins by outlining a heuristic framework for considering the possible links between welfare systems and family structure. It then outlines some of the theoretical debates on the factors influencing demographic behaviour and draws upon supporting empirical sources as appropriate.

### 1.5.1 Understanding welfare systems and family structure

A wide range of factors influence demographic decision-making. Conceptually, the welfare system can be depicted as affecting family structure through its impact on employment and family income (see Figure 1.1) (Fein *et al.*, 2002). To the extent that the welfare system affects family income then it influences the direct financial costs and benefits that people consider when making partnership and childbearing decisions. So, for instance, for single women entering employment the welfare system may increase union formation by eliminating the marriage penalty (or disincentive) previously associated with being on benefits. On the other hand, entering employment may reduce marriage because it enhances women's financial independence. (These possible mechanisms are discussed further ahead.) In some instances the benefit may directly target demographic behaviour, for instance, in the US, TANF payments can be capped when recipients have additional children (known as the 'family cap').

In addition, there may be other non-financial factors arising from the welfare system's effect on employment that affect demographic behaviour. For example, union formation may increase because gaining work improves a woman's self-confidence, self-esteem or widens her social network which then includes more potential partners. However, it may also mean she has less time to search for a partner or to socialise or even for childbearing, so marriage or cohabitation is less likely to occur. (These factors, may, of course, also affect men.)

Whether and how employment and family income affect demographic behaviour will be mediated by individual and cultural factors – and as such they help explain, or clarify the nature of, the relationship between the variables. There are also moderating individual factors that will affect the direction and strength of the relationship between the variables.

Exploring the relationships between these numerous variables is further complicated by decisions on marital status, childbearing and employment being inter-linked (or joint) rather than independent events. Failure to take the interaction between these processes into account will lead to biased estimates in empirical models (Aassve *et al.*, 2006:781-782).

Childbearing cohabitation Marriage/ (e.g. stress, self-esteem, social networks, access Psychological factos (e.g. available time, to family planning) Cultural factors Figure 1.1 Heuristic model for how the welfare system might influence family structure Opprtunities aspriations) influences: Mediating Moderating influences: e.g. gender, age, marital attainment, parental history, educational Family income education, race Benefit income Employment Based on diagram in Fein et al. (2002:2) Welfare system: Other services **Employment** programmes Tax credits Benefits

#### 1.5.2 Marriage and welfare systems

Becker's household production theory of marital formation and dissolution (for instance, Becker (1981)) is very influential in the studies reviewed here. His theory posits that women will choose marriage when its economic benefits exceed the economic benefits of being outside of marriage – that is, people marry to increase their material well-being (that is, to maximise their utility). The relative economic opportunities available both inside and outside of marriage are seen to influence the decision to marry. Marriage is seen as beneficial where there is specialisation between partners – one engages in paid work and the other in home production. Gains in efficiency – through one being employed and earning and the other engaged in household production – allow both to have more leisure time and a higher standard of living when married than when single. Specialisation leads to the prediction that higher-wage women will marry lower-wage men, and vice versa (Burstein, 2007:390). In this way, efficiency is maximised because the relative abilities of each partner, their comparative advantages, differ the most. However, if the wages for males and females converge and/or partners secure similar non-monetary gains from employment or childrearing then the benefits from specialisation and so the gains from marriage decline (Ellwood and Jencks, 2004:14).

However, the notion that material gains only accrue from specialisation has been challenged. Lam (1988 cited in Burstein, 2007:390) argues that sharing 'public goods' (for instance, housing and the benefits of having children) is another source of material gain within marriage. Contrary to Becker's theory these gains may lead to women and men with similar wages marrying. Oppenheimer (1997 cited in Burstein, 2007:391) argues that specialisation is a risky strategy due to possible job loss or desertion. She maintains that couples can maximise their material gains if they both work (full-time) and buy-in household services. Weiss (1997 cited in Burstein, 2007:391) highlights other sources of material gain in marriage: economies of scale from a larger household and that whilst one partner works the other can undertake education and training (that is, invest in human capital). Indeed, economies of scale, in terms of resources and time, provide a 'powerful incentive' (Ribar, 2003:6) for individuals to join together in a single household and share resources.

Traditional economic theory implies that for women there is a choice between being married and being single and the decision is influenced by the woman's wages, the wages of potential partners, and benefits and tax credits as well as other observable and unobservable factors (Dickert-Conlin and Houser, 1999:4). However, the effect of earnings on the decision to marry or cohabit is ambiguous. Females with higher earnings or good employment records are **more** likely to marry or cohabit because they are attractive as potential partners, but equally they are **less** likely to marry or cohabit because they have the financial independence to do so (the 'independence effect'). The former implies that marriage is a 'normal good' – as opposed to an 'inferior good' where consumption falls as income increases – and the correlation between wages and prevalence of single motherhood is

negative. The latter implies that with rising female labour participation rates and higher real wages the economic or 'opportunity cost' of remaining single (or getting divorced) has fallen.

Studies that have sought to investigate the association between women's employment and marriage have had mixed results, finding both positive and negative effects (Burstein, 2007:405; Harknett and Gennetian, 2003:457). More recent studies suggest that women – like men – are more likely to marry the more they earn (Gassman-Pines and Yoshikawa, 2006:12-13). Yet studies have tended not to consider the distinction between deciding not to marry and marrying later on in life (Burstein, 2007:405-406). The independence hypothesis does not help to explain why women with higher incomes might choose to marry later on, say, after they have established their careers.

A related idea is the 'marriage market' – as expressed by the ratio of men to women. Geographically, the number of marriageable men and women may be constrained and out of balance. The reason why some low-income single women do not marry has been attributed to the lack of marriageable men within a locality (Burstein, 2007:396). Here marriageable men may be seen as those with steady earnings from employment – a view associated with the sociologist William Julius Wilson. In some communities in the US the availability of marriageable men may be reduced by unemployment, imprisonment or premature death (due to gang violence) (Roberts, 2007:4) or long-term severe socio-economic disadvantages (Geronimus, 1997:415). However, simply dividing people into marriageable and unmarriageable may be too crude a distinction, especially as what someone looks for in a potential partner may change over time and some individuals may restrict their preferences for non-economic reasons, such as not considering people from a different ethnic community. In addition, falls in marriage rates have also been observed for 'marriageable men', such as those with higher levels of education (Del Bono, 2004:3; see also Lichter et al., 1997:115). Indeed, high-earning African American males, who due to their incomes could be regarded as 'marriageable', have lower marriage rates than white Americans.

Moreover, in a marriage market where there is a scarcity of marriageable men, the marriageable men may use their relative scarcity to avoid marriage. Not surprisingly some studies find that the marriage market is not significant in accounting for marriage or cohabitation (see Eissa and Hoynes, 2000b:15).

Nevertheless, there is empirical evidence from the US that increasing men's employment does increase rates of marriage (Burstein, 2007:389; Bitler *et al.*, 2004:215). The related theory is that falls in levels of male unskilled employment and wage levels have created imbalances in marriage markets, especially amongst African Americans. However, Edin (2002) claims that whilst this view is persuasive, the fall in US marriages is greater than the theory would imply (see also Gennetian and Knox, 2003:9). Moreover, if in the US the fall in marriage rates amongst African American single mothers is due to increased unemployment amongst African American males, then welfare programmes that increase female employment

and earnings may not have large effects on marriage rates, indeed they might accelerate the decline in marriages (Gennetian and Knox, 2003:10).

The impact that the welfare system has on marriage is theoretically ambiguous. Some writers have used Becker's theory to suggest that increases in benefits and tax credits will decrease the chances of marriage. Tax credits and benefits are seen to influence behaviour because of the loss of income (the marriage penalty) a woman would incur being married compared to being single. Similarly, the emphasis on obtaining paid work in welfare systems might mean that greater financial independence for women leads to less marriage (the 'independence effect' (see previously)). However, the increasing emphasis in welfare systems on conditionality and work requirements might mean that remaining on benefit is less attractive and leads to more marriage. In the US, there is some evidence that women are less likely to enter marriage unless they have sufficient bargaining power in the form of their own income or self-sufficiency (Gennetian and Millar, 2004:279). This implies that increases in employment/earnings will promote marriage. Indeed, men may prefer to marry women who are financially independent. So, in practice, it is difficult to predict how increases or decreases in benefits and tax credits will affect marriage (and cohabitation). Moreover, the theory is unable to give clear predictions of the magnitude of the effect of the welfare system on marriage; it might be large or small.

Furthermore, it is possible that the economic factors influencing union formation and dissolution through the welfare system operate differently for low-income and high income groups (Burstein, 2007:389). This is because: benefits might have a larger impact upon individuals with lower incomes compared to the effect that tax credits have on people with 'higher' incomes; employment programmes usually increase the employment prospects and so the earnings of lower income males rather than those of better-off males (who tend to be in full employment); and for low-income women the benefits of employment in order to improve union stability (reduced financial pressures) may not exceed the costs (less leisure time) because their work is often poorly paid. Thus, indicators of (potential) income, such as educational attainment, may be used in empirical studies to explore associations between tax and transfer payments and family structure to see if economic mechanisms vary between disadvantaged and non-disadvantaged groups.

There is empirical research that shows that decisions to marry (or divorce) are not simply affected by the tax and transfer systems. Other factors that appear to influence such decisions include:

 Characteristics that affect the suitability of a person as a potential spouse (Carasso and Steuerle, 2005:161), including drug and alcohol abuse, mental health conditions (such as depression), fear of divorce and experience of domestic violence (Roberts, 2007:4-5). Men and women with low-incomes may have less access to services that help them to address these issues than people with higher incomes and so, it is argued, they avoid people with these characteristics.

- A related point is the degree of trust that can be placed in a potential partner.
   Changes in gender roles mean that women may be less willing to tolerate 'bad'
   behaviour by men (Ellwood and Jencks, 2004:20). For women this can be with
   respect to a male's to sexual faithfulness, money management skills and the
   nature of his interactions with her children (Edin, 2002; Carasso and Steuerle,
   2005:161). For some the level of risk might mean avoiding marriage and for
   others delaying marriage in order to observe and assess a potential spouse's
   behaviour.
- A potential spouse's suitability as a parent (Carasso and Steuerle, 2005:161). For single mothers this may in turn be influenced by the number and age(s) of her child(ren). In the US separated women with no children are more likely to remarry than women with three or more children (Bumpass *et al.*, 1990 cited in Gennetian and Knox, 2003:11). This might be because some men are unwilling to take on the additional responsibilities associated with marrying a woman with children, especially if he is not the biological father. Or he may be concerned about the effect of the marriage on any children he already has and he may have ongoing financial commitments to these children (Roberts, 2008:4). Unmarried mothers may also be worried about the effect that a marriage may have on her existing children, especially if she expects they will respond negatively (Roberts, 2008:3-4).
- The family context, in terms of parental education and the family's religion (Plotnick, 2004:7).
- For single people there is a possible cost of marriage in terms of forgoing possible gains from any investment in human capital (Plotnick, 2004:8). Thus higher levels of education can lead to delayed union formation (Aassve *et al.*, 2006:795).
- The desire for a fulfilling relationship (Carasso and Steuerle, 2005:161), and young people's expectations and desires about marriage (Plotnick, 2004). More recent cohorts are more likely to endorse gender equality and increasingly women expect to have an equal say in family decisions (Ellwood and Jencks, 2004:20-21). The level of autonomy or control desired by single women may convince them that they do not wish to become dependent upon men and that they want to live alone (Edin, 2002; Gennetian and Knox, 2003:8). Or it may mean that women postpone marriage until their children are in school when they could then enter employment and through making a financial contribution to the household be entitled to have a say in family decision-making. In any event in the UK younger cohorts have a lower rate of union formation (marriage and cohabitation) than older cohorts (Aassve et al., 2006:795).

- Having financial stability in order to minimise the stress that would otherwise characterise low-income households (Edin, 2002). Low-income mothers have to worry about money because they have so little, and erratic financial contributions from males can mean that couples separate or do not marry. In the US low-income mothers can impose a 'pay and stay' rule if male partners are out of work and not making a financial contribution to the household they can lose their right to co-reside.
- Wider socio-cultural norms about the need for substantial assets and income prior to marriage (Burstein, 2007:396).
- Individuals in the UK with a limiting long-term health condition have been less likely to establish a first or subsequent relationship than those without a health condition (Clarke and McKay, 2008).
- Race and ethnicity (see previously).
- Educational attainment which in turn will affect potential earnings (see previously).

There are other approaches to family formation and dissolution in the economics literature (see Bergstrom, 1996). For instance, cooperative and non-cooperative bargaining models, which build upon game theory, acknowledge that the members of a 'family unit' may have different objectives (Bergstrom, 1996:1924- 1930; Ribar, 2003:6). Thus how married and unmarried parents respond to financial incentives in the welfare system may differ.

## 1.5.3 Cohabitation and welfare systems

Many of the economic factors promoting marriage also favour cohabitation. Indeed, for low-income groups the benefits of marriage over cohabitation may be small, as the latter can provide some of the benefits of marriage such as companionship and economies of scale in household production. Cohabitation can be conceptualised as a substitute for marriage or as a complement to marriage (see Moffitt *et al.*, 1998:260). If it complements marriage, cohabitation can be seen as a precursor to marriage. Partners can use this trial period to gather information on the suitability and quality of the union. Cohabitation may also be attractive for single mothers as it provides a degree of flexibility during periods of financial instability and uncertainty (Edin and Lein, 1997) cited in Harknett and Gennetian, 2003:458). However, cohabitation is not necessarily a perfect substitute for marriage (Gassman-Pines and Yoshikawa, 2006:12; Moffitt *et al.*, 1998:260):

- It is not a good predictor of a successful marriage; in many countries former cohabiters have higher rates of divorce than non-cohabiters (Svarer, 2004).<sup>13</sup> Indeed, some cohabiters claim they have no intention of marrying their partner.
- What is distinctive about marriage is a commitment that is backed with some legal rights and its contractual nature may reduce risks and encourage investment in human capital (Burstein, 2007:391).
- Some people give reasons for cohabitating, for example, sharing household expenses, that imply the arrangement is not a trial for marriage.
- Cohabiting unions may be less stable than married unions (see ahead and Section 1.4).

According to economic theory, to the extent that benefits are reduced by taking into account the additional income of a partner then there is a disincentive to cohabit. However, some argue that single parents may have an incentive to cohabit rather than marry if it is possible to conceal the additional income from a partner from the authorities.

Financial considerations can be an important consideration for cohabiters when deciding whether to marry (Smock *et al.*, 2005). Specifically, male earnings, occupation and educational level are positively associated with transitions from cohabitation into marriage. In the UK union formation is influenced by male economic opportunities. Male unemployment is negatively associated with union formation—high levels of unemployment restrict women's partnership options. The influence of women's earnings and employment is more ambiguous. Qualitative research (conducted in the US) suggests that the main economic issues that can affect the transition from cohabitation to marriage are (Smock *et al.*, 2005:687-692):

- 'having enough money' marriage requires a certain level of financial stability, cohabitating couples are unlikely to marry if they are struggling financially;
- being able to pay for a 'real' wedding the cost of a wedding can be an impediment to marriage;
- having achieved a set of financial goals before the marriage, for example, buying a house or a car and completing education;
- whether the male partner earns enough to be an economic provider; and
- lacking money can lead to stress and relationship conflict which in turn adversely affect the quality of relationships.

This is not a universal finding – Denmark is an exception, where cohabiters have a lower risk of marital dissolution, possibly because there is little social stigma associated with cohabitation couples are able to use it as a 'trial' for marriage (Svarer, 2004:533).

A cohort analysis of British single parents over the period 1991 to 2001 shows that the 1991 lone parents who were married or cohabitating by 2001 tended to be (Marsh and Vegeris, 2004:35-37):

- younger (and age had the largest effect controlling for other factors);
- those in 1991 who had been never partnered or who had been married were more likely to be married by 2001 than those who left a cohabitating relationship in 1991;
- those with a new child since 1991; and
- those working 16 hours or more a week in 2001.

Moreover, the factor that influenced whether they married rather than cohabited was whether the 1991 single parent had a new child. Getting a new partner (and paid work) was often associated with being on an 'upward and improving path' (Marsh and Vegeris, 2004). Many became dual-earner families, although for younger single parents entering a new union could be accompanied by having a baby and a temporary exit from employment.

Union formation (marriage and cohabitation) in the UK can be influenced by the birth of a child, specifically birth order:

'...experiencing a first birth has a strong positive effect on forming a union, and this is so for both genders. However, having a second birth outside a union actually lowers the rate of union formation. The positive effect of the first birth event is consistent with economic theory, in that individuals consider a cohabiting union or a marriage to be more beneficial once they have acquired marital-specific capital. However, there might also be normative forces at play, in the sense that individuals might feel a pressure to 'legitimize' the child. The negative sign of the second birth event indicates that those who do not form a union after the first birth are at a disadvantage in the marriage market when they have the second child. There might also be a selection effect here: individuals who are willing to have one child outside a union might be more willing to repeat the experience. The subsequent birth events have no significant effect on union formation.'

(Aassve et al. (2006:796))

## 1.5.4 Union dissolution and welfare systems

The Beckerian model of marriage is also influential within the literature on separation and divorce. It posits that people divorce/separate when the benefits minus costs of becoming single exceed those of remaining married/cohabitating.

Some argue that welfare benefits contribute to divorce – fathers are seen as prepared to leave their spouses (and children) knowing that the State will provide adequate financial support (Zimmerman, 1991:139). Indeed, it is claimed that women might agree to this arrangement since it provides them with enhanced

economic security and stability. However, research shows that there is a wide range of other factors that influence divorce/separation (and union) decisions including (Dickert-Conlin, 1999; Harknett and Gennetian, 2003):

- The earnings potential of the spouses (Ono, 1998): There is some evidence that women's employment and earnings increase marital stability, although other studies show a decrease (Burstein, 2007:407). Indeed, the outcome is theoretically ambiguous, see Ono, 1998:
  - 'the stability effect' posits that if the spouse has a high earnings potential this will make the person more attractive as a partner and/or by reducing financial hardships lessens the likelihood of relationship conflict and so reduces the risk of separation. Unions where the male is in full-time employment and/or has high earnings are more stable because the couple have a higher standard of living (Burstein, 2007:404) known as the husband's income hypothesis (Ono, 1998). Similarly, increases in women's earnings may reduce levels of financial distress and so reduce the risk of dissolution; but
  - 'the independence effect' suggests that a person with high earnings is less in need of a partner and hence if already in a partnership separation is more likely. Typically, this independence effect is seen as applying to women a wife's resources are positively related to marital dissolution. Any increase in the wife's resources must, however, be sufficient for her to live separately from her husband (Ono, 1998:686). Alternatively, employment can increase the stress on partners and impose time constraints, which in turn increase the likelihood of union dissolution.
- Presence of children may reduce the likelihood of separation because they represent an emotional and time investment in a marriage. However, in households with young children and mothers with specific impairments (for instance, depression) there is a higher risk of union dissolution (Clarke and McKay, 2008). Moreover, two-parent families with a disabled child aged one to two years are at greater risk of becoming single parent households.
- The duration of the marriage: Those married for longer are less likely to separate.
- Age at marriage: Women who marry early are more likely to separate from their husbands.
- In the UK younger cohorts are more likely to dissolve a union than other cohorts (Aassve *et al.*, 2006:795).
- The presence and stability of an impairment: Households where adults have no impairment or where the experience of the impairment was stable are less likely to dissolve their union (Clarke and McKay, 2008).
- The existence of a prenuptial agreement may affect the likelihood of a divorce, although there is disagreement as to whether it increases or decreases the likelihood of marital dissolution (Rainer, 2007:338).

- The liberalisation of divorce laws is seen by some as making divorce easier (Rainer, 2007).
- Educational attainment: however, in the UK there is no evidence that level of education has an affect on union dissolution (Aassve et al., 2006:795).

## 1.5.5 Fertility and single motherhood and welfare systems

Decisions on fertility and single motherhood are complex decisions influenced by a wide range of cultural, economic, legal, political, social and religious factors (Kearney, 2004: 296; Maynard *et al.*, 1998:152-153). However, the nature of the interaction between the various factors is still not fully understood.

The view that there is a systematic interplay between economic considerations and fertility dates back to Malthus (1798, cited by Clarke and Strauss (1998)). The classical theory of population was summarised by Blaug (1978):

'...the production of children, [is] not a means of spending income on 'consumer goods' to acquire satisfaction, but as a method of investment in 'capital goods' for the sake of a future return.'

(Blaug, (1978:78, cited by Clarke and Strauss, 1998))

The work of Becker is, again, influential in the literature with decisions on single parenthood and childbearing seen as being affected by the benefits and costs of the options available to women. The underlying economic model assumes that marriage will immediately lead to parenthood (Ellwood and Jencks, 2004:14), with children regarded as consumption rather than investment goods (Clarke and Strauss, 1998). A basic economic model of fertility would therefore predict that the opportunity cost of bearing and raising children affects parental decisions about starting families and increasing family size. Thus, an increase in family income would increase fertility, at least over short periods in which the cost of human capital does not change significantly (Baughman and Dickert-Conlin, 2007). However, Clarke and Strauss (1998) suggest that:

'...for individuals in poverty, various public cash and in-kind transfers create a series of economic incentives which...make the childbearing decision equivalent to the Malthusian analysis that children are income-producing assets as well as sources of utility. In the modern welfare state, it is the transfer system...that creates income-producing opportunities.'

((Clarke and Strauss, 1998:825)

Yet this 'traditional' economic model has relatively little to say about non-marital births and the 'decoupling' of marriage and parenthood (Ellwood and Jencks, 2004:19).

There are two broad approaches in the literature on the effects of economic factors on non-marital births (Del Bono, 2004:2). The first highlights the role of welfare systems in 'making single motherhood more economically attractive' and is particularly associated with Charles Murray (1984). The second approach

focuses on men and women's economic opportunities, including the availability of marriageable men, and male and female unemployment and potential earnings.

The literature suggests that welfare benefits promote both single parenthood and non-marital births. So, for instance, benefits and tax credits could be expected to generate incentives for single childless women to have children, as entitlement may require having a dependent child. AFDC in particular has been seen as providing a financial incentive leading to a higher incidence of single parent families (Moffitt, 2003:332); although the welfare effect is complicated by the programme's eligibility rules which mean that AFDC '...does not discourage marriage or cohabitation universally but only if it is with the male who is the children's actual father.' (Moffitt, 2003:333). However, and as already noted in Section 1.4.3, the real value of AFDC declined during the 1970s and 1980s, when female headship rates were increasing, so undermining the argument that there is a simple connection between welfare benefits and single parenthood. Moreover, the impact of benefits on single parenthood is theoretically ambiguous. If marriage entry is voluntary and its dissolution a random event then AFDC is effectively a form of insurance that ought to encourage individuals to marry, as it reduces the risks to the woman relative to what they would have been in the absence of the benefit (Moffitt, 1998b:53 and 2003:332-333). In practice this 'insurance role' is unlikely to have a significant impact because many women become single mothers before marriage and the moral hazard problem is potentially large (that is, those in receipt of AFDC have an incentive to engage in behaviour that might lead to parenthood).

It is also posited that receipt of welfare benefits will lead women to have larger families than they might otherwise have (Presser and Salsberg, 1975:227; Gauthier and Hatzius, 1997:295-6). Indeed, that additional benefits for each dependant child reduces the direct and indirect costs to families of having children. (An alternative view is that benefits may alter the timing of childbearing – encouraging early entry to motherhood – but not family size.) It is this economic theory that has been used to justify the imposition of family caps (Maynard et al., 1998:150). The cap increases the recipients' net economic cost of childbearing (by an amount equal to the benefit increase that previously would have been paid), and, ceteris paribus, ought to discourage extra childbearing. However, Maynard et al. (1998:151) argue that the effect of family caps on fertility is likely to be small. First, the resulting income changes were small both relative to the cost of having children and in absolute money terms. In addition, the reductions could be offset by increases in Food Stamps and housing subsidies; and access to Medicaid was unaffected by family caps. Secondly, non-economic factors were likely to be more influential on fertility decisions than economic incentives. Moreover, pregnancies can be unintentional, and by implication not susceptible to economic factors.

Other economic explanations focus on economic opportunities, notably male unemployment and employment. The notion of a marriage market is influential (see Section 1.6.2), and that in poor neighbourhoods (notably inner city areas) the increase in single mothers was due to a lack 'marriageable men'. Willis (1999)

argues that unmarried women are more likely to have children if there is an excess supply of women and the benefits of marriage are limited by available males having low-incomes. On the other hand, if there are marriageable men this can lead to increases in the marriage rate and the fertility rate, which in turn increases the pool of women at risk of becoming single mothers due to divorce and separation. Empirically, however, the marriage market thesis only accounts for a small part of the decline in marriage rates amongst African Americans (see previously and Del Bono, 2004:3).

Other writers have adopted a broader perspective and examined the affect of local labour market factors on cohabitation and childbearing decisions. These studies emphasise not only the influence of male employment and unemployment on female marriage and fertility rates, but also the opportunity cost to women of childbearing. The expectation is that increases in male employment within an area will lead to an increase in marriage rates, but a fall in fertility.

The effect of female wages on single parenthood is theoretically ambiguous. Improved economic opportunities for women, via higher incomes, could enable them to support their dependant children on their own (an 'independence effect'). However, higher female wages might discourage women from childbearing (a 'substitute effect'), because the perceived 'opportunity cost' of a birth is higher – women will have to forgo some potential earnings and will have less leisure time. It is also possible to argue that a lack of labour market opportunities may increase the attractiveness of single motherhood, because in these circumstances its perceived 'opportunity costs' are reduced.

In the UK, predicted female wages are negatively associated and male unemployment positively associated with non-marital childbearing (Del Bonon, 2004:11-14). That is, the better the economic prospects for women the lower the likelihood of non-marital births. Thus, women with higher levels of education or who are in employment have lower rates of childbearing (Aassve *et al.*, 2006:795-796). Similarly, college educated women in the US are more likely to delay childbearing than those with less schooling (Ellwood and Jencks, 2004:28). This might be because the former, faced with a satisfying alternative, choose to delay having children, or in order to maximise life-long earnings they have to invest in their careers early on and this necessitates delaying childbearing. These findings for predicted wages suggest that there is an opportunity cost to childbearing.

Other factors influencing fertility and single parenthood include:

- Cultural and social norms: For instance, in the US teenagers from immigrant families that are less well socially integrated into society hold more traditional views about non-marital childbearing (Plotnick, 2004:28).
- Being in a union: In the UK, having children is positively associated with being in a union (especially for men) (Aassve *et al.*, 2006:796). Moreover, for UK single parents having additional children is positively associated with having a new partner and being younger (aged under 30 years) (Marsh and Vegeris, 2004:41-43).

- Advances in reproductive technologies: Women have increased control over their reproductive choices and this, arguably, has led to behavioural changes and an increased tolerance for cohabitation (Ellwood and Jencks, 2004:25-26; Roberts, 2007:2).
- Levels of self-esteem and confidence: For young women, high levels of self-esteem are less likely to be associated with non-marital childbearing (Plotnick, 2004:9).
- Family background, including religious upbringing (Plotnick, 2004).
- Race and ethnicity (Plotnick, 2004).
- In the UK younger disabled adults (aged 20-34) were more likely to have dependent children than their peers (Clarke and McKay, 2008). However, older disabled people (past mid-30s) were less likely to have dependent children.

Some writers, especially in the US, have attributed the increase in single parenthood amongst low-income women to their values and attitudes. Indeed, nationally representative survey data shows that AFDC benefit recipients compared to other women have held statistically significant different familial values and attitudes towards marriage and childbearing (p<0.001) (Mauldon *et al.*, 2002:2). However, Mauldon *et al.* (2002:2-3) argue that the differences between the two groups of women are small and can be attributed to factors other than welfare receipt. When controlling for demographic characteristics the difference for views on childbearing disappears. Suggesting:

'...that the reasons these women became single parents may have more to do with their social circumstances and economic conditions than with their fundamental preferences.'

(Mauldon et al. (2002:1))

Indeed, most female recipients at risk of further unmarried childbearing aspired to marriage (Mauldon *et al.*, 2002:3).

## 1.6 Methodological issues

Within the reviewed literature there are three broad methodological strategies:

- non-experimental;
- experimental, and
- quasi-experimental.

In terms of the volume of evidence there are considerably more published analyses using non-experimental methods.

## 1.6.1 Non-experimental methods

Within the non-experimental literature there is a further distinction between those

that seek to establish the magnitude of any welfare effect on family types, but do not explore whether actual behavioural changes as a consequence (referred to here as *scenario modelling*); and those that use empirical data to investigate the relationship between taxes and transfers and **actual** behavioural responses.

### Scenario modelling

For the literature reviewed in this study, the *scenario modelling* approach is conducted only for estimating union/marriage bonuses and penalties. (A bonus (penalty) occurs when net income (from work and benefits) for a married/cohabitating couple is greater than (less than) for two single people.) Typically, studies adopting this approach find that there is a financial disincentive for single parents to marry – but (by definition) do not provide evidence of actual behavioural changes. Studies use either (a) 'stylised' or prototype families to estimate typical union bonuses or penalties, or (b) survey or administrative data to simulate outcomes. A weakness of using 'stylised' families is that they cannot do 'justice to the wide variation in family circumstances, and does not provide any guide to the relative importance of each effect.' (O'Donoghue and Sutherland, 1999:566).

Studies adopting this *scenario modelling* approach tend to impute union bonuses and penalties by 'divorcing' married couples, rather than 'marrying' single people, and calculate the difference between what they would receive in benefits/tax credits as married couples and as two unmarried individuals. As a consequence the studies may be better at estimating the potential marriage bonuses and penalties of those who have chosen to marry rather than remain single, as the latter may differ from the former in a number of important ways (for instance, in income and number of children).

Whichever approach is adopted, the comparator for estimating the bonus/penalty is unobserved, and the calculations are sensitive to the assumptions made about living arrangements and dependents (Eissa and Hoynes, 2000:690; Holtzblatt and Rebelein, 2000:1131). In calculating partnership bonuses and penalties researchers can, for instance, assume that benefit take-up is 100 per cent, and ignore housing benefits (see, for example, Ellwood (2000b: 1065-1068)). Critically, when 'splitting' married couples, the analysis tends to assume that the employment patterns and earnings of the now separated couple are unaffected. Yet people's behaviour, including their labour supply, does often change following a divorce – the assumption of no change is essentially unrealistic (Burstein, 2007:410; Ellwood, 2000b:1089 – Anderberg (2007) and Rosenbaum (2000) provide examples of where these assumptions are relaxed). For this reason some analysts use longitudinal data in order to estimate incentives and disincentives following a change in marital status.

The estimates of marriage bonuses and penalties also tend to ignore the economies of scale from running one household rather than two.

Furthermore, the estimates tend to assume that people are aware of the incentive structures they face. However, benefit and tax systems can be relatively complex

with individuals having to estimate interactions between multiple benefits and tax credits. Something which the analysts' own models can have difficulty achieving. According to Ellwood (200b:1073) models often have problems capturing the administrative arrangements that interact with benefit and tax systems. For instance, the models do not allow for variations across states in, say, sanctions regimes or the requirements claimants must meet before they can register for benefit under TANF. This omission will bias the estimates obtained.

In addition, it is unlikely that partners 're-optimise each period', that is, re-determine their marital status each year. There are (fixed) costs for entering and leaving marriage that will constrain some people's ability to simply respond to financial incentives.

Moreover, the interpretation of the analysis is made more complex if incentives and disincentives do affect behaviour. For instance, if survey data are used to estimate union bonuses and penalties and the welfare system affected behaviour, then a sample of married couples (to be 'divorced' in the calculations) would over-represent families with large bonuses and under-represent those with large penalties (Hoffman and Seidman, 2003:87).

## Empirical studies of demographic responses using non-experimental methods

In terms of the second *behavioural* non-experimental approach that focuses on actual behavioural responses to welfare systems, there is a sizable body of mainly US research that explores the impact of the benefit and tax system on family structure. These studies require that similar family types receive different levels of benefit and, in order to see if variations in welfare affect demographic behaviour, US researchers are able to take advantage of the variation in levels of benefit (AFDC and TANF) between states. Researchers have used the following non-experimental approaches to identify the impact of welfare benefits on demographic behaviour (Hoynes, 1997a:92-94; Moffitt, 1998b:57-59; Moffitt, 2003:336):

- Cross-sectional analyses of data for a point in time. Regression analysis is used to examine how, say, marriage or fertility varies by differences in the level of benefits between states, individual characteristics, and, less often, other state characteristics. However, the estimated impact of the welfare system will be biased if there are any omitted state factors from the dataset, such as social norms or religious characteristics, that are correlated with both family structure decisions or (through voter preferences) state welfare policy (Ellwood and Bane, 1985, cited in Hoynes, 1997a:91; Hoynes, 1997b:129; Moffitt, 1994; Ribar, 2003:1). This is addressed by the next approach.
- Analyses using pooled cross-sectional data or panel data for more than one point in time. Using these data with regression analyses, researchers can control for unobserved variables. Known as 'state fixed effect' models, they can be used to capture unobserved factors common to all residents of a state, such as State divorce laws or support services. Their advantage is that the analyst does

not have to specify exactly which variables are missing, only the structure of the omitted variable (Ribar, 2003:19). Most fixed effect models reviewed here are for US States, but they can be developed for other geographical areas – counties, regions or countries.

Compared to cross-section designs, the inclusion of area fixed effects can reduce the significance and magnitude, even the direction, of any welfare effect on family structure (see, for instance, Moffitt, 1994). For example, Lichter *et al.* (1997:126-128) report that a \$100 change in welfare payments is associated with 0.817 percentage point change in female headship when there are no controls for unobserved state level variables (p<0.01). However, the inclusion of state fixed effects reduced this impact by more than 50 per cent to 0.392 percentage points (p<0.01).

Models with State fixed effects also tend to include year fixed effects. These control for average changes in, say, marriage or divorce within a given year that are common to all areas (say, States).

- The use of panel data enables the identification of unobserved individual effects (Hoynes, 1997a:91). To the extent that studies do not control for individual effects, but only State effects, they may be biased through inter-State migration and sample attrition and entry. The individual effects models capture time-invariant unobserved variables at the level of the individual.
- Time series analyses have been used. However, a number of US researchers (for instance, Hoynes, 1997b:125-126) have observed that if financial incentives affect family formation then the observed fall in the real value of welfare benefits (especially in AFDC since the mid-1960s) should be associated with declines in female headship and non-marital births, but instead they have continued to increase. Suggesting that factors other than financial incentives are operating to influence demographic behaviour. Yet, controlling for all the possible factors that change over time is very difficult.

In summary, the main non-experimental methods used in the US seek to investigate whether there is a welfare effect on family structure by using variations in benefits between individuals within states, between states and between states over time (Hoynes, 1997b:124; Moffitt, 1998b:57). These non-experimental methods are discussed further in Appendix A. However, and not wishing to prejudge the review, it appears which source of variation and controls for unobserved heterogeneity are used affects the reported significance and magnitude of the welfare effect on demographic behaviour.

## 1.6.2 Experimental methods

There has been some use of social experiments to assess the impact of welfare reforms on family structure. Random assignment designs do overcome one of the potential problems with non-experimental methods – having to control for unobserved variables. Social experiments can also help to establish causation between variables (see Stafford, 2002).

However, social experiments may not be ideal for assessing welfare effects on family structure (Hoynes, 1997b:133; Moffitt, 2003:348; Maynard et al., 1998:162-165). This is because the programmes to be tested have a wider impact on the community, possibly changing expectations and attitudes and thus the behaviour of the control group. This 'contamination' of the control group could bias the experimental results. (However, this might be overcome if the unit of analysis was the community. That is, if there was a random assignment of communities to those that received an intervention and those that did not, as opposed to the random assignment of individuals within a given community.)

Moreover, the designs of the demonstration projects tend not to isolate particular reforms so that impacts of specific measures on demographic behaviour can be identified (Maynard et al., 1998:157-162). Instead packages of reforms are evaluated, making it difficult to determine the effects on demographic behaviour of particular programme elements. Camasso (2004:453) points out that a package of reforms is likely to have a greater impact than each of the components separately. Thus experiments '...have not been very successful in isolating which broad components of reform...influence recipient outcomes' (Camasso (2004:453); and see Moffitt and Ver Ploeg, 2001; US General Accounting Office, 2001). The challenge facing policy evaluators, according to Camasso (2004:454), is that, unlike some earlier programmes which introduced varying levels in phased implementations, more recent programmes have introduced components simultaneously to all recipients in the treatment group. According to Camasso, this has resulted in 'black box' evaluations which are only able to examine the impact of the bundle rather than its component parts (Camasso, 2004:454; and see Greenberg and Shroder, 1997; Wiseman, 1993). Fien (1994) has identified four strategies to evaluate 'reform bundles'; as summarised by Camasso (2004:453) these are:

- examine customer exposure to each provision separately;
- identify the unique goals of each component;
- conduct cross-state comparison where components differ between states; and
- design multiple treatment experiments.

The particular context of these experiments also means that their findings do not necessarily apply to other contexts – for example, that findings from low benefit areas apply to high benefit areas. Any findings from evaluations incorporating interventions that are meant to affect family structure tend to be site-specific and hence cannot be readily generalised to other geographic areas. In addition, even if the design was robust, the outcomes reviewed here – marriage, childbearing and divorce – could take several years to respond to policy interventions and so any experiment would have to be administered for a long period of time (Moffitt, 1998b:56).

## 1.6.3 Quasi-experimental methods

A number of studies also incorporate difference-in-difference designs; in particular UK studies use this approach to explore the impact of WFTC on demographic behaviour. These allow for impacts before and after a policy reform to be investigated. They involve comparing demographic outcomes for a group affected by a policy or programme with one unaffected by the intervention both before and after the change. The use of a comparison or control group means that unobserved variables that change over time are to some extent controlled for in the analysis.

One issue is the selection of an appropriate control group, for instance, single women without children are often used as the control for single parents. However, women without children may not be a good comparator – their trends in, say, union formation may be very different from women with children. Similarly, married women may not be a good comparator for single women, due to selection effects. Women who marry typically have lower earnings than those that do not and this might be because they would be at a relative disadvantage in the labour market and so '...concentrate more of their efforts towards finding a spouse who likes work or has a comparative advantage in the labor market.' (Rosenbaum, 2000:9).

Moreover, if the impacts of the welfare system on family structure are both positive for some groups of individuals but negative for others, then difference-in-difference may not capture these effects, because they are 'averaged out'. In addition, if the intervention being assessed is a 'package' of different proposals, difference-in-difference cannot precisely disentangle the effect of just one policy/ programme.

## 1.6.4 Other methodological issues

#### Tax credits and benefits

In general, the studies reviewed here consider the impact of the benefit system or the tax system on family structure. However, the two systems for low-income families may interact and as a consequence estimates based on only benefits or only tax credits may be biased. Dickert-Conlin (1999:232) (see also Dickert-Conlin and Houser, 1998) looks at the impact of the US tax and transfer system on separation decisions and, following sensitivity analysis, finds that analysts ought to consider both as:

'...earlier work on the effect of taxes or transfers in isolation obtained results that were biased towards zero.'

## Longitudinal data

In some instances, non-experimental and experimental studies use longitudinal data in the form of repeated cross-sections. However, analyses based on such data adopt a 'stock approach' through examining states at given points in time (Anderberg, 2007:21). An alternative approach is to use panel data, which allows a 'flow approach' whereby transitions into, say marriage, can be explored. The advantage of examining flows is that the stock might not be typical of, for instance, married women divorcing, as those becoming divorced are more likely to have been married for a shorter period of time than the stock of all married women.

## Lags in response

Several studies seek to examine the association between current welfare provision and demographic outcomes. However, there may be a considerable lag between changes in transfer and tax programmes and demographic responses, especially if family-related decisions are influenced by social norms which might take time to evolve. The extent to which the literature takes into account this possible lag is limited (Hoynes, 1997b:126).

## 1.7 Structure of report

The next three chapters review in turn the impact of welfare systems on partnering, union dissolution and childbearing and single parenthood. Chapter 5 considers the implications of the reviewed literature for the UK and some conclusions are drawn in Chapter 6.

In Chapters 2, 3 and 4 some of the more 'technical' details about individual studies are placed in text boxes. Readers focusing on substantive findings can skip over these boxes, whilst those seeking more information about the data and analytical methods used should read the text boxes.

As the impacts of the welfare system on family structure are often reported in the literature to two decimal places the effects are expressed in subsequent chapters as numbers and not, as is customary in this report series for numbers under ten, in words.

# 2 Partnering

## **Summary**

This chapter considers the evidence on whether, as some commentators argue, welfare systems discourage marriage. A union/marriage bonus (penalty) occurs when a couple receive more (or less) in benefit/tax payments than they would if they were single. For union/marriage bonuses or penalties to exist there are two pre-conditions: the unit of assessment is the family (rather than the individual) and different marginal tax rates apply at different income levels. Trade-offs in policy objectives mean that a benefit and '...tax system cannot be simultaneously progressive, treat family as unit of taxation and be neutral with respect to marriage.' (Hotz and Scholz, 2003:163).

## United Kingdom

Two United Kingdom (UK) studies were reviewed. One study finds that single parents who were Family Credit (FC) recipients in 1991 were more likely to have found a partner by 2001 than were non-recipients. However, the second study reports that following the introduction of Working Families Tax Credit (WFTC), single parents were less likely (by 2.4 percentage points) to form a union than single childless women. This impact was more pronounced for women with younger children (two to three percentage points). The difference in the direction of the impacts might be because WFTC was more generous than FC. Effectively, WFTC may have reduced the opportunity cost of being a single parent.

## Organisation for Economic Cooperation and Development

A study of the impact of social security (payroll) taxes for defined-benefit, Pay As You Go (PAYG) pension schemes on family formation across the Organisation for Economic Cooperation and Development (OECD) finds that social security payroll taxes slightly reduce net marriage rates. However, a number of simplifying assumptions are made which limit the usefulness of this analysis.

Continued

#### United States

Compared to the UK, there is a more extensive United States (US) literature; although it is focused on single parents. The US literature on the impact of welfare systems on union formation presents mixed findings – there are studies showing increases, decreases and no significant effects.

A review of earlier research by Moffitt (1998b) shows that some studies found a marriage penalty; a significant minority found no effect; but all those studies with statistically significant results show that Aid to Families with Dependent Children (AFDC) had a negative effect on marriage. His review, together with other research, also highlights that impacts in the US vary by race and gender.

Studies of US welfare reform (that is, AFDC waivers and Temporary Assistance for Needy Families (TANF)) using non-experimental approaches also report mixed findings, whilst a meta analysis of six experimental programmes reveals no consistent relationship between welfare programmes and union formation (Gennetian and Knox, 2003).

There is, however, limited evidence from the Minnesota Family Investment Program (MFIP) and New Hope projects that in the longer-term welfare systems may increase the stability of marriage.

The impact of Earned Income Tax Credit (EITC) on actual marriage decisions is small and statistically insignificant – there is no 'real evidence that EITC marriage penalties were reducing marriage.' (Ellwood, 2000b:1100).

#### Canada

The Self-Sufficiency Project (SSP) (a wage supplementation programme) highlights the importance of local context. The programme was implemented in two culturally diverse provinces and found that it decreased the probability of being ever married or cohabitating in British Columbia by 2.5 percentage points (a 16 per cent fall); but increased the probability of being ever married or cohabitating in New Brunswick by 4.3 percentage points (a 22 per cent rise).

A second Canadian study on the removal of marriage penalties for surviving spouses in the pension system showed that remarriage rates for females aged under 65 and for males aged 45 to 59 increased significantly. However, there are outstanding queries about the quality of the data used.

#### Overall

The studies demonstrate that any welfare effect on union formation varies by sub-group and they can also be lagged. US and Canadian studies also highlight that findings vary by local context. Data problems also undermine the robustness of some studies.

Tentatively, this review finds that whilst there are some studies showing an effect on marriage/cohabitation, there is in general no strong evidence for a large and significant welfare effect on union formation – the evidence is inconclusive.

## 2.1 Introduction

Do welfare systems encourage or discourage marriage? Some UK commentators, for example, Campbell and Roberts (2002), Civitas (2002) and Morgan (2007) have – like many US researchers and policy makers – expressed the concern that there may be a 'marriage penalty' (as opposed to a 'marriage bonus') in the benefit and tax credit system. So, for instance, did WFTC provide an incentive for single mothers not to marry because of the 'extra' resources it provided to single parent families?

Ensuring that the welfare system does not, at least, penalise marriage may be a policy concern, not simply because policy makers might wish to promote 'healthy marriages' as an end in itself, but also because some argue that child outcomes are better in two-parent families (Roberts, 2007:3). Marriage may also have other benefits for partners, such as a lower incidence of health problems (see Arnato, 2007:952; Gassman-Pines and Yoshikawa, 2006:11). Indeed, given that the family income of marriage women is higher than that for single women, some see (stable) marriages as a route out of poverty for low-income single women (Morgan, 2007:35).

This chapter briefly discusses union/marriage bonuses and penalties, then considers the published evidence for whether welfare systems affect union formation decisions for the UK, see Section 2.2.1, across the OECD, see Section 2.2.2, the US, see Section 2.2.3 and Canada, see Section 2.2.4.

## 2.1.1 Union/marriage bonuses and penalties

Tax and transfer systems can generate union/marriage bonuses and penalties.<sup>14</sup> In the tax system a marriage penalty or bonus arises when a change in marital status leads to a change in disposable income (Alm *et al.*, 1999:194; Carasso and Steuerle, 2005). A penalty occurs when net income (from work and/or benefits) for a couple is less than for two single people living apart. The penalty arises because the earnings of one spouse are taxed at a different rate because of marriage (Carasso and Steuerle, 2005:168). In general, marriage penalties are higher when couples' earnings are similar; and lower when one earns significantly more than the other (Carasso and Steuerle, 2005:168; Hotz and Scholz, 2003:163). As Alm *et al.* (1999:195) note:

'When people with similar earnings marry, their combined income pushes them into higher tax brackets than they face as singles, and they pay correspondingly higher income taxes with marriage. Conversely, the marriage of two people with very dissimilar earnings means that the individual with higher income moves into a lower marginal tax bracket as a result of the marriage, thereby reducing the combined tax burden of the two partners.'

The term 'union/marriage bonus and penalty' is used in this report rather than 'cohabitation bonus and penalty'.

The tax system can also generate union/marriage bonuses - when a couple receive more in tax payments than they would if they were single. Bonuses tend to occur when partners have dissimilar earnings or only one partner is in paid work (Hotz and Scholz, 2003:163). The benefit system, like the tax system, generates union/marriage bonuses and penalties. The complexity of the tax and benefit systems mean that individuals can face both marriage bonuses and penalties (Carasso and Steuerle, 2005:158).

In principle, any tax or transfer scheme can generate union/marriage penalties and/or bonuses. Two pre-conditions are required: first, the unit of assessment for taxes or benefits must be the family (rather than the individual), and secondly, different marginal tax rates apply at different income levels. So, as Carasso and Steuerle observe:

'If graduated taxes were accommodated by individual filing or if all income and transfers were taxed at a flat rate, there would be no marriage penalties.'

Carasso and Steuerle (2005:157)

That is, the only way to avoid union bonuses and penalties is to have an individualised tax and benefit system or a proportional joint tax system (with no tax allowances) (Brewer, 2007:218). As a consequence a benefit and '... tax system cannot be simultaneously progressive, treat family as unit of taxation and be neutral with respect to marriage.' (Hotz and Scholz, 2003:163; see also Holtzblatt and Rebelein, 2000:1110). By treating families with the same income equally (so achieving horizontal equity) and having a progressive tax system it is impossible to be marriage neutral (Eissa and Hoynes, 2000b:2-4; Rosenbaum, 2000:3).

Varying tax rates and the withdrawal of benefit as income rises can combine to produce high effective marginal rates of taxation and so union/marriage penalties. (The effective marginal rates of taxation is the proportion of the next pound (or dollar) of income that is lost due to benefit withdrawal or payment of income tax and National (Social) Insurance.) Effective marginal rates of taxation can be higher for low- to moderate-income families compared to higher income households. For low-income households with one or more earners high effective marginal tax rates can be the result of the interaction of a number of tax and benefit programmes. High effective marginal tax rates arise when policy makers pursue two policy objectives: vertical redistribution of income (giving more tax and benefits to those on lower incomes) and containment of benefit programme costs (by restricting entitlement to low-income groups). Thus union/marriage penalties in the transfer system can arise, not because policy makers wish to penalise marriage, but as a consequence of targeting benefits (Alm et al., 1999:198).

Union/marriage penalties and bonuses reside in the UK tax and transfer system. Although assessment in the personal income tax system is individually based, tax credits (like benefits) are assessed on joint income. Furthermore, whilst contribution tests for National Insurance benefits tend to be based on individual income (Adam et al., 2002:16), the expansion of means-tested benefits means that there has been

an expansion in joint assessment. Union/marriage penalties can arise because a single parent already in receipt of Income Support (IS) or a tax credit can experience a fall in benefit or tax credits due to the extra income brought to the family by any new partner. Union/marriage penalties can be relatively large; in the UK many two-parent married couples needed a joint income of £50,000 per annum in 2004/05 to avoid any marriage penalty (Beighton and Draper, 2007:12).

In the UK, for out-off-work benefits and tax credits a cohabiting couple will generally receive a lower income transfer payment if they declare they are living together than if they claim to live separately. This is because (Brewer and Shaw, 2006:9):

- out-of-work benefits and tax credits are assessed on the joint income of a couple; and
- entitlement to tax credits is based on the joint income of a family and not the number of adults in a family.

The main exceptions to this union/marriage penalty are as follows (Brewer and Shaw, 2006:9):

- couples claiming tax credits and the secondary carer has no income here it would not matter if the family claimed as a couple or as a single parent;
- couples with a joint income of less than £50,000 claiming tax credits and the primary carer has an income high enough to only have entitlement to the family element of CTC if s/he were to claim tax credits as a single parent;
- couples where the primary carer has an income in excess of £58,000 as they would have no entitlement to tax credits even if they claimed as a single parent; and
- couples claiming out-of-work benefits where the secondary carer has a low-income (less than £31.95 per week) or is in receipt of disability benefits.

FC and WFTC generated union/marriage bonuses and penalties because they were only available to families with children, see Section 1.4.2 for further details. The typical case for a partnership bonus was when the woman was not working and the man worked for a low wage (see first bullet above). This is because under FC and WFTC neither the woman nor the man were likely to be eligible for the credit if they separated; the woman because she was unlikely to work for 16 or more hours per week, and the man because he was unlikely to be the primary carer. In couples where the woman already worked full-time (more than 30 hours per week) the couple would never receive a tax credit partnership bonus, because the presence of the male could never increase their tax credit award. Under FC and WFTC a partnership penalty was likely if the woman in a couple was the primary carer and worked for 16 hours or more per week (and consequently could receive the credit as a single parent) but they received a smaller credit as a couple due to the joint income test.

For WTC/CTC there are similar partnership bonuses and penalties (Anderberg *et al.*, 2008:5). In addition, childless couples face bonuses and penalties. Further details of partnership bonuses and penalties in UK tax credits and out-of-work benefits are provided in Appendix B.

Means-tested benefits, notably IS, mainly generate partnership penalties. For a couple where the woman does not work and the man works less than 16 hours per week, the IS partnership penalty is equal to what benefit the woman would be eligible for if she were single. Prior to the introduction of CTC the size of the penalty depended upon the number and ages of the children in the family. Under CTC, the child premia in IS were transferred to the new tax credit.

Whether these union bonuses and penalties actually affected marriage and cohabitation decisions is discussed in Section 2.2.

# 2.2 Research evidence on welfare systems and partnering

## 2.2.1 UK evidence on union formation and the welfare system

There are UK studies that estimate the size and distribution of union bonuses and penalties and who is affected, but do not provide evidence of any impacts on actual demographic behaviour (see Appendix B for two examples). Unfortunately, there is relatively little UK evidence on the effect of welfare systems on actual partnering decisions. A key UK study is by Francesconi and van der Klaauw (2007), who use British Household Panel Survey (BHPS) data for 1991 to 2001 to explore the impact of the WFTC on the formation of unions (marriage and cohabitation) by single mothers.<sup>15</sup>

The authors also examine the impact of WFTC on employment, use of childcare and fertility (see Section 4.2.1).

## Box 2.1 Further details - Francesconi and van der Klaauw (2007)

The BHPS data comprised unmarried, non-cohabiting females aged 16 or over, born after 1940 and living in England (Francesconi and van der Klaauw, 2007:8). The data excluded females who were in school full-time, or were long-term sick or disabled. The sample comprised 3,333 women, of whom 1,507 were lone parents and 1,826 were childless throughout the period.

They used linear probability models to estimate transitions into partnership by single childless women and single parents. A difference-in-difference approach is used with outcomes for single mothers compared with those for single childless women (the control group) before and after the introduction of WFTC. The use of panel data means that whether WFTC led to changes in the rate at which single mothers move into partnership can be estimated. To account for compositional changes over time, the analysis controls for certain socio-demographic variables (woman's age, number of children by age band (0-4; 5-10; and 11-18 years), ethnic origin, highest educational qualification, tenure, region, and the interactions between woman's age and number of children by age, age and educational qualification, and educational qualifications and number of children by age group. The models also include fixed effects (see Appendix A) to capture non-WFTC policy changes; the underlying assumption is that single mothers and single childless women responded in the same way to these other policy reforms, enabling the analysis to identify the net impact of WFTC. The use of panel data also allows the inclusion of individual fixed effects to account for changes in unobserved characteristics at the micro level.

The authors find that WFTC resulted in a 'substantial' reduction in the partnership rates of single mothers (Francesconi and van der Klaauw, 2007:25-26). Following WFTC single parents were on average 2.4 percentage points less likely to form a union than single childless women (p<0.05). This represented a 28 per cent decrease on the average annual rate of union formation by single mothers (8.5 per cent), which they view as a 'large effect'. Most of this effect was due to a reduction in partnership formation by mothers with one pre-school child (aged 0-4 years), who were 2.8 percentage points less likely to form a union than single childless women. However, if the single mothers had an older child there was no statistically significant effect. For mothers with two or more children partnership rates were also lower for those with pre-school children (2.5 percentage points) as well as for those with primary age school children (5-10 years) (1.5 percentage points). Therefore, the effect of WFTC on reducing the chances of union formation for single parents (compared to single childless women) was more pronounced for those with younger children.

Further analysis shows that if the comparison group is changed to those single childless women with low educational qualifications similar results are obtained (to the overall sample). (This comparison is undertaken because single parents tend to have low educational qualifications (Francesconi and van der Klaauw, 2007:8).) In addition, an analysis of women by age (those aged up to 30 and those older) revealed similar results – the fall in partnership rates affected all single mothers irrespective of their age.

However, these findings contrast with an earlier analysis of a 1991 cohort of lone parents (Marsh and Vegeris, 2004). An analysis of what factors were associated with a 1991 lone parent being married or cohabitating by 2001 showed that those claiming FC for at least 12 months between 1991 and 2001 were **more** likely to find a partner (44 per cent compared to 30 per cent) (Marsh and Vegeris, 2004:37). That is, FC did not always act as a financial disincentive to forming unions notwithstanding that dual-earner families could lose entitlement to the benefit.

'This is partly due to the fact that some of those who became single-earner couples claimed Family Credit as a couple and partly due to the fact that people rarely referred to the Family Credit rulebook when deciding who to live with. Becoming a couple whose earnings are above the qualifying level for Family Credit would be an incentive of itself.'

Marsh and Vegeris (2004:37)

In addition, the length of the duration claiming IS since 1991 was **not** significantly associated with entering a union by 2001.

The difference in the direction of the impact between these two studies might be due to WFTC being more generous than FC, see Section 1.4.2. Thus, FC may have not been a disincentive to union formation, but the more generous WFTC was. Effectively, WFTC may have reduced the opportunity cost of being a single parent.

The review found no studies on the impact of WTC on union formation.

## 2.2.2 Cross-country comparative studies on union formation

Not only may the payment of benefits affect union formation, but the contributions levied to finance benefits may also affect behaviour. Ehrlich and Kim (2007) investigate the impact of social security (payroll) taxes for defined-benefit, PAYG pension schemes on family formation and fertility across the OECD (the findings for the latter are discussed in Section 4.2.2). They use two approaches to assess this effect: a simulation model and regression analysis.

The simulation model uses OECD data for 1965 to 1989 and shows that a one per cent increase in defined-benefit, PAYG taxes **reduces** the probability of adult marriages by 0.231 per cent.<sup>16</sup> However, this simulation model incorporates a

Formally, the social security tax rate used is the old-age, survivor and disability-insurance proportion of social security benefits as a share of GDP (Ehrlich and Kim, 2007).

number of (simplifying) assumptions that limit its applicability to policy making. For example, the authors assume that births only occur within marriage, that only social security finances retirement, and that all adults pay the same taxes and receive the same defined benefits (and hence only children born to married parents pay social security contributions).

The regression analysis uses data for 28 OECD countries for the period 1960 to 1992, and shows that a one per cent increase in the pension portion of social security benefits as a percentage of GDP **reduces** the net marriage rate for those aged 15 and over by 0.369 per cent. Moreover, social security taxes for defined-benefit, PAYG schemes accounted for 51.5 per cent of the fall in net marriages (from 9.97 to 4.86) in OECD countries over this period (Ehrlich and Kim, 2007:18). The analysis controls for stage of economic development, other government taxes and benefits and for unobserved country-level variables.

Although Ehrlich and Kim (2007) identify a negative effect on union formation from social security taxes, their analyses, which are limited by the simplifying assumptions made, only find a small sized reduction in net marriages across the OECD.

## 2.2.3 US evidence on union formation and the welfare system

Compared to the UK there is a more extensive literature in the US on the impact of the welfare system on marriage/cohabitation. In part this reflects that some supporters of welfare reform in the US saw legislative changes, such as the Personal Responsibility and Work Responsibility Act 1996, as partly designed to increase the marriage rate (and to reduce the birth rate) of unmarried women.

As outlined in Appendix D, analysts have demonstrated that tax credits (EITC and child tax credit) and a wide range of means-tested transfer programmes (AFDC, TANF, food stamps, housing assistance, child care and Medicaid) include marriage penalties (and to a lesser extent marriage bonuses) (Carasso and Steuerle, 2005:158-9). Transfer programmes for low-income families in particular contain marriage penalties – benefits received before marriage can be reduced or even terminated on marriage. Moreover, under the tax system married and cohabiting couples are treated differently, but in the benefit system their treatment can vary – depending upon the father's biological relationship with the child, and how open the couple is about their relationship with welfare agencies (Carasso and Steuerle, 2005:160).

Traditional economic theory would suggest that AFDC should lead to a reduction in the marriage rate (see Section 1.5.2), whilst the policy intent behind AFDC waivers and TANF was that reform should lead to an increase in the marriage rate. Overall, however, the evidence from the US on the impacts of the welfare system on partnering is mixed. Some studies find that the tax and transfer systems can lead to increases in marriage (see Schoeni and Blank (2000) ahead), others report a decrease in marriage (see, for instance, Bitler *et al.* (2004) and Moffitt *et al.* (1998)) ahead) and others find small or insignificant effects (see, for instance,

Ellwood (2000b) ahead). As mentioned in Section 1.6.1, the significance and magnitude of estimated effects in non-experimental studies is sensitive to the specification of the analytical models – especially controls for unobserved variables (state and individual effects), which tend to reduce the size and significance of effects. Moreover, a meta-analysis of social experiments conducted by Gennetian and Knox (2003) (see ahead) suggests that, in general, welfare reforms had no impact on marriage or cohabitation rates.

This section considers the impact of welfare benefits (AFDC/AFDC-UP, AFDC waivers and TANF), tax credits (EITC) and an area initiative, the New Hope project, on union formation. For each of these three areas there is a general discussion, followed by an outline of the non-experiment and experimental findings as appropriate.

#### US welfare benefits and union formation

This review benefits from a major review of the early US literature on the impact of welfare systems on union formation (and childbearing) by Moffitt (1998b), see Appendix C<sup>17</sup>. He reviews 68 estimates from a variety of studies made available between 1970 and May 1996 and found that whilst some found a marriage penalty, a significant minority found **no** effect. However, all of the analyses with statistically significant results show that AFDC had a **negative** effect on marriage (Moffitt, 2003:336); a finding that is in-line with traditional economic theory on benefits and family formation.

Work by Moffitt and colleagues also shows that studies of the presumed anti-marriage bias of AFDC have been based on the belief that AFDC benefits are mainly (even solely) available to single mothers (Moffitt *et al.*, 1998:259). As a consequence models estimating the effects of the programme on marriage tend to assume that women face a simple choice: being unmarried and on AFDC or married and off welfare. In other words, early studies often ignore the possibility of cohabitation, or effectively estimate an impact on unions by including it with marriage.

An earlier study by Moffitt (1990) also illustrates the importance of producing separate estimates for the impact of AFDC on marriage for men and women by race. He finds, using Current Population Survey (CPS) data for 1985, that a more generous benefit system significantly reduced the likelihood of black American males being married, but had no significance for women or white males.

Post-1997 studies that show AFDC had a **negative** effect on marriage include Moffitt *et al.* (1998). They find that a \$100 increase in monthly benefit reduced the probability of marriage by 2.5 to 4.9 percentage points (the former estimate

Moffitt (1992) also conducted an even earlier review of the literature, which concludes that AFDC has some effect on marriage (and non-marital childbearing) but not enough to account for the increase in single mothers in the US during the 1980s.

is obtained using PSID data and the latter NLSY data; see Box 2.2 for further details of the study) (p<0.10) (Moffitt *et al.*, 1998:266-72). However, these results are for a model that included a limited number of demographic controls. When additional controls are included, for instance, for the marriage market and family background, the percentage point estimates change; the PSID estimate remains negative but is no longer statistically significant and the NLSY estimate is significant but reduces slightly in size to 4.5 percentage points. Thus this study provides only limited evidence of a negative effect of AFDC on marriage rates.

Relative to single motherhood, Moffitt *et al.* (1998) do show that the marriage rate was positively related to age, education and the presence of young children; and was lower for black women compared to non-black women. In addition, further analysis controlling for AFDC-UP shows, using the PSID data (but not the NLSY data), that this programme did encourage marriage. This contradicts other earlier studies that tend to find that the AFDC-UP programme had little or no positive effect on marriage rates.

Furthermore, Moffitt *et al.* (1998:266-72) find only weak evidence that AFDC rules provided incentives to cohabit. Specifically, they found that cohabitation rates compared to rates of single parenthood were not significantly related to the state level of AFDC, but were negatively related to education and were lower for black women compared to non-black women. However, a telephone survey of state AFDC agencies conducted in 1993 shows that in most states the programme's rules were relatively conducive to cohabitation (Moffitt *et al.*, 1998:264-5). In particular, the cohabitation of men unrelated to the children in the family was treated leniently—their income was not counted against the grant and contributions to the family were often ignored. These rules appear, according to Moffitt *et al.* to give a '... significant monetary encouragement to cohabitation while on AFDC ...' (1998:266). Indeed, having a lenient policy for in-kind contributions by unrelated contributors was positively associated with AFDC recipients' cohabitating; albeit the evidence of incentive effects is only 'weak'.

## Box 2.2 Further details - Moffitt et al. (1998)

Moffitt *et al.* (1998) developed multinomial logit models for partner status (cohabiting, married or neither) and for the joint decision of partner status and welfare status (on or off welfare, and cohabiting, married or single mother). The analysis used two data sets: the Panel Study of Income Dynamics (PSID) and the National Longitudinal Survey of Youth (NLSY). It was, nevertheless, a cross-sectional analysis. The PSID data were of all women aged 18-55 with children aged under 18 when interviewed in 1987 (1,258), whilst the NLSY data were of all women in the 1987 survey, who at that time were aged 22-29 with children aged under 18 (1,430).

For each dataset the authors give a 'small model' which includes controls for demographic variables (age, education, race and number of children) and an 'extended model' that incorporates additional controls for family background, the marriage market, religiosity and urban residency.

The extent of the analysis was constrained by the small number of cohabiting women in the two samples in receipt of AFDC (43 and 82, respectively). The authors also highlight that some of the results obtained are spurious or anomalous.

A more recent study by Blackburn (2000) illustrates how the impact of benefits on marriage can vary by race. He examines the association between the generosity of benefits and the probability of never-married mothers marrying. Using the NLSY for 1979 to 1992 he finds that for the non-black population of unmarried single mothers an increase in benefits is associated with a significant **fall** in the probability of marriage – a \$100 increase in monthly benefit reduced the probability of marriage by 19.5 per cent (Blackburn, 2000:129-131). Moreover, states with AFDC-UP had a higher likelihood of marriage.

However, for black never-married mothers higher benefits were associated with **increased** chances of marriage, and AFDC-UP reduced marriage probabilities. That, for instance, a \$100 increase in monthly benefit increased the probability of marriage by 19.8 per cent was not expected nor anticipated by the underlying economic theory. Thus, the study only provides limited support for the hypothesis that unmarried women with children will find marriage less desirable when benefits available outside of marriage are higher. However, its findings should be treated with caution as it does not include controls for unobserved heterogeneity.

The study also found that higher expected earnings were associated with a higher probability of marriage, and that never-married mothers who grew up in female headed households were less likely to marry (Blackburn, 2000:141). The analysis found no evidence of a marriage market effect; in the male labour market opportunities were not significantly associated with the likelihood of marriage.

#### Further details - Blackburn (2000) Box 2.3

Blackburn (2000) estimates proportional hazard models for the duration between first birth and marriage. The NLSY covers people aged 14 to 22 in 1979, and the analysis is based on women who had their first non-marital birth between 1978 and 1991.

It excludes observations where the mother is married within six months of the first birth – a procedure adopted by other analysts.

Two welfare benefit variables are used: the sum of AFDC and Food Stamp monthly benefits for unmarried women living in each State with no other income, and whether or not the State had AFDC-UP. Other hazard rate model variables include the time period since the women became nevermarried mothers, the woman's expected earnings (as measured by age, years of schooling and a cognitive and applied skills aptitude test), and 'marriage market' variables (such as, percentage of a county's population in manufacturing and the area's unemployment rate). Three separate analyses are conducted for 'pooled' data, black people and non-black people. The nonblack never married sample includes a small number of Hispanic women.

Non-experimental and experimental studies of welfare reform (that is, AFDC waivers and TANF) and union formation find mixed evidence of a welfare effect. Some programmes appear to increase union formation, but only for sub-groups, and other programmes decrease union formation. In terms of non-experimental studies Bitler et al. (2004) find that AFDC waivers and TANF reduced marriage rates, but Schoeni and Blank (2000) find AFDC waivers increased marriage but TANF had **no** significant effects. For both studies the comparison is made with states not implementing 'major' AFDC waivers. The waivers covered introducing family caps, extending income disregards, amending time limits and expanding eligibility for AFDC-UP. These two non-experimental studies are discussed ahead, and then findings from experimental studies of welfare reform and marriage are outlined.

Bitler et al. (2004) use state level vital statistics on marriage and divorce for 1989 to 2000 to assess the impact of state waivers and TANF on flows into (and out of) marriage. 18

18

## Box 2.4 Further details - Bitler et al. (2004)

Bitler et al. argue that flow, rather than stock, data are required to assess the 'immediate impacts of welfare reform' (2004:219). The authors regress the rate of new marriages (or divorces) with measures of welfare reform, labour market characteristics and demographic variables and other controls (including state and year fixed effects). The labour market variables cover unemployment and earnings and the demographic variables include the proportion of the states' populations that were African American and Hispanic, the proportions of the population living in metropolitan areas, the age distribution of women and female educational attainment. The analysis gives average estimates for the impacts of AFDC waivers and for TANF, and in both cases the comparison is AFDC without waivers.

The disadvantages of using state level vital statistics are: the effects across different age, race and educational backgrounds cannot be explored and the data are aggregated by the state within which the marriage or divorce occurred and not by state of residence. Sensitivity analysis using more detailed data suggests that the latter did affect the impact estimates, but the coefficients were always negative and sometimes significant.

Bitler et al. (2004:222-223) find that, as with other studies, adding state and year fixed effects to the model to control for unobserved variables leads to a large reduction in estimated impacts, see Appendix A. For example, for AFDC waivers the estimated effect on the rate of new marriages is, with labour market and demographic controls, a 15.6 per cent decline relative to the AFDC programme. However, adding state and year fixed effects gives a 4.8 per cent decline in new marriages (p<0.10). Moreover, welfare reform, in the form of AFDC waivers, which might be expected to increase the number of new marriages (given the policy objectives) instead, resulted in a decline of nearly five per cent. Similarly, TANF led to a reduction of 23.3 per cent in new marriages compared to the AFDC programme (p<0.01)<sup>19</sup>. Although the TANF estimate is relatively large there is little variation in the data because it covers a short period of time, and so may not be a robust estimate. The authors also show that the maximum amount of benefit paid to a family of four produced a fall in marriage; however, when State and fixed effects are included in the model this effect became positive, but was statistically insignificant (Bitler et al., 2004:223). The authors conclude that: 'we can say with some confidence that welfare reform is not "pro-marriage" on balance.' (Bitler et al., 2004:232).

Schoeni and Blank (2000) use data on all women aged 16-54 years from the CPS for 1977 to 1999 to investigate the impacts of AFDC waivers 1992 to 1996 and of TANF. The estimates show that the AFDC waivers significantly **increased** rates of marriage by 2.3 percentage points for those with less than 12 years of schooling (Schoeni and Blank, 2000:17 and 31) (p<0.10). As the authors anticipated, the effects for more educated women are lower; there is a statistically significant

Bitler *et al.* (2004) also specify models that include state specific time trends, and these models give similar impact estimates to those reported here.

reduction in the rate of marriage of 1.4 percentage points for women with 12 years of schooling, and the impact for women with more than 12 years of schooling is insignificant (0.7 percentage points). So whilst the impact for the Schoeni and Blank (2000) study is in the opposite direction to the Bitler *et al.* (2004) study, it is for a sub-group – those with less education – admittedly one likely to be affected by welfare reform. However, the CPS is believed to underreport both marriages and divorces and this undermines the robustness of studies using the dataset (Bitler *et al.*, 2004:219).

Their estimates also show that post-1995 policy changes (TANF) had **no** significant impact on marriage (Schoeni and Blank, 2000:21 and 33). That is, the less educated were as likely to get married as the better education following the introduction of TANF.

## Box 2.5 Further details - Schoeni and Blank (2000)

The analysis involves collapsing CPS data by calculating the mean marriage rate (and single headship rate, see Section 4.2.3, for four age groups and three educational attainment groups for each state and year – giving a dataset with 14,076 observations (Schoeni and Blank, 2000:9).<sup>20</sup>

Schoeni and Blank (2000) are particularly interested in the outcomes for women with less than 12 years education (high school dropouts) because they were the group most likely to be affected by AFDC and TANF. They did not expect women with 13 or more years of schooling to be affected by the welfare programmes. (The third educational group was women with 12 years of schooling.)

Different analyses are conducted to obtain the AFDC waiver and TANF estimates. For the waiver analysis, regression is used.<sup>21</sup> The analysis involves comparing outcomes in waiver and non-waiver states, controls for the labour market (as the performance of the economy was lower in waiver states compared to non-waiver states) and demographic characteristics (age, education and race), state fixed effects, year fixed effects and state specific time trends (Schoeni and Blank, 2000:13-14).

To estimate the impact of TANF on marriage rates, Schoeni and Blank (2000:20-23) adopt a difference-in-difference approach. Differences in marriage outcomes for 1995 and 1998 are compared for women with less than 12 years schooling against women with more than 12 years schooling. Controls, year effects and fixed effects similar to those in the regression analysis are included in the analysis (the trend data are excluded).

As well as marriage and headship, Schoeni and Blank (2000) estimate a range of outcomes, such as welfare participation, percentage working earnings, and hours worked.

The regression analysis is also used to estimate the effects of TANF, but they are regarded as being unreliable because there is a lack of observations and state TANF programmes were implemented over a relatively short period of time.

Both the Bitler *et al.* (2004) and Schoeni and Blank (2000) analyses of the impact of TANF on marriage (and divorce) is limited by the relatively short period of time over which TANF was implemented nationally. All states had implemented TANF between September 1996 and January 1998 – a 16 month period (Bitler *et al.*, 2004:218). The speed of implementation means that – unlike the AFDC waiver data – there is relatively little variation between states. Their findings could be biased by random factors that only affected this 16 month period and hence do not provide a good guide to longer term impacts. Both sets of authors are more confident about their AFDC waiver results.

Random assignment studies have also been analysed to investigate the impact of benefits on partnership decisions. In general, social experiments, like non-experimental studies, are inconsistent on whether there is relationship between welfare policies and marriage rates for single parents (Gennetian and Knox, 2003:1). Overall, experimental programmes have tended not to have an effect on marriage, although some have effects for sub-groups. Some of these studies are discussed ahead and in Section 2.2.4.

Gennetian and Knox (2003) conducted a meta-analysis of six welfare programmes at the 18 months to four years follow-up point and find no consistent relationship between these programmes and union formation for either the benefit population overall or sub-groups. The studies cover a ten year period, from 1991 to 2001. All of the programmes were random assignment experiments and incorporated different treatments or were implemented in different areas giving Gennetian and Knox (2003:46) 14 sets of results to review.<sup>22</sup> Whilst the nature of interventions in the programmes varied, across all programmes they included: increased earnings disregards, treating two-parent families the same as single parent families, mandatory employment services, and time-limited eligibility to welfare benefits. A focus of the Gennetian and Knox (2003) review is to explore differences by sub-groups of single parents, as they anticipate that the programmes' effects on marriage and cohabitation will vary by sub-group. The sub-groups examined were defined by: age of youngest child, number of children, age of single parent, priori marital status, race, level of disadvantage (defined in terms of benefit and work history and educational attainment), whether has prior work experience and length of benefit receipt.

Gennetian and Knox (2003:12) report findings for whether the single parents were either married or cohabiting when they were surveyed at the follow-up point of each study. As a consequence their reported marriage and cohabitation rates are lower than if respondents had been asked whether they had **ever** been married

This review includes one of the studies discussed below, namely the MFIP. The other five programmes were Florida's Family Transition Program, Vermont's Welfare Restructuring Program, the national Evaluation of Welfare-to-Work Strategies, Connecticut's Jobs First Program, and Los Angeles Jobs First/Gain. Gennetian and Knox (2003:6) also selectively include the estimates from other programs when appropriate.

or cohabited. The authors estimate the impacts on marriage and cohabitation for sub-groups for each study and then use meta-analysis to estimate average effects across the studies overall and for both sub-groups and types of intervention.<sup>23</sup> Their findings are as follows:

• For single parents across all the studies the welfare programmes had **no** significant effect on marriage or cohabitation (Gennetian and Knox, 2003:14-15). There was an insignificant and small fall in the marriage rate of 0.3 percentage points to 10.6 per cent for those in the treatment group; and an insignificant and small increase of 0.1 percentage points to 10.9 per cent in the cohabitation rate.

The analysis also finds a wide variation in rates of marriage and cohabitation across individual sites. Gennetian and Knox (2003:15) suggest this variation is due to both differences in the length of the follow-up time period of the studies and differences in local cultural, economic or political factors.

- The welfare programmes had **no** significant effect on marriage or cohabitation for specific family sub-groups (Gennetian and Knox, 2003:16).<sup>24</sup>
- By type of policy intervention the only significant effect found is the small effect that improved earnings disregard policies without time limits had on increasing marriage amongst young mothers, those not previously married, and those most disadvantaged (an increase of 3.2 percentage points) (Gennetian and Knox, 2003:21). This finding is consistent with those from wage supplement programmes such as the Canadian Self-Sufficiency Project described ahead.

The implication is that the positive findings of some other studies are '... idiosyncratic to the particular programs, populations or sites ... rather than part of a consistent story of effects on marriage for a broad set of programs.' (Gennetian and Knox, 2003:21).

The Gennetian and Knox (2003) meta-analysis did not include results for the California Work Pays Demonstration Project (CWPDP), which was an AFDC waiver project commencing in 1992. Hu (2003) uses this programme to explore short-and long-term impacts of AFDC-Basic and AFDC-UP on marriage formation and dissolution. It is the explicit modelling of two-parent married families that makes this study of particular interest. The treatment group received enhanced work incentives in the form of lower maximum benefits, a lower benefit withdrawal rate (from 100 per cent to 67 per cent) and an extension of the income disregard (\$30 per month) beyond 12 months (Hu, 2003:947). The dataset comprised

A key assumption underpinning the analysis is that each study's sample is drawn from the same underlying population which has one 'true' effect size.

One statistically significant relationship was found, but within this sub-group there was a ten per cent likelihood of finding two significant relationships by chance alone. Thus Gennetian and Knox (2003:16) conclude that there was no variation in effects by family sub-group.

longitudinal administrative and survey data. The latter was administered in two waves: between October 1993 and September 1994 and between May 1995 and May 1996. The analysis is for female respondents only and includes controls for demographic characterises.

Hu (2003:949-951) finds that, compared to the control group, those women in the AFDC-UP treatment group had significantly **higher** rates of marriage and cohabitation at wave 2.<sup>25</sup> That is, two-parent families receiving the stronger work incentives had marriage/cohabitation rates seven percentage points higher than the control group. Further analyses suggest that this increase is due to a higher proportion being married, rather than due to more cohabitation (Hu, 2003:953-954). The results also imply that there is a time lag before financial incentives affect behaviour – wave 2 was conducted between 29 and 41 months after the start of the experiment.

In addition, for single parents on benefit (AFDC-Basic), the work incentives had **no** statistically significant effect on the proportion married/cohabiting at waves 1 or 2. However, further analysis shows that single parents in receipt of AFDC-Basic and in the treatment group were **more** likely to *cohabit* with the children's father by wave 2 compared with the control group (Hu, 2003:954-955). Cohabiting with the father meant that the family became eligible for AFDC-UP, but the treatment group's lower maximum benefit meant that the woman faced a lower loss of benefit, and hence there was a smaller disincentive to cohabit.

So Hu (2003) provides some evidence that benefits can promote the stability of marriage over approximately three and a half years, albeit for a relatively small proportion of the then total AFDC caseload, that is, two-parent families. In the analysis Hu (2003:946) ignores the effect of other welfare programmes (notably EITC) and as Dickert-Conlin and Houser (1999) and Ellwood (2000b) show, this may be a serious omission that limits the applicability of the findings. The Gennetian and Knox (2003) observations about the importance of local context and the idiosyncratic nature of experimental findings potentially apply to the study.

The MFIP was included in the Gennetian and Knox (2003) meta-analysis discussed previously. However, it is discussed further here because further analysis is available on its longer-term impacts on union formation. Gennetian and Millar (2004) explore whether MFIP had an impact on the marriage of single and two-parent families. MFIP included an enhanced disregard whereby recipients received a higher basic award and were allowed to retain 38 per cent of their earnings (compared to the rapid withdrawal of benefit under AFDC) and changed eligibility rules by removing the 100 hour per month and work history requirements in AFDC-UP, which had served to limit the eligibility of two-parents for AFDC. Thus MFIP should have led to an increase in marriage rates.

There were no statistically significant findings for wave 1 of the survey. This might be because wave 1 was administered only 10 to 21 months after the commencement of the experiment, possibly too soon to measure changes in marriage/cohabitation and divorce/separation (Hu, 2003:951).

The authors make use of survey data from a social experiment used to evaluate the MFIP. The survey is of a representative sample of programme participants who were randomly assigned to treatment and control groups between April and October 1994 and the follow-up survey was administered three years after they joined the programme. The analysis is regression-adjusted to allow for slight differences in the survey data for the treatment and control groups. Separate analyses are conducted for single and two-parent families (defined as both married and cohabitating couples) by benefit status (that is, whether they where new applicants or long-term recipients) in order to reflect differences in MFIP eligibility rules.

Compared to the AFDC control group Gennetian and Millar (2004:285-295) estimate that:

- For single parents who were long-term recipients MFIP **increased** marriage after three years the proportion married had increased by 4.2 percentage points to 12.6 per cent (p<0.05). The authors describe this increase as 'modest'; however, it represents a 50 per cent increase on the control group. (The authors suggest that this increase in marriage might be due to an increase in the MFIP mothers' income, which would increase their 'bargaining power within marriage, thus making them more comfortable and willing to marry ...' (2004:296)).
- For single parents who were recent applicants (that is, families who had received AFDC for less than 24 months at the time of random assignment or who made a new claim on the day of assignment) MFIP had **no** impact on marriage.
- For two-parent families who were already in receipt of AFDC-UP when they started on the programme, MFIP significantly **increased** the proportion in marriage and cohabitation over the three years by 19.1 percentage points to 67.3 per cent. Most of this increase is due to a significant reduction in separations (-9.5 percentage points) the divorce rate for the treatment group was also lower (three percentage points), but this fall was not statistically significant. Thus for two-parent recipients MFIP increased marriage/cohabitation stability by nearly 40 per cent compared to the control group.<sup>26</sup> MFIP might have had this effect because it reduced the stresses on these families by increasing family income which in turn allowed some two-earner families to reduce their working hours.

With respect to two-parent families, Gennetian and Millar (2004:296) conclude:

'... the MFIP results suggest that providing more generous benefits to lowincome working families can help keep families together – at least in the short term – perhaps by reducing the many stresses that these couples face.'

Whether the Minnesota's impacts last beyond three years is considered in Gennetian (2003), who examines the impact of MFIP on the full sample of two-

Sample sizes were too small to estimate the impacts of MFIP on marriage/cohabitation for two-parent who were recent applicants.

parent families over a seven year follow-up period.<sup>27</sup> The outcomes of interest are the divorce rate for those that were married at the start of the study, and the marriage and any subsequent divorce for those that were cohabitating at the beginning of the study. As with the findings for the three year follow-up period, the analysis distinguishes between families who had been long-term and recent recipients at the time of random assignment. She finds that effects were more pronounced for the former group than the latter group.<sup>28</sup>

For existing benefit recipients, MFIP (in comparison to the AFDC control group) increased overall marital stability by significantly decreasing divorce by 3.5 percentage points to 10.4 per cent (a 25 per cent reduction) over the seven years of the follow-up period (Gennetian, 2003:6-8). Moreover, this effect was more pronounced for: black married couples (minus 9.9 percentage points), families with children whose youngest child was aged under six years at study entry (minus five percentage points), and larger sized families (three or more children) (minus 5.2 percentage points). In addition, MFIP reduced the divorce rate of those that were married when they entered the study (by minus 2.9 percentage points), but this fall is not statistically significant. However, for black, non-Hispanic married couples at baseline, MFIP did reduce the divorce rate by 20.5 percentage points over the seven year follow-up period (a 73 per cent reduction to 7.8 per cent) (Gennetian, 2003:14-16).

For existing recipients who were cohabitating at study entry, MFIP did **not** encourage marriage (Gennetian, 2003:17). However, of those that did subsequently marry the couples in MFIP were less likely to get divorced by the end of the follow-up period compared to the AFDC control group (minus 5.3 percentage points in the divorce rate, or a 66 per cent reduction). Thus MFIP also promoted marital stability amongst this group of former cohabitating couples.

For recent two-parent applicants, MFIP had **no** significant effect on the divorce rates over the seven year follow-up period (Gennetian, 2003:21).

Both the three- and seven-year follow-up studies suggest that newer and existing benefit applicants may respond differently to marriage incentives. Implying that other studies should explore the distinction between longer term and more recent claimants. However, a limitation of the Gennetian (2003) analysis is that it only focuses on formal/legal divorces and the data do not capture separations or living apart. In addition, whilst Gennetian (2003) shows that benefits might help to sustain marriages by decreasing the divorce rate, the quality of these sustained marriages is unknown.

This finding confirms that of Hu (2003), namely, that benefits can promote the

The analysis required the matching of administrative marriage and divorce records with MFIP data.

Note, effects are for martial status at the time the respondent entered the study (whether treatment or control groups).

stability of two-parent families (see previously). However, the MFIP was included in the Gennetian and Knox (2003) meta-analysis and, as Gennetian and Millar (2004:277) acknowledge, the finding that earnings disregards could increase marriage and marital stability, might not be generalisable from Minnesota to other localities.

Both the Hu (2003) and Gennetian and Millar (2004) studies also only show that their treatment groups had less adverse effects on union formation than the old AFDC policy (as represented by the control group). They are of limited use in increasing our understanding of the overall effect of TANF on current rates of union formation and dissolution (Burstein, 2007:413). By definition they also provide no evidence of any impacts on non-participants in the experiments.

In practice, welfare reform may have had a relatively small impact on marriage because welfare reform gave women greater financial independence through the emphasis on obtaining employment. This argument is supported by Nandi (2008), who, using data from the National Longitudinal Survey of Youth for the period 1979 to 2002, finds that single, non-employed women who at some point in time were welfare recipients gain more, in terms of household income, if they obtain full-time employment (whilst remaining single) compared to if they enter marriage or cohabitation. The estimated income gains from full-time employment were 220 per cent if they worked full-time compared to 119 per cent from cohabitation, 182 per cent from marriage and 113 per cent from working part-time (p<0.01) (Nandi, 2008:20 and 23). As a route out of poverty, therefore, full-time employment is a better option than marriage or cohabitation. This arises because female benefit recipients' potential partners tend to have low paid jobs; even though spouses tend to have higher earnings than cohabitors.

#### EITC and union formation

The US literature on the impact of EITC on partnering is less extensive than that for benefits, but still broader than the UK literature. There are a small number of studies that have examined actual behavioural responses, and, in general, the effect of EITC on actual marriage decisions is small and statistically insignificant (Hotz and Scholz, 2003:184-5; Hoffman and Seidman, 2003:98; see also Baughman and Dickert-Conlin, 2003).<sup>29</sup>

For example, Dickert-Conlin and Houser (2002:35) conclude that, on average, EITC did **not** influence marriage decisions. Their analysis is based on women aged between 18 and 50 with children, using panel data from the Survey of Income and Program Participants for the period 1989 to 1995. (The EITC received is imputed as the maximum Federal and State EITC for a woman holding constant her demographic and family characteristics. Their model includes controls for individual and State characteristics, individual fixed effects and time varying covariates. It

This contrasts with the US *general* tax system, where the probability of marriage does fall as the marriage penalty increases (see Alm *et al.*, 1999; Eissa and Hoynes, 2000). The EITC effects are small possibly because marriage related tax liabilities are small relative to total income for most households (Dickert-Conlin, 1999:222; Dickert-Conlin and Houser, 2002:30).

also treats cohabitating couples as unmarried.) However, there were differences by whether the women were married or unmarried the first time they entered the panel (Dickert-Conlin and Houser, 2002:35-36). For the initially married women, EITC encouraged them to remain married (or discouraged divorce) – a \$100 increase in EITC increased the probability of marriage by 0.8 percentage points. For unmarried women there was no statistically significant correlation.

Ellwood (2000b) comes to a similar conclusion, namely, that EITC - and other welfare reforms - estimated impacts on marriage are small and ambiguous. Ellwood (2000b:1100) concludes that there is no 'real evidence that EITC marriage penalties were reducing marriage.'

# **Box 2.6** Further details - Ellwood (2000b)

Ellwood (2000b:1073) uses a difference-in-difference approach to examine changes in outcomes for sub-groups both geographically and over time. This is done by treating the rapid expansion in EITC, which did not affect people equally, as a natural experiment. Using data from the Current Population Survey 1975 to 1999 he takes women aged between 18 and 44 and using their personal characteristics (such as age, education and race) predicts a 1998 wage for them for each year. The women are then allocated to wage/ skill quartiles for each survey year. Ellwood (2000b:1075) then tracks what happens to similar quartiles of women over time. If financial incentives vary for women in the bottom quartile compared to the top quartile, and he argues that they do, then there is a natural experiment allowing within and between quartile comparisons for groups of women. The expectation is that over time marriage should be more common for women with children in the bottom quartile (Ellwood, 2000b:1093-1097). However, the data show that the marriage rate for those in the bottom quartile continued to fall over the observation period – they had not responded to the incentives created by EITC and other welfare reforms. A finding confirmed by a small impact estimate on marriage of minus four percentage points when using differencein-difference to compare the bottom quartile with the third quartile for 1986 and 1999. This finding is inconsistent with economic theory, which would predict a marriage effect. A similar impact estimate for marriage (minus three percentage points) was found for a comparison of women with children and women without children in the bottom quartile. However, as Ellwood (2000b:1096) points out, his analysis uses women's ages to predict wages, and women in the bottom quartile tend to be younger than those in the other quartiles, and if younger women postponed marriage decisions for reasons unrelated to EITC this could bias the findings. So Ellwood limits his sample to women aged 24 to 44 years. This shows that the proportion of married women in the lowest quartile flattens out after the early 1990s, implying that the decline in marriage amongst the lowest skill women aged over 24 had been slowed. Unfortunately, the differences in the trends with other quartiles are 'too tenuous to draw strong conclusions.' (Ellwood, 2000b:1096).

Eissa and Hoynes (1999 cited in Dickert-Conlin and Houser, 2002:28 and Hoffman and Seidman, 2003:97-99) also find that EITC had modest impacts on marriage. They calculate the income tax cost of marriage and *simulate* how EITC affects the probability of marriage.<sup>30</sup> They find that increasing the tax cost of marriage by \$1,000 reduces the probability of marriage by 1.3 percentage points. Further simulations show that EITC (as operating in 1997) increased the marriage rate by one percentage point for families with low-incomes (less than \$15,000) and reduced the marriage rate for those with incomes over \$25,000. Moreover, changes in EITC between 1984 and 1997 increased marriage rates by one to five per cent for low-income families (that is, with incomes below \$25,000), and reduced marriage rates by one per cent for families with middle incomes (of between £25,000 and \$75,000). These are weak incentive and disincentive effects.

Eissa and Hoynes, (2000b) extend this analysis to look at the trade-off between marriage and cohabitation. They impute the presence of cohabitating households using data on the relationship to the head of household for other household members. Their analysis shows that increasing the tax cost of marriage by \$1,000 significantly **reduces** the probability of marriage by 0.4 percentage points compared to cohabitation. This analysis incorporates Federal taxes and EITC. The analysis is used to simulate the effect of EITC (and other tax components) on marriage rates. Simulations for the same period (1984 to 1997) show that EITC for low-income families (\$10,000 to \$15,000) increased the likelihood of marriage by 4.3 percentage points, but decreased it for higher income groups – two to three percentage points for middle income groups and less than one percentage point for the higher income group (\$75,000 and over) (Eissa and Hoynes, 2000b:19 and 37). These simulations for EITC are described as 'modest'.

However, other factors, even for a married couple in the phase-out stage, may help maintain marriage (Ellwood, 2000b:1070-1072). Whilst a woman in a low-income married couple where both partners are working might gain a larger EITC by divorcing, the family does enjoy the extra income from EITC and this might reduce the level of family stress and increase the probability of the marriage surviving. Moreover, EITC also provides a lower work incentive for the second earner to work, and they may decide to spend more time with the family because in Beckerian terms, see Section 1.5.2, this extra time gives marriage a comparative advantage over divorce. Moreover, people may consider longer-term benefits and costs of marriage and so not respond to shorter-term gains or losses arising from FITC.

#### New Hope programme

One other US study, the New Hope programme, deserves mentioning because it is in effect an outlier – finding a relatively large positive effect on marriage.

In their analysis they use repeated cross-sectional data from the Current Population Survey for 1985 to 1998 for women aged 18 to 47 years, and use childless women as controls. The models include controls for individual characteristics and state fixed effects.

New Hope was a US voluntary programme operated in two low-income neighbourhoods in Milwaukee, Wisconsin during the mid and late-1990s that sought to increase participants' levels of employment and income. It offered participants in full-time employment an anti-poverty package comprising earnings supplements, low-cost health insurance, child-care subsidies and community service jobs for those unable to find paid work in the private sector (Gassman-Pines and Yoshikawa, 2006:15). Programme benefits were provided for three years. The programme was provided outside of the welfare system and was open to both welfare recipients and non-recipients.

New Hope was evaluated using random assignment. Gassman-Pines and Yoshikawa (2006:13) assess whether New Hope increased marriage rates five years after random assignment for never married mothers. (Participants were randomly assigned in the New Hope programme between August 1994 and December 1995.) The programme has been shown to increase participants' employment rates and income. Gassman-Pines and Yoshikawa (2006) consider whether the increased financial stability and feelings of well-being it provided also led to an increase in marriage rates.

# Box 2.7 Further details - Gassman-Pines and Yoshikawa (2006)

Gassman-Pines and Yoshikawa (2006:16) count as being married at year five those women whose spouse lived with them and those whose spouses lived apart. Their measures of income and employment relate to years one and two and the indicators of the woman's well-being, such as depression, parenting stress and material hardship, relate to year two only. There is, therefore, a three year gap between the main outcome variable of interest and some of the factors Gassman-Pines and Yoshikawa (2006) believe may have affected marriage rates of never-married women. The logit analysis includes some demographic and socio-economic baseline variables to help control for confounding variables. These covariates include race, presence of two or more children, earnings, whether they had a car, and respondent's age (Gassman-Pines and Yoshikawa (2006:18).

They find that New Hope significantly **increased** the rates of marriage of never-married women (Gassman-Pines and Yoshikawa, 2006:18). Five years after random assignment 20.7 per cent of those in the New Hope action group were married compared to 11.8 per cent of those in the control group – an 8.9 percentage point increase. This is one of the largest reported increases. Gassman-Pines and Yoshikawa (2006:25) suggest that this might be because the follow-up period (five years) is longer than for most other studies (see, for instance the Gennetian and Knox (2003) meta-analysis discussed previously). They argue that to detect changes in demographic behaviour longer follow-up periods are required (Gassman-Pines and Yoshikawa, 2006:25).

Additional (non-experimental) modelling shows that (Gassman-Pines and Yoshikawa (2006:20):

- There was an association between annual income and marriage (p<0.10). Each \$1,000 increase in annual income raised the likelihood of a woman who was never-married at baseline being married by year five by 1.08 times.
- Women who endured higher levels of material hardship at year two were less likely to be married by year five (p<0.05).

However, Gassman-Pines and Yoshikawa (2006:26) acknowledge that how increases in women's income might promote marriage is unknown. In addition, there was no association between marriage and other measures of financial stability and women's wellbeing.

Gassman-Pines and Yoshikawa (2006:24) seek to establish that the economic effects of New Hope (increases in employment and earnings) preceded marriage. Analysis shows that New Hope had no significant effect on marriage by year two (eight per cent of the action group were married compared to six per cent of the controls). Yet New Hope had significantly increased employment amongst never-married women by year two. Moreover, over years three to five, when most of the marriages occurred, New Hope did not significantly increase employment or earnings (albeit income did significantly increase (p<0.10)). Thus according to Gassman-Pines and Yoshikawa (2006:24) New Hope:

'... provides evidence that an anti-poverty program that increases employment and income among never-married mothers can affect their entry into marriage.'

Nonetheless, the majority of women who were never-married at baseline were five years later neither married nor cohabitating. Moreover, New Hope had no significant impact on unmarried cohabitation (Gassman-Pines and Yoshikawa (2006:18).

The New Hope study finds a relatively large effect. However, the sample used in the Gassman-Pines and Yoshikawa (2006) study is relatively small (337). It also comprised a group of participants who were likely to be less disadvantaged than the population in receipt of welfare benefits in general. The study is also of one location, and as outlined in Section 1.6.2 and ahead social experiments may not adequately take into account local context, which may affect the results obtained.

# 2.2.4 Canadian evidence on union formation and the welfare system

This sub-section considers two Canadian studies. The first is of the Canadian SSP, which is an important study because (like MFIP) it highlights the importance of local context. The second study is unusual because it is focused on pension reforms.

#### SSP

Two social experiments in Canada provide differing findings on whether an earnings supplement paid to single mothers altered rates of marriage and cohabitation. Harknett and Gennetian (2003) explore the effect of SSP on union formation (marriage and cohabitation) using experiments in two culturally and geographically distinct provinces: British Columbia and New Brunswick. SSP was an earnings supplement paid to single parents in receipt of benefits (Income Assistance) when they entered full-time employment and was payable for up to three years (see Harknett and Gennetian (2003:452-453) for further details). SSP also disregarded any income or assets from a spouse or partner, and so eliminated any union/marriage penalties. The amount paid was also unaffected by whether they married or cohabitated with a partner. The experiment was conducted between 1992 and 1998 and single parents were randomly allocated to a 'supplement offer' group and a control group. Members of the treatment group only received SSP if they obtained full-time work within one year of the experiment commencing.

Qualitative research reveals that some former single parents attributed their subsequent marriage to their move into employment as it allowed them to meet their husbands, although, for some, paid work did not lead to an enhanced social life due to increased exhaustion and lack of time (Harknett and Gennetian, 2003:458-459). The financial independence provided by employment also allowed a small number of recipients to leave physically and emotionally abusive relationships. The analysis of welfare effects is based on surveys of the 'supplement offer' group and a control group and it excludes male single parents and mothers who were married or cohabitating at baseline. Control variables (such as age, educational attainment, martial status, age of youngest child and prior earnings) were used in the regression analysis to adjust for some slight differences in the composition of the sample groups.

Harknett and Gennetian (2003:463-467) find that over the 36 month follow-up period:

- SSP significantly **reduced** single mothers' rate of marriage in British Columbia from month nine onwards;
- SSP had **no** effect on rates of cohabitation in British Columbia;
- SPP significantly **increased** the rate of marriage amongst single mothers between months six and 34 in New Brunswick; and
- SSP significantly **increased** cohabitation rates in New Brunswick for a few months in the second and third years of the follow-up period.

Overall, compared to the control group, SSP decreased the probability of being ever married or cohabitating in British Columbia by 2.5 percentage points (a 16 per cent fall) but increased the likelihood of being ever married or cohabitating in New Brunswick by 4.3 percentage points (a 22 per cent rise). The difference in the two impact estimates is statistically significant. SSP had a larger impact on marriage compared to cohabitation in both provinces: -2 percentage points

compared to -0.1 in British Columbia; and 2.4 percentage points compared to 1.6 in New Brunswick. It follows that the effects of SSP were larger in New Brunswick than in British Columbia.

Harknett and Gennetian (2003:467-474) suggest that the differences by province arise from different labour market and local policy contexts. Benefits were less generous in New Brunswick and the economy was weaker compared to British Columbia and this could have encouraged people to pool their incomes through forming unions. They also reason that as the impact of SSP on income and employment were similar in both provinces, then these economic mechanisms cannot account for the observed difference between provinces.

Harknett (2006) extends the SSP analysis by using non-experimental methods to assess whether the effects on those that took up the offer of SSP differ from those that did not. Using statistical matching techniques (propensity score matching) she compares the 36 per cent in the action group that actually received SSP with a matched sample from the control group. A similar comparison is made between the 64 per cent in the 'supplement offer' group who did not take up SSP with a matched 'non-takers' group from the control group. Matching is necessary because the 'takers' or SSP recipients in the action group are self-selecting and systematically differ from 'non-takers'.<sup>31</sup> In general, 'takers' are less disadvantaged – they have more education, more work experience, face fewer barriers to work, and have greater work aspirations – and they are younger (Harknett, 2006:745-746). These characteristics are also associated with an increased likelihood of marriage or cohabitation.

This further analysis shows that (Harknett, 2006:755-764) in New Brunswick SSP recipients, in comparison to 'non-takers', were **more** likely to marry or cohabit over the 36 months of the study. Furthermore, those 'takers' who were less advantaged in the labour market – with more obstacles to work, lower educational attainment and limited work experience – were more likely to form unions.

For British Columbia the results of the analysis are unusual. The expectation is that the effects of SSP should be concentrated amongst actual SSP recipients rather than non-recipients. Yet the fall in union formation is slightly higher for 'non-takers' (-3.1 per cent) than for 'takers' (-2.9 per cent). Harknett (2006:758) claims that this means that the British Columbia results 'lack robustness'. However, the difference between the two comparison groups is small, and whether the difference is statistically significant is unknown. It is also possible that the British Columbia results reflect the effect of unobserved factors and in particular 'local' contextual variables.

Harknett (2006:763) does show that the fall in union formation in British Columbia is focused amongst the more advantaged in the labour market. Suggesting an 'independence' effect; this sub-group were able to forgo the income of a potential partner.

However, the matched comparison groups may also differ on unobserved characteristics.

#### Pension reform

A second Canadian study by Baker et al. (2003) looks at the impact of the removal of marriage penalties for surviving spouses in the pension system during the 1980s. This is a time-series cross-sectional study. Simplification of the Canadian Income Security system led to the removal of 'substantial' marriage penalties, whereby surviving spouses lost their pensions on remarrying (Baker et al., 2003:13-14). The analysis uses administrative data for 1975 to 1995 and takes advantage of Canada having two separate but similar pension schemes: the Quebec Pension Plan and the Canada Pension Plan for people not working in Quebec. Marriage penalties were removed from the former in 1984 and from the latter in 1987. The analysis estimates remarriage rates for males and females in Quebec (the rest of Canada) using the rest of Canada (Quebec) as controls. Other control variables cover province and year effects and the ratio of unmarried females to unmarried males as a measure of the 'marriage market'. The analysis examines flows into marriage by previous marital status for males and females separately for set age groups. The interpretation of the analysis is complicated by rapid increases and decreases in the remarriage rates of younger males (aged under 35) in Quebec between 1986 and 1989 (Baker et al., 2003:21-22). The authors cannot explain these 'spikes' in the data and so focus on the results for older males.

Baker et al. (2003) find that the marriage penalties had a 'large and persistent' effect on the remarriage rates of widows aged 15-59 years and on prime age widowers. In general, the removal of the marriage penalty increased remarriage rates of females aged under 65 by between 21 and 108 per cent (Baker et al., 2003:22-23 and 44). Results for females aged over 65 are small and statistically insignificant. They estimate that the remarriage rates of widowers aged 45 to 59 increased by 20 to 52 per cent following the removal of the marriage penalties. The results for older males are ambiguous and can be statistically insignificant. Widowers also had higher remarriage rates than widows. The impacts of the removal of the marriage penalty in the pension scheme was larger in Quebec, which is possibly because the size of the penalty was larger in the province than elsewhere. Baker et al. (2003:29) suggest that the rise in remarriages after the removal of the penalties was mainly by individuals who had already identified a partner and possibly had been previously cohabitating.<sup>32</sup> It was thought likely that it would take a longer period of time for any widows or widowers who decided post-reform to re-enter the marriage market to find a suitable partner and remarry.

A separate analysis using survey data suggests that individuals most affected by the removal of the penalty were both more educated and wealthier (Baker *et al.*, 2003:28-31). The latter is possibly because the laws relating to the treatment of assets and income in Canada favour marriage over cohabitation. However, the sample sizes in this analysis are small and many of the coefficients are not statistically significant.

The marriage penalties in the survivor pensions only applied to people who remarried and not to those who cohabited.

Baker *et al.* (2003) do find a positive 'pension' effect for women aged under 65. However, the findings should be treated with some caution as there are outstanding queries about the quality of the data.

#### 2.3 Overview

Underpinning the majority of the studies reviewed in this chapter is Becker's economic theory of marriage – that women will decide to marry if the benefits of doing so exceed the costs, see Section 1.5.2. Associated debates about whether there are local 'marriage markets' (in particular a lack of marriageable men) and an 'independence effect' (whether higher female earnings lead to more or less marriage or cohabitation) inform the analysts' selection of control variables.

The Beckerian utility maximisation model predicts that marriage penalties in the welfare system would lead to decreases in marriage (and increases in divorce). It is possible to find selected studies that provide some evidence for this outcome. There is US evidence that welfare reform (in the form of AFDC waivers and TANF) has been less 'anti-marriage' than AFDC (Burstein, 2007:421). However, in the non-experimental literature, observed welfare effects are less strong once controls for unobserved variables are included in models. Moreover, a major meta-analysis of US studies finds no significant overall effect on marriage/cohabitation.

Evidence for the impact of tax credits on union formation in the UK and US is fairly consistent – there is, at most, a modest negative impact on marriage. In the UK the introduction of WFTC meant single parents were less likely to marry than childless single women.

Several of the studies demonstrate that any impacts vary by sub-group. Differential impacts have been found by age of child, benefit duration, marital status (marriage vs. cohabitation), race and gender, type of policy intervention, and whether married or cohabitating. Welfare effects on union formation can also be lagged, and US and Canadian studies highlight that findings vary by local context. Data problems also undermine the robustness of some studies.

Overall, there is no set of studies that encompass analyses of an extensive range of sub-groups, explores long term effects, has high external validity, and uses robust data. Tentatively, this review finds that whilst there are some studies showing a negative effect on marriage/cohabitation, there is, in general, no consistent evidence for a large and significant welfare effect on union formation.

# 3 Union dissolution

## Summary

A relatively small number of studies on welfare systems and union dissolution are reviewed. These studies do not always distinguish between marital breakdown and cohabitation breakdown.

Only one United Kingdom (UK) study is reviewed, which reveals that the introduction of Working Families Tax Credit (WFTC) had no overall impact on mothers' decisions to dissolve unions. However, if the male partner had low earnings (that is, working less than 16 hours per week) the WFTC increased the divorce/separation rate (by 2.4 percentage points; an 80 per cent increase). In addition, for this sub-group, the divorce/separation rate was slightly higher where:

- the woman had a 'high' level of education ('A' Level or above) (2.6 percentage points); or
- there were young children (aged under five years) in the family (three to four percentage points).

For the US there is weak evidence that those who could gain under the tax system could separate, but mixed evidence as to whether benefits had a significant impact on divorce/separation rates.

An Australian study shows that over a 12 month period income support recipients were more likely to separate than non-recipients (by 1.8 percentage points). However, this study also illustrates the importance of other factors that have an affect on union dissolutions (such as, levels of satisfaction and whether the couple were cohabitating rather than married). In some cases these other factors are more important in terms of the size of the impact they have on union dissolution.

Overall, the small number of studies, the mixed findings, queries about the robustness of some of the studies and that they do not distinguish between dissolving marriages and cohabitations mean that there is no overwhelming evidence that welfare systems have had a major impact on union dissolution.

## 3.1 Introduction

This chapter takes forward the discussion in the previous chapter on partnering by considering union dissolution. There are fewer studies on union dissolution than on partnership formation and childbearing. In general, the studies consider the dissolution of unions, and do not always distinguish between marital breakdown and cohabitation breakdown. This is a potential shortcoming as conceivably different factors leading to dissolution may be operating in marriages and cohabitations.

The evidence of the impact of welfare systems on union dissolution is considered by country:

- the UK, see Section 3.2.1;
- the US, see Section 3.2.2), and
- Australia, see Section 3.2.3).

#### 3.2 Research evidence on union dissolution

# 3.2.1 UK evidence on union dissolution and the welfare system

Only one UK study, Francesconi *et al.* (2007), has been found on welfare systems and union dissolution. They explore whether WFTC had any impact on mothers' decisions to divorce/separate.<sup>33</sup>

# Box 3.1 Further details - Francesconi et al. (2007)

Francesconi *et al.* (2007) develop a household bargaining model and test it using longitudinal data on 3,235 married and cohabitating couples from the British Household Panel Survey (BHPS) for the period 1991 to 2002 (Francesconi *et al.*, 2007:27). The women in the couples were aged at least 16 years and born after 1946. Of the families, 1,430 had dependent children. The BHPS data for married couples show that divorce rates were slightly higher for those couples with children (three per cent) than those without children (two per cent) (Francesconi *et al.*, 2007:28 and 48). To focus on the population of interest, the data excluded couples where the man worked more than 16 hours per week and had high earnings (that is, was in the top earnings quartile). Also excluded from the analysis were: couples where one partner was long-term sick or disabled, or in school full-time; and couples where the man worked more than 16 hours and had a high income (earnings in the top quartile of the income distribution).

Their paper refers to 'married mothers', but the analysis does not distinguish between married women, cohabitating couples or other types of 'living-in' partnerships in the sample. Thus the results more properly cover partnership dissolution. Their study also considers the impact of WFTC on work incentives and use of paid child care.

#### Box 3.1 continued

They adopt a difference-in-difference approach. The statistical analysis is based on regression models that take married mothers with children as the treatment group and married women without children as the control group. The models include controls for individual characteristics (such as age, education, region of residence, and number and ages of children) and individual fixed effects (Francesconi *et al.*, 2007:29-30). The models also capture non-WFTC policy changes, such as the introduction of the minimum wage.

The analysis shows that for the sample as a whole there is **no evidence of an impact on union dissolution** (see Section 2.1.2) for married/cohabitating mothers following the introduction of WFTC (Francesconi *et al.*, 2007:36). Similarly, the reform did not affect the divorce/separation rate of married/cohabiting women with partners who were in WFTC eligible employment.

However, there were significant differences for certain sub-groups. For mothers in partnerships with low earning men (that is, where he did not work or worked for less than 16 hours per week) the introduction of WFTC led to a 2.4 percentage point **increase** in the divorce/separation rate compared to their childless counterparts (p<0.05) – equivalent to nearly an 80 per cent increase in their divorce/separation rate:

'The result implies that for such women, the introduction of WFTC reduced the gains from marriage. This is consistent with substantial improvements in the employment and financial positions of single mothers as a result of the reform ...'

Francesconi et al. (2007:36)

Further analyses of this sub-group (married/cohabitating mothers in poor households) show that the divorce/separation rate was significantly higher (2.6 percentage points) following the reform for women with a 'high' level of education (that is, with an 'A' level or higher qualification) (Francesconi et al., 2007:38 and 56). Across the sub-group there was no statistically significant effect for those with lower educational qualifications. However, controlling for number of children shows that for married/cohabitating women with husbands with low earnings and young children (that is, children aged four years or under) the introduction of WFTC significantly increased the divorce/separation rate (by 3 to 4 percentage points) irrespective of whether the mothers had low or high educational qualifications. Thus the impact of WFTC on union dissolution was higher in families with young children (see Section 1.4.2). Francesconi et al. (2007:38) point out that the child credit element of WFTC was generous for families with young children. A generous childcare credit implied that there were potentially large financial gains for single parents in eligible employment, but no gains if they remaining married to a husband working fewer than 16 hours per week.

#### Hence:

'The requirement for both parents to work at least 16 hours per week may therefore inadvertently have contributed to a considerable increase in the divorce rate for a subset of married women.'

Francesconi et al. (2007:28)

Indeed, there were no statistically significant effects on the divorce/separation rate for women married to men working 16 or more hours per week with either low or high earnings. Suggesting that the more generous tax credit that these sub-groups received left couples unaffected by the reform.

There are some limitations to this analysis, first, (and by definition) the effects are only for women with children, secondly, the analysis does not distinguish married from cohabiting mothers, yet United States (US) evidence suggests these groups may respond differently to welfare incentives, and thirdly, it is not clear that childless married women are the best comparator for married parents.

More UK studies are required in order to determine whether Francesconi *et al's* (2007) findings are replicated and if they also apply to the current welfare system.

## 3.2.2 US evidence on union dissolution and the welfare system

Some commentators claimed that in the US the welfare system can act as a disincentive for marriage. A woman with little or no earnings might have been entitled to Aid to Families with Dependent Children (AFDC) and later Temporary Assistance for Needy Families (TANF) if she separated from the father of her children. Moreover, as AFDC-Basic and the associated Food Stamps and Medicaid were not available to couples with children, it has been suggested that they encouraged divorce and separation and so the formation of single parent families.

Dickert-Conlin (1999:218) claims to be the first to consider the impact of the combined tax and transfer systems on decisions to separate. She argues that the effects of the transfer system ((AFDC), Food Stamps and Social Security Insurance) and the tax system (including Earned Income Tax Credit (EITC)) should be considered together because of the way in which they interact for low-income households. In principle the benefit system incorporates marriage penalties – most married low-income couples would receive higher benefits if they separated – whilst the tax system offers marriage penalties and bonuses – in general, it rewards marriage for low-income single earner couple, but penalises marriage for two-earner couples. Using panel data from the 1990 Survey of Income and Program Participants, Dickert-Conlin (1999) examines whether married women aged 18 to 44, who were calculated to have a marriage penalty, subsequently separated over a 17 month follow-up period. The estimates for marriage penalties are derived using a simulation model (as used in Dickert-Conlin and Houser (1998)) that 'divorces'

married couples.<sup>34</sup> Over the subsequent 17 months 131 couples separated – a divorce rate of four per cent. Her analysis shows that those couples more likely to gain through separating by receiving a lower (1990) tax liability were more likely to separate (Dickert-Conlin, 1999:228-229).<sup>35</sup> The estimate for the penalty arising from the benefit system, although suggesting that the more a couple would gain in benefits if they separated the more likely they were to separate, was statistically insignificant. The model predicts that making the tax system marriage neutral would increase the chances of couples separating by 3.3 per cent (Dickert-Conlin, 1999:230). Making the tax system marriage neutral, somewhat surprisingly, increases the rate of separation, because the so-called 'marriage tax penalty' was, in fact, an average subsidy of \$498 per annum to married couples. Dickert-Conlin (1999:234) concludes that the analysis provides '... weak support that taxes [which include EITC] affect the decision to separate.' The model, however, only considers the survival of marriages over a short period of time.

Ono (1998:685) using data from the Panel Study of Income Dynamics for 1968 to 1985 finds that receipt of AFDC in the previous year had **no** significant effect on couples' decisions to separate. Where couples in receipt of AFDC did separate this was because of the husband's poor economic position rather than the possibility of the wife becoming economically independent via AFDC after marriage. Ono (1998:688) concludes:

'... the results in this study do not support the argument that a wife's high potential for receiving AFDC after marriage encourages marital dissolution.'

These findings resonate with Zimmerman's earlier study on states' expenditure on public welfare and divorce (Zimmerman, 1991). She found that, controlling for other variables, the best predictor of (then) current state divorce rates was the states' prior divorce rates (Zimmerman, 1991:144). Zimmerman (1991:141) interprets states with higher per capital expenditure on public welfare programmes as being more socially integrated. Such societies are seen as sharing a general consensus on rules and norms and supportive of family stability and cohesion. Accordingly, Zimmerman (1991:146) concludes that 'welfare state activity' does not have a destabilising effect on family life.

To assess the effect of the tax and transfer systems on separation decisions she uses probit regression and controls for individual characteristics (for example, age, education and race) and state characteristics (for instance, state unemployment rate). Variables are also included to control for the costs of separation (for example, number of children in the family and marriage duration). The non tax and transfer explanatory variables are lagged to 1990 as they presumably affected the decision to separate, prior to the actual separation. Also, the women in the analysis were in their first marriages and had remained married throughout the 1990 calendar year.

The coefficient is 0.0041 and is significant at the six per cent level. A \$100 increase in tax liability when married would lead to a 0.04 percentage point increase in separations.

However, these studies do not capture later welfare reforms in the US. The Bitler *et al.* (2004) study already mentioned, in Section 2.3, also considered flows out of marriage due to state waivers and TANF. By considering flows the authors exclude the stock of long-term marriages that were at a low risk of dissolution. Bitler *et al.* (2004:224-225) find that AFDC waivers reduced the rate of new divorces by 5.5 per cent (p<0.01) compared to the AFDC programme, whilst TANF reduced new divorces by 12.6 per cent (p<0.10).<sup>36</sup> In other words, welfare reform, as intended by policy makers, did lead to a fall in the number of new divorces. As a consequence there may also be fewer remarriages, as most divorcees remarry. However, the doubts expressed, in Section 2.2.3, about the robustness of the TANF results apply here as well. Moreover, the fall in new divorces in AFDC waiver states is what would be expected given the policy intent. Yet the comparison of wavier states with non-waiver states gives limited guidance on the welfare effect of current programmes.

# 3.2.3 Australia evidence on union dissolution and the welfare system

There is some evidence from Australia that receipt of welfare benefits (Income Support (IS)) imposes a marriage penalty. Bradbury and Norris (2005), using administrative and survey data, show that married benefit recipients were more likely to separate than non-recipients.<sup>37</sup> Using a sample of the administrative data for Family Tax Benefit Part A (FTBA), which is a means-tested benefit paid to families with dependent children, they find that over a one year period (June 2002 compared with June 2001) recipients of the higher rate of FTBA were, on average, three to four percentage points more likely to separate than those receiving the low rate of benefit (p<0.05) (Bradbury and Norris, 2005:428-431). The higher rate of FTBA was paid to couples in receipt of IS or who had low to moderate family incomes, whilst the lower rate was paid to those with higher incomes. The analysis also shows that women in common law marriages were slightly more likely to separate (by 5.29 percentage points (p<0.05)) than those with low-incomes. Moreover, having a younger child decreased the likelihood of separation, but having an older child slightly **increased** the chances of separation. This analysis, which was restricted to FTBA recipient families with at least one child aged under 16, included a limited number of controls for age of recipient, relationship type (whether legal or common law marriage) and age and number of children. Given findings reported elsewhere in this review, this limited degree of control raises doubts about the robustness of these findings.

Bradbury and Norris (2005:431-442) also report more elaborate analyses using longitudinal survey data with controls for income, demographic and human capital

These estimates are based on models that include labour market and demographic controls as well as state and year fixed effects.

The analyses reported here cover both legal marriages and de facto or common law marriages.

variables (for example, relationship type, age at marriage, marriage duration, number and age of children and educational attainment) and, what they term, potential intervening variables (such as, level of satisfaction with life and the relationship, mental health, emotional well-being, saving habits, social functioning and 'vitality'). The data were taken from the first two waves of the Household, Income and Labour Dynamics in Australia survey. Using logistic regression the authors examine for mixed-sex legally married and de facto couples of working age the variables associated with separation. In particular whether, over a 12 month period couples who were income support recipients, were **more** likely to have separated than non-IS recipient couples. They find that couples receiving IS were, on average, more likely to separate (by 1.8 percentage points (p<0.10)) than couples not in receipt of benefit.

However, the analysis reveals that other factors were significant; indeed, some had a larger impact upon separation. In particular, those currently in a *de facto* relationship (that is, cohabitating) were 3.4 percentage points more likely to separate than those who had never been in such a relationship. Confirming other research that cohabiting couples were more likely to end their relationship than legally married couples. Couples were more likely to separate where:

- they married at a young age (each additional year of average age at marriage reduced the likelihood of separation by 0.46 percentage points);
- they were dissatisfied with their relationship (2.4 percentage points);
- the male was anxious or depressed (0.9 percentage points);
- they had a low score for social functioning (0.84 percentage points); and
- the couples had high levels of vitality (1.3 percentage points) that is, they did not say they lacked energy or felt tired a result that is 'puzzling'.

This analysis is limited by the short period (one year) over which changes in relationships are observed and this complicates the interpretation of the findings. For instance, male depression may not be a cause of separation as instability in the relationship before the survey interview might have led to their poor mental health. The observation period is also too short to test if unemployment is associated with dissolution. This is a potentially serious shortcoming because other research shows an association between union dissolution and unemployment (see Sections 1.4.1 and 1.5.4). Moreover, there may be unobserved variables not included in the analysis that would account for the association between receipt of IS and separation. Indeed, there could be other factors that lead some individuals to have low-incomes and to be more likely to separate.

#### 3.3 Overview

As outlined in Section 1.5.4, the Beckerian model of marriage underpins the studies discussed in this chapter. The theory posits that unions dissolve when the benefits of becoming single again exceed the costs. The studies tend to include data on

female and male earnings because of possible 'stability' and 'independence' effects. The former suggests that higher earnings make a partner more attractive and/or reduce levels of conflict within a union and so reduce the likelihood of dissolution. The latter, on the contrary, suggests that increased earnings mean that an individual is in less need of a partner in order to survive financially and so increases the risk of union dissolution.

Across the three countries considered here, the evidence for a welfare effect on union dissolution is mixed. To the extent that there is an effect it appears to be larger for certain sub-groups; the UK and Australia studies suggesting that the effect is greater in families with lower earnings. However, the small number of studies, the mixed findings, queries about the robustness of some of the studies and that the studies tend not to distinguish between dissolving marriages and cohabitations mean that there is no overwhelming evidence that welfare systems have a major impact on union dissolution.

The Bradbury and Norris (2005) and Zimmerman (1991) studies also serve to highlight the factors other than the welfare system that affect people's decisions to dissolve unions (and some of these factors are outlined in Section 1.5.4).

# 4 Childbearing and single parents

#### **Summary**

In the United Kingdom (UK) and the United States (US) there has been some concern expressed about the recent growth in non-marital births and single parenthood. One policy response in the US has been the introduction of the family cap, which seeks to limit further births by not paying extra benefit when women have children above a threshold. Several of the US studies reviewed in this chapter examine the impact of the family cap.

Of the three UK studies of Working Families Tax Credit (WFTC) reform, two find no significant impacts (Francesconi and van der Klaauw (2007); Francesconi (2007)); and the third an increase in birth rate for couples (Brewer et al. 2007).

The review found relatively few studies on fertility and moves into single parenthood at the European level. González (2005 and 2007) finds a positive and significant association between welfare benefits and the incidence of single motherhood. However, as González' (2007) analysis becomes more sophisticated the size of the impact of benefits on single motherhood and single headship becomes smaller and eventually statistically insignificant.

An Organisation for Economic Co-operation and Development (OECD) study modelling impact of social security (payroll) taxes for defined-benefit, Pay As You Go (PAYG) pension schemes on fertility finds that they do reduce fertility. However, whilst the PAYG pension scheme may not compensate parents other parts of the tax and transfer system may do so. In addition, individuals are not simply utility maximisers, hence it is possible that the 'unintended consequences' of PAYG will be less than their models suggest.

Continued

Findings from US literature on the impact of Aid to Families with Dependent Children (AFDC) on female headship show only small or no effects. Hoynes (1997) finds 'no evidence that AFDC has a significant effect on female headship decisions.' Lichter et al. (1997) find that many of the variables they analyse only have a relatively small effect on female headship, and argue that cultural changes underpin the observed increase in female headship. More recent studies that take into account unobserved heterogeneity at the level of the individual find that the welfare effect is not only small but statistically insignificant.

Early US studies suggest positive welfare effect on the fertility of unmarried women. But effects largely disappear when analysts control for unobserved state characteristics. Subsequent studies suggest minimal or no effect of welfare on fertility – with exception of family cap policies. The family cap is a mixed picture – most studies find little or no effect; a small number of studies report either a small positive effect (Sabia, 2006; Horvath-Rose *et al.*, 2008), or counter-intuitively a small negative effect.

#### 4.1 Introduction

In the UK and the US there has been some concern expressed about the recent growth in non-marital births and single parenthood (see Sections 1.4.1 and 1.4.3). This chapter reviews the literature on whether or not welfare benefit systems have an impact on women's fertility and living arrangements.

Having a child out-of-wedlock is one route to single parenthood; the others are union dissolution and widowhood. All of these routes into single motherhood are covered in this chapter (rather than elsewhere in the report). However, many of the empirical studies do not distinguish between the routes followed; much of the US literature is focused on female headship decisions which encompass fertility decisions by unmarried women as well as the other two routes to single parenthood. The literature can also distinguish between single mothers who live on their own with their child(ren) (that is, lone parents or single heads) and a wider group of single mothers who co-reside with a family member, such as a grandparent.

This chapter considers three UK studies (Section 4.2.1), three cross-national comparisons (Section 4.2.2) and several US studies (Section 4.2.3). With respect to childbearing many of the US studies focus on the family cap.

## 4.2 Research evidence

Much of the literature on the impact of welfare systems on single parenthood is from the US. As outlined in Section 1.6.1 these studies, typically, take advantage of state variations in benefits and Earned Income Tax Credit (EITC).

They tend to model the probability of being a single parent/female head or having a non-marital birth as a function of one or more of the following sets of variables:

- State characteristics, including the generosity of benefits. The analyses often allow for unobserved variables that might affect demographic behaviour (in the US literature this is known as state fixed effects).
- Individual demographic characteristics, including race, age, and educational attainment/number of years of schooling. Some analyses allow for unobserved variables at the level of the individual that might affect demographic behaviour (known as individual fixed effects).
- Economic characteristics of women, including female wage levels and labour market attachment. These variables reflect theoretical debates about whether there is an 'independence effect' whereby higher family income enables women to support children on their own or whether the lack of economic opportunities reduces the 'opportunity cost' of childbearing and so leads to non-marital births.
- 'Marriage market' characteristics, including local gender ratios, male earnings and employment prospects. These variables reflect the notion that the supply of marriageable men will affect the marriage behaviour of women.

In general, the results of the analyses are mixed – some find that the welfare system does significantly affect the formation of single parent families and non-marital births whilst others find no effect.

# 4.2.1 UK evidence on welfare systems and childbearing and single parenthood

Of the three UK studies that focus on the 1999 WFTC reform, two studies find no significant impacts and the third finds that the introduction did increase the birth rate for couples.

Francesconi and van der Klaauw (2007) use British Household Panel Survey (BHPS) data for 1991 to 2001 to explore the impact of the WFTC on the fertility decisions of single mothers and moves into lone motherhood by single childless women.<sup>38</sup> The analysis reveals **no** statistically significant effect of WFTC on either the birth rate of single mothers or on the transition into lone parenthood by single childless women. Although WFTC did reduce the likelihood of an additional child by single parents by 0.7 percentage points, which is a relatively large reduction of 20 per cent over the annual average birth rate (3.7 per cent), it is not statistically significant (Francesconi and van der Klaauw, 2007:26-27). Sub-group analysis by number and age of child(ren) also shows no significant impact of WFTC on the fertility rates of lone parent.

The authors also examine the impact of WFTC on employment, use of childcare and marriage rates. The analytical method is difference-in-difference. Further details of the study are provided in Section 1.6.3.

Francesconi and van der Klaauw (2007:27-28) also investigate whether WFTC, which was more generous to single parents than Family Credit (FC), provided incentives for single childless women to become unmarried mothers.<sup>39</sup> As mentioned previously, no significant effect was found. Indeed, WFTC had the opposite effect; it reduced the likelihood of single childless women forming lone parent households by 0.16 percentage points. Although statistically insignificant, this is a relatively large effect, a 15 per cent reduction over the average annual rate (1.3 per cent) of entry into lone parenthood for single childless women over the observation period.

That WFTC did not lead to a statistically significant increase in fertility suggests that the gains from working (and the reduction in childcare costs) outweighed the incentive effects to have extra children that arising from the credit being more generous than its predecessor, FC.

However, the findings for couples are mixed. The study by Francesconi *et al.* (2007) mentioned in Section 3.2.1 found no statistically significant effect of the impact of WFTC on either the birth rate of married mothers or entry into motherhood for married women without children (2007:37). Yet Brewer *et al.* (2007) find that the increase in the generosity of payments to families with children (through the introduction of WFTC and changes to IS) led to a 1.2 percentage points **increase** in births amongst couples with low levels of education (Brewer *et al.*, 2007:25). Equivalent to a ten per cent increase in births or nearly 20,000 additional births. The effect of the reform was also stronger for women who had not previously had children, and this may reflect that payments increased more for first births than for second or subsequent births. The effect was reduced for women for families where the youngest child was aged under three or aged eight or over, which may reflect preferences about the timing of children.

Brewer et al. (2007:29) argue that because the reforms did not have a significant effect on the woman's age at first birth, then the estimated increase in births represents an increase in total births rather than families bringing forward the timing of a birth that would have happened anyway.

That WFTC increased the birth rate for couples but not for single parents might be because, whilst WFTC increased employment for single parents (and so reduced fertility), for women in couples employment impacts were smaller (some women even reduced their hours of work) and so any effect on childbearing would be larger (Brewer et al. (2007:13-14).

The analysis is of entry into single parenthood by single childless women – moves into single parenthood by married women are not considered.

## Box 4.1 Further details - Brewer et al. (2007)

Brewer et al. (2007) use a difference-in-difference approach and data for women aged 20 to 37 years in couples (either married or cohabitating) from the Family Resources Study (FRS) and Family Expenditure Survey (FES) for 1995 to 2003. The differences compared are changes in fertility for low-income couples before and after the WFTC reform. For both before and after the reform, couples affected by the reform are compared with couples unaffected by the reform. The comparison with couples unaffected by the reform means that some control is introduced into the comparison for unobserved factors that vary over time. The low-income couples affected by the reform are defined as where the male and/or female partner left school at or before the compulsory school leaving age, whilst those affected by WFTC are defined as where both partners left school at age 18 or above. The analysis includes controls for socio-demographic variables (such as age, education, tenure, number of children and ethnicity) and male and female hourly wages.

# 4.2.2 Cross-national evidence on welfare systems and childbearing and single parenthood

Much of the analysis using data for the US is comparative in that it can incorporate differences between states – that is, regression models with state fixed effects.<sup>40</sup> In this sub-section, however, the focus is on cross national studies.

The review found relatively few studies on fertility and moves into single parenthood at the European level. Two studies by González (2005 and 2007) provide an analysis of the impact of welfare benefits on the incidence of single motherhood. The earlier study uses pooled data for 14 countries from five waves of the Luxembourg Income Study for the 1980s and 1990s.<sup>41</sup> The data were for all women aged 18 to 55 years. Estimates were made for two routes into single parenthood; those that were never married, and those who were divorced, separated or widowed. However, the analysis did not distinguish between cohabitating and married couples.

A number of the variables in the models represent women's expectations about their wages, hours of work, benefits and husbands earnings. In general, the values for these variables are the relevant averages for a given country and period. Thus expected benefits are the 'average received by each household type in a given country and period.' (González, 2005:16). The benefits include family-related

In regression analysis fixed effects are used to take into account omitted factors or unobserved heterogeneity that may affect the variables of interest. This gives more robust results, but involves assuming that the variables are time invariant, and requires variation amongst the observations for a variable.

The 14 countries are: Australia, Austria, Belgium, Canada, Finland, Germany, Hungary, Ireland, Israel, Italy, Luxembourg, Netherlands, UK and US.

benefits (like Child Benefit), social assistance benefits (for instance, IS) and in-kind benefits (like Food Stamps and childcare subsidies).

The study finds a positive and significant (p<0.001) association between welfare benefits and the incidence of single motherhood. A ten per cent increase in benefits paid to single parents (keeping benefits to other households unchanged) would **increase** the incidence of single mothers by five per cent (González, 2005:18). Other significant findings were:

- Higher male earnings (per annum) and employment rates reduced the incidence of never married mothers but increased the incidence of divorced, separated or widowed mothers. The overall effect on single mothers, however, was small and insignificant.
- Higher female wages (per hour) reduced the incidence of single mothers (for both never married and divorced, separated or widowed).

However, these findings should be treated with some caution. The Luxembourg Income Study is not a panel dataset and so the analysis does not control for unobserved variables at the individual level, and as a consequence may over-estimate the significance of the impact of welfare benefits on single parenthood. Moreover, data were not available for all waves for all of the countries; indeed, data are available for all five waves for only two countries (Luxembourg and the US). The analysis also excluded from consideration that unmarried mothers might marry and divorced mothers might remarry (González, 2005:9).

The second study (González, 2007) examines the impact of welfare benefits on the incident of single motherhood and single headship amongst women across 14 European countries. Here 'single motherhood' is defined as an unmarried woman aged 18 to 35 living with a dependent child aged under 18 years and not cohabitating with a partner, but they could be co-residing with other relatives such as a grandparent. 'Single headship' (or lone parenthood) is a sub-set of single mothers who live on their own with their dependent child – that is, there is no co-residency. Separate analyses were conducted for these two groups. Young women were examined because their family formation decisions are more likely to be influenced by benefit levels and labour market conditions.

The analysis uses pooled data from the European Community Household Panel (ECHP) for 1994 to 2001. The 14 countries, grouped by social protection system, are:<sup>42</sup>

- Anglo-Saxon countries: Ireland and UK;
- Nordic countries: Denmark and Finland;
- Benelux countries: Belgium, Luxembourg and Netherlands;
- Central European countries: Austria, Germany and France.
- Southern European countries: Italy, Spain, Greece and Portugal.

Of the EU15 countries Sweden is omitted because longitudinal data are not available in ECHP. Denmark is the omitted country in the regressions.

The main measure of benefits used in the analysis is the predicted benefit (for social assistance and family allowance schemes) for a single parent with two children who works 20 hours per week (González, 2007:398).

The reported analysis moves from simple descriptive analysis to logit regression, which becomes increasing advanced by incorporating country fixed effects and country-specific time trends, as well as individual fixed effects. The country fixed effects control for unobserved factors at country level (such as, public tolerance of single parenthood) that might affect the level of benefit and prevalence of single mothers. To allow for unmeasured changes over time that were correlated with changes in welfare benefits the analysis includes time trends (that vary by either individual counties or by the groups of countries listed previously).

The analyses also include individual-level controls – age and educational level – that seek to measure the respondents' labour market prospects, attractiveness as a partner, and preferences for marriage and children, plus country-level controls – male unemployment rate and median adult male wage rate – as indicators of labour market conditions (González, 2007:398-399). As the author notes (González, 2007:410) more detailed variables on the labour market, education and the 'marriage market' would be 'desirable'.

The descriptive analysis, a plot of the proportion of single mothers against predicted benefits levels for each country for each year, shows that '... countries with higher benefit levels also have higher incidence of single mother households.' (González, 2007:400). (The correlation coefficient is 0.34.) The logit regression analysis allows for changes across countries and over time to be taken into consideration (González, 2007:403-407). Having controlled for unmeasured factors that could lead to changes in benefits and the incidence of single motherhood at different rates in different countries or groups of countries, González (2007:394) finds that '... an increase in yearly benefits to single mothers of €1,000 would increase the incidence of these households by about two percent.'<sup>43</sup> This single motherhood is due to both out-of-wedlock childbearing by never married women and divorce by married women with children González (2007:406-408).

González (2007:405 and 406-407) also finds that less educated women are more likely to become single mothers and single heads of households.

Essentially, as the analysis becomes more sophisticated – that is, takes into account more possible unobserved heterogeneity – the size of the impact of benefits on single motherhood and single headship becomes smaller and eventually statistically insignificant. It is the inclusion of controls for unobserved factors at the level of the individual that reveals no statistically significant impact of benefits on the incidence of single mothers. Rather than saying that the analyses with significant positive findings might be spurious, the author attributes this insignificance to the lack of variation in the data, which in turn is due to the further disaggregation of the data.

If country and time effects are not taken into account the impact of benefits on single mothers is higher. An increase in yearly lone parenthood benefits of €1,000 increases the incidence of single mothers by about 18 per cent.

That benefits have no significant impact on US female headship decisions once individual effects are included in models is also a finding of Hoynes (1997a) (see ahead for further details). This may mean that the positive effect mentioned previously'... may be attributable to unobserved, individual-specific heterogeneity.' (González, 2007:409). However, González (2007:409) queries the need to include individual fixed effects in the analysis arguing that migration across European countries is less than across US States and hence their inclusion is less important. In addition she points out that significance levels will fall because of the reduced number of cases available for the analysis.

The significance of the González (2007) study is not simply the reported positive effect of benefits on single mothers but that it highlights how data limitations give rise to interpretative and analytical challenges. Although the total number of observations is 172,437, for the regression analyses with fixed effects there is little variation in the benefits variable once other variables are introduced (González, 2007:410). The analysis does not capture long-term trends in benefits or the incidence of single mothers; indeed, the average number of observation periods per respondent is six.

A third study by Ehrlich and Kim (2007) models the impact of social security (payroll) taxes for defined-benefit, PAYG pension schemes on fertility and family formation across the OECD (the findings for the latter are discussed in Section 2.2.2). They use two approaches to assess this effect: a simulation model and regression analysis. The simulation model shows that a one per cent increase in defined-benefit, PAYG taxes reduces the family fertility rate (that is the number of children per parent) by 0.075 per cent and the total fertility rate by 0.306 per cent.<sup>44</sup> Moreover, social security taxes for defined-benefit, PAYG schemes accounted for 48.1 per cent of the fall in the total fertility rate in OECD countries between 1965 and 1989 (Ehrlich and Kim, 2007:9). The regression analysis shows that social security taxes for pensions reduce total fertility rates for all women aged 15-49 (averaged over a five year lead period) by 0.284 per cent and the family fertility rate by 0.119 per cent (Ehrlich and Kim, 2007:17).

The authors contend that if a parent has a lower (or zero) rate of fertility this does not affect their pension, as the benefits are defined by the system and are the same as for parents with many children:

'Since defined benefits are independent of contributions made by children, parents are not compensated individually for raising more or better-educated children. As a result, PAYG payroll taxes induce behaviour that is not socially optimal: They diminish the incentive of individual workers to bear and invest in children, save for retirement, or generally form families altogether, because they lower the private rewards from family investments relative to alternative individual pursuits.'

Ehrlich and Kim (2007:4)

Formally, the social security tax rate used is the old-age, survivor and disability-insurance proportion of social security benefits as a share of Gross Domestic Product (GDP) (Ehrlich and Kim (2007).

However, whilst the PAYG pension scheme may not compensate parents, other parts of the tax and transfer system may do so. In addition, individuals are not simply utility maximisers, hence it is possible that the 'unintended consequences' of PAYG will be less than their models suggest.

# 4.2.3 US evidence on welfare systems and childbearing and single parenthood

Out of wedlock childbirth and single parenthood have been perceived to be problems by policy makers and others in the US. Indeed, an 'explicit goal of policymakers in drafting welfare reform policies was to reduce incentives for non marital childbearing.' (Horvath-Rose et al., 2008:119-120). Section 101 of the Personal Responsibility and Work Reconciliation Act of 1996 (PRWORA) see Section 1.4.3) states 'Congress makes the following findings:

- The increase in the number of children receiving public assistance is closely related to the increase in births to unmarried women.
- The negative consequences of an out-of-wedlock birth on the mother, the child, the family and society are well known.
- Therefore, in the light of the demonstration of the crisis in our Nation, it is the sense of the Congress that the prevention of out-of-wedlock pregnancy and reduction in out-of-wedlock birth are very important Government interests and the policy ... is intended to address this crisis.'

Horvath-Rose et al. (2008:120)

PRWORA granted some flexibility to states to decide on the conditions of entitlement to welfare benefits (Ryan et al., 2006:106). For example, it allowed states to decide how long a new mother could remain out of the workforce following child birth. Under the Job Opportunities and Basic Skills (JOBS) programme introduced by the Family Support Act in 1988 mothers with young children were usually exempted from the requirement to work until the child was three years old. However, under Temporary Assistance for Needy Families (TANF) two states did not provide any exemption at all so that mothers of newly born babies could be required to work immediately whilst in a further six states only women with a child under six months was exempted from the requirement to return to work. In the other four, mothers were not required to work until their child was two years old. According to Ryan et al., (2006:105) the limited availability of affordable childcare in many areas makes these obligations 'quite burdensome'.

The family cap is one of three broad government initiatives under TANF that directly target the prevention and reduction of out-of-wedlock pregnancies. The other two are the establishment of a 100 million dollar incentive each year for five years divided amongst the top five states with the largest decline in the illegitimacy ratio of abortion to live births (US congress, 1996 cited by Ryan *et al.*, 2006:104) and an Abstinence Education Grant Program (AEGP) (Camasso, 2004:454). The family cap, which was also a feature of some AFDC waiver programmes, was

specifically designed to curb the number of children women could afford to have by reducing benefits for children born while the mother is in receipt of welfare benefits (Horvath-Rose *et al.*, 2008:120) thus ending the traditional practice of providing families on welfare with additional benefits in respect of a newly born child (Kearney, 2004:295). Nineteen states adopted family cap policies prior to PRWORA and a further six included family caps as part of their TANF programmes (Horvath-Rose *et al.*, 2008:120). Five states have introduced partial family caps either by reducing but not eliminating additional benefit for a new born child or paying the increase in-kind or to a third party (Kearney, 2004: 39).

There are a number of US based studies that have investigated the impact of welfare benefits on single parenthood and fertility. The empirical studies can be roughly divided between pre- and post- welfare reform periods (Joyce *et al.*, 2004:477). Crudely, if there is a welfare effect then analyses of AFDC recipients can be expected to show an increase in fertility rates and single parenthood and the welfare reform studies (of AFDC waivers and TANF) a fall in non-marital births and single parenthood (reflecting policy aspirations like the imposition of the family cap).

## AFDC and fertility and single parents

Much of the literature on the impacts of means-tested programmes on fertility and single parenthood focuses on AFDC. Underpinning the US analyses of AFDC is a popular view that the programme encourages childless single women to have children (Moffitt, 2003:332). Under AFDC a women having a child outside of marriage was eligible for the benefit, but typically was not if married (Moffitt, 2003:334). Thus, it was believed that AFDC included a financial incentive for non-marital childbearing. Furthermore, in some States the amount of benefit was related to family size with larger sized families receiving more benefits and this might create a further incentive to have children. However, during the 1990s real expenditure on AFDC was falling and so does not really explain the increases in non-marital childbearing and single parenthood.

#### Methodological issues

The early US literature on the impact of AFDC on female headship was mainly based on state-level data and used non-experimental designs. The findings from these analyses are mixed, and analysts like Hoynes (1997b:129) conclude that they 'find no evidence that AFDC has a significant effect on female headship decisions.' (The Hoynes (1997b) analysis is discussed ahead.) Later studies controlling for unobserved variables across states do find significant and positive effects on female headship. However, the size of the effect is small. Moffitt's review (1998b) suggests that more generous AFDC benefit was associated with higher levels of single parent families, but the magnitude and significance of the association was uncertain. For the models that control for omitted state variables, Moffitt's (1998b:74-75) review finds significant and insignificant welfare effects. More recent studies that take into account unobserved heterogeneity at the level of the individual find that the welfare effect is not only small but statistically insignificant.

A similar conclusion applies to non-marital births. Much of the work in the welfare and fertility literature relies on cross-state generosity of welfare benefits paid to families with children. Early cross-sectional work suggests that AFDC had a positive effect on the fertility of unmarried women. However, Hoynes (1997a) finds that those effects largely disappear when controls for unobservable state characteristics are included. And when the study controls for individual effects as well as state effects there is no evidence that welfare increases the tendency to form female headed households for either black or white women (Kearney, 2004:299).

Klawitter et al. (2000) use data from the youngest cohorts of women in the National Longitudinal Survey of Youth (NLSY) to estimate models of the determinants of the initial use of AFDC. The authors claim to be one of the few studies to address the timing and determinants of initial AFDC participation. Klawitter et al. (2000) find little evidence that that financial, or other welfare incentives, affect the likelihood or timing of first entry onto welfare. Klawitter et al. (2000) find that whilst benefit levels do not appear to affect participation, the presence of a programme for medical aid to families who are not in receipt of welfare appears to affect the entrance on to AFDC for some groups. The authors suggest that this finding 'suggests that some needy mothers may stay off AFDC if they have access to an alternative source of health care assistance.' (Klawitter et al., 2000:543). Also, significant for AFDC participation are parental poverty, family structure, educational achievement and race (Klawitter et al., 2000:527). The authors point out that school-related factors are prominent amongst the significant variables and:

'... may provide the best path for lower welfare participation rates and brighter future economic outcome ... policies aimed at keeping young women in school might provide alternatives to early childbearing and skills that allow older mothers to earn enough to support their families.'

Klawitter et al. (2000:544).

Overall, non-experimental methods show that any effects of AFDC on non-marital childbearing are insignificant, but if significant, are small.

Findings that show a welfare effect also find differences by race. For example, Lundberg and Plotnick (1990; 1995) find effects of welfare on pregnancy and abortion for white teenagers but not for black teenagers. They suggest that 'the different racial results may reflect ... important unmeasured racial differences in factors that influence fertility and marital behaviours' (Lundberg and Plotnick, 1995:177). However, Arcs (1994; 1996) analysing data from the NLSY found no effect of welfare for either black or white women. Duncan and Hoffman (1990) rely on cross state comparison using the Panel Study on Income Dynamics (PSID) and do not find a significant effect of AFDC on births for black teenagers while An et al. (1993) also undertake cross state comparisons using PSID data and do not find a significant effect on the probability that a woman aged 13 to 18 has a non marital birth (discussed by Kearney, 2004: 299). However, as Kearney (2004: 299) observes, a weakness of these and other studies that rely on cross state

comparisons is that the findings are potentially biased by unobserved differences between states (see Appendix B).

Studies of AFDC, fertility and single parents

More recent studies of AFDC's effect on fertility continue to give mixed results. For example, Rosenzweig (1999) and Hoffman and Foster (2000) find that AFDC **increased** non-marital births (at least for sub-groups), whilst Fairlie and London (1997) and Acs (1996) reveal **no** significant welfare effect on childbearing.

Rosenzweig's (1999) assesses the role of AFDC and marital prospects in the fertility and marriage choices of young women in the context of a model incorporating heritable characteristics that determine marriageable prospects. The analysis is of the fertility and marital experience up to the age of 23 of eight birth cohorts of women in the NLSY. In the analysis Rosenzweig (1999) controls for both unobserved variables at state level and for cohort effects. Rosenzweig (1999) finds a significant and large positive effect of AFDC on non marital childbearing:

'Higher AFDC benefit levels and lower marital prospects induce young women to choose to have a child outside of marriage.'

Rosenzweig (1999:S3)

Hoffman and Foster (2000) use data from the PSID to replicate Rosenzweig's (1999) analysis and explore the reasons as to why his findings differ from previous research. They are able to reproduce his main findings in a model that includes state and cohort fixed effects. Controlling for fertility separately by age shows no effect on teen non-marital births, but a large effect on the behaviour of women in their early 20s (Kearney, 2004:299).

On the other hand, Fairlie and London (1997) examine the relationship between incremental AFDC benefits and fertility using the 1990 panel of the Survey of Income and Program Participation (SIPP) and find a positive, although statistically insignificant, association between incremental AFDC benefits and fertility among women in receipt of AFDC. However, they also find a similar association for several non-recipient comparison groups including married women and single women who were not in receipt of AFDC. They conclude that the association found for AFDC mothers is spurious (Fairlie and London, 1997:575).

Arcs (1996) also finds no welfare effect. Arcs (1996) examines the relationship between AFDC and births to women who already have a child, using data on young mothers from the NLSY. Arcs (1996) finds that variations in benefits have no statistically significant impacts on the subsequent childbearing decisions of young mothers in general or of women who received welfare in particular. Furthermore, mothers who received welfare to support their first child are no more likely to have additional children on welfare up to the age of 23.

With respect to single mothers, and as mentioned in Section 1.4.3, increases in the proportion of single parents have accompanied falls in the real value of AFDC. Moffitt (2000) uses time series analysis to investigate if changes in wage rates can account for this increase in female headship. Using Current Population Survey (CPS) data from 1968 to 1996 for all men and women aged 18 to 65 with less than 12 years of schooling he shows that changes in wage rates do help explain the increase in single motherhood. Whilst real wage rates for less educated women stagnated over this period, those for less educated men declined sharply. This increased the female to male wage ratio, and – according to a Beckerian model of marriage – reduced the gains from marriage. This fall in the returns to marriage 'outweighed the effects of the decline in welfare benefits.' (Moffitt, 2000:376). It also follows that the decline in marriage is not a consequence of these women becoming more financially independent. The results are stronger for white women and men, but still 'suggestive' for the African-American population.

As mentioned elsewhere in this report, findings can be sensitive to the analytical approach adopted. Moffitt (1998b:61-65) shows how relatively small changes in the data and approach used can influence the assessment of the welfare effect on rates of single parenthood. Using CPS data for 1993 he plots the proportion of white single mothers in each state against real monthly AFDC benefits for a family of four with no other income, and fits a simple least-squares regression line. This reveals only a very weak positive relationship between the two variables. However, when the dependent variable is modified to women aged 20 to 44 without a high school diploma (a sub-group with high participation rates in AFDC), a stronger positive relationship is found. Similar results are found if never-married single mothers are correlated with benefit amounts. This positive correlation arises because many states with generous benefits had high rates of single motherhood (such as, New York, California and Illinois). Using data for other years produces similar results.

To explore whether changes in benefits over time have an impact, data for levels of benefit and rates of single motherhood in 1970 are compared with those for 1993. This helps to explore whether, for instance, the increases in the rates for white single mothers were greater in those states that reduced, in real terms, their benefits the least. Such a result would confirm the simple cross-sectional benefit level analysis. However, for this change comparison the relationship between the two variables is 'weak', because the increases in the rates of white single mothers were fairly even across the states. <sup>46</sup> The difference between the two sets of analyses might be because other factors that affect benefit levels and the incidence of single parenthood are omitted from the analyses.

Single mother rates are calculated as a fraction of all women aged 18 to 64. Possibly a better measure would have been to use women aged 18 to 45, and so taking out of the base older women who were less likely to have children.

<sup>&</sup>lt;sup>46</sup> Similar findings were found for black women (Moffitt, 1997b:66).

The role that the analytical method adopted can play in these studies is emphasised in the study by Hoynes (1997a), who examines whether the welfare system (ADFC and Food Stamps) affects the incidence of female headship, and finds that it does not.<sup>47</sup> Unlike previous studies her analysis includes individual effects as well as state effects, year effects and controls for state, women and welfare characteristics.<sup>48</sup> This leads Hoynes (1997a) to conclude that earlier studies over-estimate the effect of welfare benefits on family structure:

'... omitting individual effects can lead to a bias through differences in the composition of states over time.'

Hoynes (1997a:111)

By implication, Hoynes' (1997a) conclusion not only applies to studies of single headship but more widely to US studies of marriage and divorce that have excluded individual effects.

## **Box 4.2** Further details – Hoynes (1997a)

Using a nationally representative longitudinal dataset, the Panel Study of Income Dynamics for 1968 to 1989, Hoynes (1997a:98-100) develops separate linear regressions for white women and black women. The dataset comprises women aged 16 to 50 who are married or heads of households – the data exclude co-resident single mothers. In the models the controls for the woman's characteristics include age, education and religious affiliation. The PSID data are matched with other state data.<sup>49</sup> The generosity of the benefit system is measured by the total value of benefits from AFDC and Food Stamps for a family of four with no other income for each state. A second welfare benefit variable is whether or not the state offered AFDC-UP. The other state variables included the unemployment rate, average wage in manufacturing, per capita income, percentage of the population aged over 65, percentage of population that are children, party of the State governor, and proportion of the State House and Senate that are Republican.

Like the later study by González (2007) mentioned previously, Hoynes finds that as more controls for unobserved heterogeneity are introduced, the size of the welfare effect diminishes and eventually becomes insignificant. That is, there is no evidence that AFDC and Food Stamps influence women's decisions on forming single person households.

- A female head is defined as a woman who is unmarried and has a child the study counts cohabiting females as non-heads of households. An earlier version of this paper is included in the Moffitt (1998b) review.
- Year effects are included to control for any common social trends that might influence marriage decisions.
- Following Becker's model of marriage, in the model the woman's earnings and those of her potential spouse which influence the net benefits of female headship and are difficult to observe directly were accounted for by the woman's characteristics and labour market variables.

More specifically, Hoynes (1997a) reports that:

- Using the data as a cross-section (to take advantage of the variation in state's benefit levels) and not controlling for state or individual effects there is a positive and statistically significant welfare effect on female headship. For white women, AFDC and Food Stamps had a modest and significant effect on headship decisions (Hoynes, 1997a:104). For each \$100 increase in the value of the benefits, female headship increased by 0.9 percentage points (a ten per cent increase). However, living in an AFDC-UP State did reduce the economic benefit of being a sole head of household a \$100 increase in the benefit was associated with a 0.8 percentage point fall in female headship. For black women the welfare effect was larger a \$100 increase in benefits increased the likelihood of single headship by 1.9 percentage points; and in a AFDC-UP state it increased by five percentage points (Hoynes, 1997a:108-109).
- Controlling for state effects makes the relationship for white women close to zero and insignificant, but essentially unchanged for black women. For white women this finding is:

'... consistent with the idea that unmeasured state effects influence white headship decisions and welfare benefits. For example, a state may have a strong two-family tradition which results in fewer female headed households and less support for the AFDC program. Not taking into account state effects attributes this difference in preferences to a welfare incentive.'

Hoynes (1997a:106)

However, controlling for state effects has 'little effect' on the relationship between benefits and female headship for black women - a \$100 increase in benefits increases the probability of female headed households by 1.1 percentage points (Hoynes, 1997a:108-109).

Adding individual effects shows that there is no evidence that benefits increased the likelihood of either black or white women forming female headed-households (Hoynes, 1997a:106 and 110). In particular omitting individual effects for black women leads to an over-estimate of the welfare effect on female headship decisions. This might be because on average black women moved to states with higher benefits (possible to take advantage of better labour market opportunities), whilst white women, on average, did not.

Dickert-Conlin and Houser (1999) using data from the Survey of Income and Program Participation and controlling for fixed effects also find that there is no significant correlation between AFDC and female headship.

A different approach on the growth in single parenthood during the 1980s is provided by Lichter *et al.* (1997), who use state and county level data and consider the affect of cultural factors as well as welfare benefits and economic opportunities. Lichter *et al.* (1997) use longitudinal county level data from the 1980 and 1990 decennial censuses to estimate the effect of changes in benefits (a package

comprising AFDC, Food Stamps and Medicaid) on unmarried female headed households with children aged under 18. The use of county level data means that intra-state differences in labour and marriage markets can be explored.

## Box 4.3 Further details – Lichter et al. (1997)

The analysis by Lichter *et al.* (1997) includes fixed effects to control for unobserved variation at state and county levels. The data is for all US counties in the 48 contiguous states. Findings are reported overall and separate analyses were conducted for non-Hispanic whites, non-Hispanic blacks and Hispanics.

The economic attractiveness of males was measured using the sex ratio, male employment rate, male earnings for full-time workers and percentage of college educated men in the county. Measures of female economic independence included median earnings of women in full-time employment and percentage of college educated women in the county. The AFDC component of the overall generosity of the benefits package represents the maximum benefit level of a family of four with no other income. To capture the cultural context of family formation, Lichter *et al.* (1997:123) use measures of urbanisation (because non-metropolitan women tend to marry earlier and are less likely to have children outside of marriage) and participation in 'pro-family' religious organisations. Other control variables include age and racial/ethnic composition of each county.

Lichter et al. (1997) find that many of the variables they analyse only have a relatively small effect on female headship, and argue that cultural changes underpin the observed increase in female headship. More specifically, they find that:

- There was a **small** welfare effect. A \$100 change in welfare benefits is associated with a 0.838 percentage points change in county female headship (Lichter *et al.*, 1997:129). This implies that over the 1980s the fall in the value of welfare benefits (principally AFDC) decreased female headships by 1.4 per cent. By race/ethnicity this welfare effect was significant and positive for black Americans but statistically insignificant for non-Hispanic whites and Hispanics (Lichter *et al.*, 1997:132).
- There was a small marriage market effect, especially amongst black Americans and Hispanics. The gender ratio was negatively associated with female headship

   a decline of 13 men per 100 women would increase female headship by one percentage point (Lichter et al., 1997:129-130). Men's economic opportunities were negatively associated with female headship a ten per cent change in male employment was associated with a 1.1 percentage point change in female headship; and a ten per cent change in male earnings with a 0.3 percentage point change in female headship.
- There was **no** support for women's economic independence being associated with female headship as women's earnings and education were negatively associated with female headship (Lichter *et al.*, 1997:130). Thus women with higher levels of schooling were potentially more attractive as marital partners.

• Cultural and demographic factors were significant and affected female headship (Lichter *et al.*, 1997:130-131). Female headship was positively associated with urbanisation, high concentrations of black people and young people and falls in local population, and negatively associated with 'pro-family' factors (the proportions of Hispanics, Catholics and anti-abortion Protestants in a county).

Lichter et al. (1997:131) argue that factors other than changes in welfare benefits and changes in the composition of the counties must explain the increase in female headship. They highlight cultural changes such as changing societal values towards sexuality and unmarried cohabitation, and that society has become more individualistic. Overall, Lichter et al. (1997:136) conclude:

'... welfare was not the primary or even a key factor responsible for the [then] recent upswing in female headship. The estimated welfare incentive effects implied that even very large cuts in welfare payment levels produce only a relatively small drop in the rate of female family headship.'

Two more recent studies report similar findings. Blau *et al.* (2004) examine the effects of welfare on single motherhood and family headship using Metropolitan Statistical Area (MSA) data and controlling for unobserved variables. They find no impact of welfare benefits on single motherhood for either white or black women and a positive effect on single headship for black but not for white women (Blau *et al.*, 2004:382).

The authors point out that while fixed effects methods improved on early cross-sectional analyses of welfare benefit levels and family formation, fixed effects methods may still give biased findings if there are unmeasured changes in factors, for example norms, correlated with changes in benefits (Blau *et al.*, 2004:383). For example, a state where the reduction of stigma of being a single parent is reducing may increase the level of welfare benefits at the same time as the number of single parents is increasing. Blau *et al.* (2004:384) argue that characterising some states as 'Liberal' as fixed effects models do misses changes in economic and social climate.

Blau *et al.* (2004) find positive cross-sectional associations between benefits and single motherhood and family headship that are consistent with early reported cross-sectional analyses reported previously. However, in line with Moffitt (1994) and Hoynes (1997) the authors find that when they add the controls for unobserved MSA variables the impact on black women reduces and that for white women disappears entirely (Blau *et al.*, 2004: 384). Adding MSA fixed effects removes a positive welfare effect on single motherhood for all groups. When the authors go further and include MSA time trends into the analysis they show a positive effect on female family headship for young black women with some evidence of a larger effect on less educated black women. However, there is still no evidence of a positive effect on either single motherhood or single headship for white women or for single motherhood for black women (Blau *et al.*, 2004:384).

The authors conclude that these findings suggest that for black women and in particular those who are less educated, restricting welfare benefits is likely to increase extended family arrangements without reducing the incidence single parenthood, which they suggest lends support to 'one of the most robust results obtained in this literature' reported by Danziger et al. (1982) and Ellwood and Bane (1985) that welfare changes mainly impact on the living arrangements of single mothers but not the incidence of single motherhood itself (Blau et al., 2004:384).

Similarly, Fitzgerald and Ribar (2004:189) find little consistent evidence that waivers affected female headship of families.

# Box 4.4 Further details – Fitzgerald and Ribar (2004)

Fitzgerald and Ribar (2004) use pooled panel data from the Survey of Income and Program Participation for years 1990, 1992, 1993 and 1996 to assess the impact of welfare waivers and TANF on women's decisions to form single headship families. The authors claim to be the first observational study to use nationally representative, individual level longitudinal data. While acknowledging the limitations imposed by the short duration of the available panel data the data nevertheless allow the authors to analyse incidence and transitions into and out of female headship. They also developed annual, county-level information on wages and jobs and marriage market conditions as well as event history methods to control for other relevant unobserved factors.

#### AFDC waivers

There have been several studies of the effect of AFDC waivers in particular the impact of family cap policies on women's fertility decisions. The family cap was explicitly designed to reduce additional births to women in receipt of welfare by reducing or removing incremental increases in benefit for additional children. Effectively, it reduces the amount of benefit paid for an additional child born on welfare to zero and so increases the marginal cost of bearing another child for a woman on welfare. Thus, other things being equal, a family cap raises the price of an additional child and is expected to decrease pregnancy rates and increase abortion rates (Kearney, 2004:303). The policy might also be expected to encourage a pregnant woman to marry before giving birth if the husband's income substitutes for lost benefit income (Baughman and Dickert-Conlin, 2007; Sabia, 2006:114). With the exception of the family cap, studies subsequent to Moffitt's review (1998b) suggest a minimal or no effect of welfare policies on fertility. There are of course counterexamples.

#### Family cap

There is a mixed picture with respect to the family cap policy. Whilst most studies of the impact of the family cap find that there is either little or no effect and some, counter intuitively, that there is a negative effect, a small number of studies

report positive effects. However, methodological concerns have been raised with each of the studies that find positive effects, either by the authors themselves or by critics. Therefore the findings of these studies need to be treated with caution. The overall picture is that even the family cap has had little or no effect on fertility, although some authors have pointed out that it is likely to have had the effect of increasing poverty within families and levels of hardship for children (Kearney, 2004; Levin-Epstein, 2003 cited by Ryan *et al.*, 2006).

Examples of studies that find no effect of the family cap on childbearing include Joyce, et al. (2004), Dyer and Fairlie (2004), Kearney (2004), and Ryan et al., (2006) and these are discussed ahead.

Joyce et al. (2004) analyse the effects of the family cap on birth and abortion rates (Joyce et al., 2004:475 and 481). The authors argue that widening the analysis of the effects of welfare reform to include abortion as well as birth provides further important information on changes in women's reproductive behaviour associated with welfare reform. For example, as discussed ahead, Camasso et al. (2003) and Jagagnnathan and Camasso (2003, cited by Joyce et al., 2004:476) find in an experimental analysis of New Jersey's family cap that a fall in birth rates associated with the family cap was accompanied by an increase in abortion rates. Abortion rates are of interest because rates of unintended pregnancy and abortion are high amongst unmarried women (Joyce et al., 2004:475; 476). Henshaw (1998) estimates that 75 per cent of all pregnancies to unmarried women and 75 per cent of all pregnancies to teens, are unintended (Henshaw, 2004, cited by Joyce et al., 2004:476). Whilst an evaluation of Delaware's waiver programme, A Better Chance, found that 82 per cent of women in receipt of welfare that had a baby during the trial said that the pregnancy was unintended (Fein, 1999 cited by Joyce et al., 2004:504). Joyce et al. (2004) suggest that the high prevalence of unintended pregnancies amongst women in lower socioeconomic statuses leads to the expectation that a substantial decline in births associated with the family cap should also be associated with an increase in abortion (Joyce et al., 2004:475).

## Box 4.5 Further details – Joyce et al. (2004)

The authors claim that their analysis is distinct in that, whilst most previous analyses of the effect of welfare reform on fertility examined only births, theirs is the only econometric study that analyses both birth and abortion rates amongst women who are at risk of receiving public assistance (Joyce et al., 2004:476). The study uses records from 24 states and a difference-in-difference design. The study uses a within – state design to compare unmarried adult women who have 12 or less years of schooling with no previous live births with similar women who have one or more births. The authors suggest that while both groups of women have a high likelihood of being affected by welfare reform only the latter group should be affected by the family cap (Joyce et al., 2004:477).

#### Box 4.5 Continued

The authors explain the need for a within-state comparison by reference to the limitations of multi-state studies of the family cap where controls for other aspects of welfare reform are included in the regressions or the study period is limited to pre-TANF years (Dyer and Fairlie, 2004; Hovarth, Rose and Peters, 2001). A potential limitation of these studies, the authors suggest, is that it is difficult to characterise and measure the differences in complex multi-dimensional welfare reform programmes that have been adopted by different states and which are enforced to varying degrees (Moffitt, 2003 cited by Joyce et al., 2004:477).

Joyce et al. (2004) report that in states with family caps birth rates fell and abortion rates rose more amongst high risk women with at least one previous live birth compared to similar childless women, a finding which they suggest is consistent with the expected effect of the family cap. However, this parity-specific pattern of births and abortions also occurred in states that implemented welfare reforms without a family cap. Thus, the effects of welfare reform may have differed between mothers and childless women, but there is **little** evidence of an independent effect of the family cap (Joyce et al., 2004:475). Joyce et al. (2004:504) conclude that 'there has been no major reproductive responses to the family cap'.

Dyer and Fairlie (2004) compare changes in fertility across family cap and non-family cap states and find **no** consistent evidence of an association between the family cap and the probability of an additional birth while in receipt of welfare.

## **Box 4.6** Further details – Dyer and Fairlie (2004)

Dyer and Fairlie (2004) use a national sample of mothers who ever received welfare drawn from CPS micro data between 1989 to 1999 to examine the impact of AFDC waiver family cap policies on the birth rate of single less educated women with 12 or less years of schooling and at least one child.

The study uses the first five states that were granted AFDC waivers to implement family caps as 'natural experiments'. They employ several techniques to increase the credibility of the experiment, specifically multiple comparison groups, controls of differential time trends, and difference-in-differences estimators.

They develop the analysis by using married women with children and with 12 or less years of schooling as a within-State comparison group. Again they find **no** consistent association between the family cap and non marital fertility. However, in a separate analysis of New Jersey's family cap policy the authors report a two percentage point decline in the fertility rate (relative to a mean of six per cent) associated with the family cap, although this decline is statistically insignificant at conventional levels.

The authors conclude that:

'... our general lack of evidence of a negative effect of the family cap and other waivers on fertility do not support the hypotheses that family cap policies reduce the incidence of out-of-wedlock birth.'

Dyer and Fairlie (2004: 441)

The authors suggest that effects of family cap policies may be limited for a number of reasons: because increments in benefit levels are much lower than the estimated costs of bringing up a child, many welfare spells are short, the importance of non-monetary factors, the unintended nature of some pregnancies and the partial offsetting of lost benefits by Food Stamps and Medicaid (Dyer and Fairlie, 2004).

Kearney (2004) also uses the variation across states in the timing of the implementation of the family cap to identify whether the denial of an increase in benefits for children leads to a reduction in births.

## Box 4.7 Further details - Kearney (2004)

Kearney (2004) claims to improve on previous cross-State studies in a number of ways through using Vital Statistics Natality Data compiled by the National Centre for Health Statistics (NCHS) which include all births in the US from 1989 to 1998. Kearney (2004) limits the sample to births to women age 15 to 34. She exploits the cross-state variation in the timing of policy implementation to compare the change in birth rates for a state that implements a family cap to those that do not while controlling for level differences in birth rates across states and years as well as differences in linear birth trends across states (Kearney, 2004:296). This allows the analysis to control for the effects of time and state fixed effects and observe whether results extend across race and age groups (Kearney, 2004:300).

However, Kearney points out that her analysis contains only limited post-family cap data as most states that introduced a cap did so in 1995-6 and at the time of her study vital statistics for births was only available until 1998. Kearney points out that it is possible that the effects of the family cap may take longer to show so further research is needed to investigate longer term effects (Kearney, 2004:318).

Kearney finds **no** evidence that family cap policies led to a reduction in births to women aged 15 – 34. She also finds no evidence of large declines in higher order births amongst groups with relatively high welfare participation rates. Kearney (2004) suggests that family cap policies are at best ineffective and at worst misguided as fewer state resources are being provided per child.

Contrary to expectations Kearney found counter-intuitive evidence that welfare reform had a small **positive** association with increased fertility about one year after the family cap was introduced for unmarried women with low education. This finding was also supported by Joyce *et al.* (2004; cited in Ryan *et al.*, 2006:107).

Further evidence that welfare receipt has minimal effect on fertility is provided by Ryan *et al.* (2006). Using discrete time event history analyses of data drawn from the PSID, they examine the association between the risk of a non marital subsequent birth and State level welfare waiver policies implemented under AFDC. The authors point out that while many studies have treated welfare policies as a group, their aim is to assess the effect of each state policy – family cap, time limits, work exemptions, job sanctions, earnings disregards, work requirements - separately in order to determine which, if any, were effective in reducing subsequent non marital childbirth as '... grouping all waivers together may obscure the true influence of specific policies.' (Ryan *et al.*, 2006:104).

## Box 4.8 Further details – Ryan et al. (2006)

Ryan *et al.* (2006) conduct discrete time event history analyses of data drawn from the PSID. PSID is a nationally representative longitudinal survey of US residents and their families (Ryan, 2006:109). The authors point out that PSID is the only data set that can provide the necessary longitudinal information on public assistance receipt, childbearing behaviours, relationship transitions, and individual characteristics. Ryan *et al.* (2006) combined the Marriage History and the Childbirth and Adoption History files with the main Family and Individual files to build complete marriage and childbearing histories for each female respondent (Ryan *et al.*, 2006:109). The data also provide state of residence annually allowing the authors to include state-level economic variables in the analysis (Ryan *et al.*, 2006:109).

The authors argue that a key strength of their research is the use of national level data on fertility behaviours of those who could be expected to be most affected by the changes – unmarried mothers who have received welfare benefits. They argue that conceptually this is an improvement over those studies that use only aggregate state data or single state specific data or those that use national data but do not identify the same high risk population (Ryan *et al.*, 2006:109). The only time invariant variable the authors used was race/ethnicity (non-Hispanic whites vs. Hispanics and non-Hispanic blacks) (Ryan *et al.*, 2006:110).

#### The research finds that:

'... although reducing the number of non-martial births is a key goal of welfare reform, state established welfare waiver policies do not operate as intended – they did not have any influence on women's childbearing behaviours in the sample, net of women's individual characteristics and state economic environments. Even the family cap policy, which was designed for the sole purpose of reducing additional births, had no significant association with non marital subsequent childbearing.'

Ryan et al. (2006:103)

Instead, Ryan et al. (2006:114) found counter-intuitive effects – the family cap was marginally associated with greater odds of a non marital subsequent birth. The authors tested multiple model specifications to see if the counter-intuitive findings were robust and found that in all models this effect persisted (Ryan et al., 2006:119). The authors report that these findings are similar to other national level studies that found counter-intuitive effects of family cap policies for at least some populations (see discussion of Joyce et al. (2003) and Kearney (2004) previously), but do not match state-level analysis by Horvath-Rose and Peters (2001) (see ahead) that showed a negative association between family cap and non-marital births at the state level (Ryan et al., 2006:119). However, they are consistent with Blank's (2002, cited by Ryan et al., 2006:120) conclusion that most studies find either minimal effects or none at all and with Levin–Epstein's assertion that 'the available research offers no compelling evidence that (family cap policies) have achieved the objective of reducing fertility.' (Levin-Epstein, 2003:1 cited by Ryan et al., 2006:120). Ryan et al. (2006:120) argue that their research suggests:

'... that personal characteristics, not public policies, are stronger determinants of women's childbearing decisions. Unmarried mothers who were younger, or who were black or Hispanic, were more likely to have subsequent births than were older, non-Hispanic white unmarried women.'

In addition, women living with cohabiting partners were more likely than never married non-cohabiting partners to have a second or higher order birth (Ryan et al., 2006:120).

The research also suggests parity toward two children (Ryan *et al.*, 2006:120). Women who had one child were more likely to have had a non-marital subsequent birth than women with two children, whilst those with three or more children were less likely to have another child. This, the authors suggest, reflects US society's two child norm:

'Given the tendency of many women to want at least two children, there may be very little hope that legislation aimed at limiting fertility can have a measurable impact for women with fewer than two children.'

Ryan *et al.* (2006, 120)

The authors conclude that:

'Overall, this paper contributes to an expanding body of research that shows minimal effect of welfare waivers on fertility.'

Ryan *et al.* (2006)

#### Indeed:

'The family cap may instead increase family hardship by limiting the amount of money families receive as their family size grows and ultimately adversely affect the well-being of children. In response some states have reversed their decisions to implement family cap policies and are eliminating them.'

Levin-Epstein (2003, cited by Ryan et al., 2006:121)

However, other studies such as Camasso (2004), Camasso *et al.* (1998; 2003), Hovath-Rose and Peters (2001), Sabia (2006), and Horvath-Rose, Peters and Sabia (2008) all find statistically significant associations between the family cap and fertility; whilst Lopoo and DeLeire (2006) find statistically significant effects of the minor parent provisions on 15 to 17 year olds. These studies are outlined ahead.

As part of AFDC waiver agreement with the Federal Government New Jersey undertook an experimental evaluation of the family cap component of the New Jersey Family Development Program. Implemented in 1992, the Family Development Program was the first to impose a cap on benefits for women who had a child while on welfare. Camasso's (2004) evaluation of New Jersey uses the sequential phasing of specific reform components by county as a natural experiment to isolate the relative effects of a family cap policy and enhanced JOBS programme on the fertility of 2,100 women on welfare who were randomly assigned to treatment and control groups. Data were collected from 1992 to 1997 (Camasso *et al.*, 2004).

#### Box 4.9 Further details – Camasso (2004)

Camasso (2004) uses the experimental design and the phased introduction of components at county level to create a multiple treatment to unbundle the effect of the family cap from that of the enhanced JOBS programme which he hypothesised also affected the fertility of recipients.

However, several concerns have been raised about Camasso's analytical approach. For example, Kearney (2004) questions whether the results of the New Jersey experiments apply to other contexts and suggests that there is evidence that the experimental design was contaminated (Kearney, 2004). Kearney (2004) cites Loury (2000) who claims that many members of the treatment and control groups did not know which policy applied to them; more than one-quarter of case workers in the New Jersey experiment admitted that they used their discretion when allocating participants to the treatment and control groups. In addition, the response rates to the surveys were low and that respondents were not representative of the larger AFDC caseload (Loury, 2000, cited by Kearney, 2004:301).

Camasso (2004) found that the family cap significantly **lowered** births and increased abortions and use of contraception for short-term welfare recipients. However, he found that the family cap did not have any effect on longer term recipients; although the enhanced JOBS programme reduced fertility, independent of the family cap, amongst longer term welfare recipients (Camasso, 2004:462).

Camasso (2004) suggests that his finding of differential impact of the family cap on new and ongoing welfare recipients may be expected from general demographic theory. More recent recipients tended to enter the programmes with more education and fewer children than ongoing recipients, which after controlling for age are indicative of lower fertility (Mellor, 1998, cited by Camasso, 2004).

However, there are doubts about Camasso's methodology. Ryan *et al.* (2006) cite Fein *et al.* (2002), Loury (2000) and Rossi's (2001) concerns about numerous aspects of the study's approach, including the sample size, representativeness and attrition, the process of random assignment, short follow-up periods and that similar services and information were being provided to the treatment and control groups, all of which they suggest cast doubt on the study's conclusions, see Box 4.9.

Three studies by combinations of Horvath-Rose, Peters and Sabia, who use aggregate vital statistics birth data, find that the family cap has an effect on non-marital births. Horvath-Rose and Peters (2000) analyse the effect of the family cap on state level non-marital birth ratios (defined as the proportion of non-marital births to total births that occur to unmarried women) between 1984 and 1996. The study concludes that the family cap decreases non-marital fertility for all race and age groups. They find that the family cap is associated with a nine per cent decline in non-marital birth ratios among teens and a 12 per cent decline among adults in models that control for both unobserved state and year variables. However, the study's methodology has been questioned, see Box 4.10.

## Box 4.10 Further details – Horvath-Rose and Peters (2000)

In their analysis Horvath-Rose and Peters (2000) control for state and year fixed effects, high school completion rate by adults aged 18-24, proportion of state population living in urban areas, proportion of State population that are fundamentalist adherents and indicator variables for waivers for minor parent provision, time limits, work requirements, AFDC-UP, child support, expanded time disregard, asset limit, school attendance and performance requirement, and parental consent requirement for an abortion, and requirement for sex education in schools.

However, Joyce et al. (2004) point out that 20 of the 23 states that implemented the family cap did so in 1995 or later. Thus Horvath-Rose and Peters had few post cap observations in their sample (Joyce et al., 2004:478). Kearney (2004) also questions the study's methodology. She suggests that the analysis confounds the marriage and fertility responses to the family cap, fails to account for the changes in the reporting of marital status in vital statistics data that occurred during the study period and presents implausible findings, specifically that there are such large effects for the entire population of women aged 20-49 most of whom are unlikely to ever receive welfare (Kearney, 2004:301-302). Joyce et al. (2004:478) also point out that some of the other results of the study are counter-intuitive which they suggest 'raises questions about the robustness of the specification.' For example, Joyce et al. (2004:478) find it strange that the minor parent provisions (see discussion ahead) are associated with a large increase in the non-marital birth ratios amongst adults while AFDC-UP programmes are associated with lower non-marital birth ratios amongst teenagers but not adults.

Sabia (2006) uses aggregate data from 1984 to 1998 to estimate the association between family cap policies and non-marital birth, pregnancy and abortion rates. Sabia finds 'robust evidence' even after controlling for a wide set of policy variables, socio-economic characteristics and time-invariant State level unobserved heterogeneity that the family cap is associated with a reduction in non-marital birth rates particularly amongst one sub group - black women (Sabia, 2006:112-113).

Sabia (2006:112-113) also finds that the reduction in non-marital birth rates particularly amongst black women were driven by a reduction in non-marital pregnancies rather than an increase in abortion or marriage rates. Sabia (2006:129) concludes that his finding suggest that 'black unmarried women are responding to the family cap by having less sex or increasing the level of contraceptive care taken during intercourse.' This finding is not consistent with Camasso et al. (1998) and Jagaqnnathan and Camasso (2003). Both analyses suggest that pregnancies and births amongst women who were in receipt of welfare declined after the family cap was introduced and abortions increased (Kearney, 2004:). However, Sabia (2006:128) points out that Camasso et al. (1998) only looked at the family cap in New Jersey, while other methodological concerns about Camasso's findings have been discussed in Box 4.9, previously.

Sabia (2006) finds that the reduction in the non-marital birth rate of black women associated with the family cap is driven by a 2.7 per cent decline in the non-marital pregnancy rate amongst black women '... which is consistent with the hypothesis that that higher direct economic costs and higher indirect stigma costs induce unmarried black women to curb their childbearing by avoiding pregnancy.' (Sabia, 2006:129). The author suggests that there is evidence that stigmatising illegitimacy may effect black women more than white women due to white people holding racial stereotypes judging unmarried black women more harshly, particularly with regard to welfare receipt. He cites studies which show that media coverage of welfare use a disproportionate share of blacks in reports on poverty whilst white people who 'harbour racial resentment' respond less generously to welfare policies targeted at supporting black people. As noted above these findings differ from the studies reviewed by Moffitt (1998) which found that welfare had a greater effect on white women. Baughman and Dickert-Conlin suggest that one of the consistent findings of these studies - that welfare seems to have a larger effect on fertility for white women than black women has 'never been explained well in the literature' (Baughman and Dickert-Conlin, 2007: no page numbers).

Sabia caveats his findings with reference to unobserved state specific time varying characteristics associated with the implementation of the family cap and the reduction in non-marital childbearing that are not caused by the policy. Under these circumstances 'the observed impact may be upwardly biased.' (Sabia, 2006:128). This concern is particularly acute because of the use of aggregate data in the analysis:

'If time-varying, unobserved, state-specific anti-illegitimacy sentiment is correlated with the implementation of the family cap, then the policy impact ... may, in part, reflect spurious correlation. Similarly, if there are unobserved policy changes that are correlated with the implementation of the family cap, then the observed policy impacts may be overstated.'

Sabia (2006:128)

Horvath-Rose, Peters and Sabia (2008) also use aggregate state level panel data from 1984 to 1999 to examine the extent to which state welfare reforms had lowered age and race-specific non-marital fertility. The authors also look at other policy initiatives: time limits; AFDC-UP (welfare eligibility for two parent families); sanctions for non-compliance with child support; expanded income disregards and asset limits; the minor parent provision that requires minor parents to live with their own parents or guardians; and school attendance and performance requirements. The last two are only included in the teen analysis (Horvath-Rose *et al.*, 2008:129).

## Box 4.11 Further details – Horvath-Rose et al. (2008)

The analysis is by age and race. Age is broken down into teen women (15-19) and post-teen women (20-49), whilst race is broken down into two groups: white women (including white Hispanic) and black women (including black Hispanic women) (Horvath-Rose *et al.*, 2008:127).

The authors claim to improve on previous research by correcting the models for bias in the panel data (Horvath-Rose *et al.*, 2008:120-121). The models include controls for unobserved variables.

The authors use two measures of non-marital fertility: the non-marital birth ratio, defined as the proportion of non-marital births to total births that occur to unmarried women, and the non-marital birth rate, defined as the proportion of all women in the population with a non-marital birth (Horvath-Rose *et al.*, 2008:124). However, the authors point to the weakness of using the nonmarital birth ratio as a dependent variable because an increase in marital births with no change in nonmarital births would result in a decrease in the nonmarital birth ratio. It is also important to note that changes in the non-marital birth rate could be caused by marriage as well as birth decisions (Horvath-Rose *et al.*, 2008:123).

The authors also point out the limitations of using aggregate data to estimate the impact of family cap on non-marital births as those data include women who are not in receipt of welfare.

With the exception of the family cap, Horvath-Rose *et al.* (2008:131) do not find that any of the policies examined have significant effects on non-marital childbearing. The authors do find evidence that the family cap is associated with a decline in non-marital birth ratios for both age and race groups. The family cap is associated with a decline in non marital birth ratio of 5.3 per cent for white teens and 1.6 per cent for black teens. For the post teen groups the family cap is associated with a decline of 3.9 per cent for white women and 6.9 per cent for black women. For their second measure of non-marital births – birth rates – the family cap is associated with a decline of non-marital birth rates among black post-teens. The family cap is also associated with a reduction in non-marital birth rates for black teens and white post-teens but these findings are insignificant (Horvath-Rose *et al.*, 2008:130).

The authors suggest that one explanation for the stronger results for non-marital birth ratios than for birth rates could be that the family cap may be positively associated with marital births which if true would suggest that the family cap may encourage unmarried women to marry prior to the birth of the child.

However, the authors suggest that the size of the association between the family cap and marital birth rates – 1.3 per cent for white post-teens to 15.8 per cent for black teens (Horvath-Rose *et al.*, 2008:132) – suggests that this explanation is incomplete and that there are unobserved characteristics associated with both higher non-marital birth rates and the implementation of the family cap (Horvath-Rose *et al.*, 2008:131).

Nonetheless, the magnitude of the effect of the family cap is small, and the authors suggest that:

'... changes in variables such as social stigma, expectations and unmeasured aspects of socioeconomic variables appear to play a larger role than changes in welfare policy.'

Horvath-Rose et al. (2008:134)

#### Social experiments

Maynard *et al.* (1998:153-155) review eight social experiments seeking to reduce the fertility of teenage parents on benefits or at risk of claiming benefits. The projects were: Job Corps, Job Start, New Chance, Project Redirection, Ohio Learnfare, Teenage Parent Welfare Demonstration, Teenage Parent Health Care program and the Elmira Nurse Home Visiting program. The results of the evaluations were mixed and show that the impacts of programmes on abortion rates can mitigate against any effects on pregnancy rates. Only two projects, Teen Parent Health Care and Elmira Nurse Home Visiting, which were both health-focused and targeted on first-time teen parents, significantly reduced the repeat pregnancy rates by 57 per cent and 43 per cent, respectively.<sup>50</sup> Another two, Job Start (an employment and training program) and New Chance (an education and training program), had

Expressed as a per cent of the average for the comparison/control group.

significant *increases* in repeat pregnancy rates of 13 per cent and eight per cent, respectively. Maynard *et al.* (1998:155) suggest that possible explanations for this are:

- The programmes provided opportunities for the women to meet men.
- The programmes promoted the women's self-esteem and ability to live independently but did not alter their contraceptive practices.

For the New Chance programme there was also a significant increase in the abortion rate (34 per cent) which offset the increase in pregnancies so that the overall increase in the birth rate (eight per cent) was statistically insignificant.

For Project Redirection and Teenage Parent Welfare Demonstration projects, falls in the abortion rate amongst participants were sufficient for the birth rates to increase significantly (by 20 per cent and seven per cent, respectively), notwithstanding that the increases in the repeat pregnancy rate were insignificant.

Maynard *et al.* (1998:156) notes that the US literature suggests that AFDC single mothers show little interest in marrying the fathers of their children, or even of pursing child support. The former could occur because the men were '... not reliable sources of economic and emotional support; welfare can at least be relied upon for economic support.' (Maynard *et al.* (1998:156). The latter may reflect the absence of any financial gain from doing so.

## Welfare reform, single mothers and living arrangements

There is also a US literature on welfare reform and single mothers and living arrangements. The regression analysis by Schoeni and Blank (2000), which utilised data on all women aged 16-54 years from the CPS for 1977 to 1999 to investigate the impacts of AFDC waivers 1992 to 1996 and of TANF, and described in Section 2.2.3 shows that the AFDC waivers significantly decreased rates female headship by 1.7 percentage points for those with less than 12 years of schooling (Schoeni and Blank, 2000:17 and 31).<sup>51</sup> The AFDC waivers had no statistically significant impacts on female headship for women with more years of education. The authors claim this finding is what would be expected, that is, women with less education (high school dropouts) were more likely to be affected by welfare programmes.

However, their difference-in-difference model to estimate the impact of TANF on female headship rates (see Section 2.2.3), shows that for women with less education female headship rates fell compared to those with more schooling. For women with less than 12 years schooling compared to women with more than 12 years education female headship rates fell by two percentage points following the introduction of TANF (Schoeni and Blank, 2000:23 and 33). Similarly, or women with less than 12 years schooling compared to women with 12 years education female headship rates fell by 2.1 percentage points. (Both findings are statistically significant at the ten per cent level.)

These findings are statistically significant at the ten per cent level.

There are a few studies that consider the impact of welfare reform on children's living arrangements. AFDC waivers and TANF can be expected to '... decrease the probability that a child lives with an unmarried parent and increase the probability that a child lives with a married parent.' (Bitler et al., 2006:5). However, a possible outcome of welfare reform is that children leave the parental home and live with others if, say, reform leads to a reduction in household income and (financial) stress.

Bitler *et al.* (2006) utilise CPS data for 1989 to 2000 and consider whether the introduction of AFDC waivers in states lead to children to live with an unmarried parent, with a married parent or with neither parent. Separate analyses were conduced for black, Hispanic and white children (aged under 16). The authors find that state waivers were associated with (Bitler *et al.*, 2006:13-20):

- A fall in the likelihood that children will live with an unmarried parent.
  - This effect is concentrated amongst Black (-15 per cent) and Hispanic children (-14 per cent) and is insignificant for white children.
- An increase in the likelihood that they will live with a married parent.
  - This effect is concentrated amongst Hispanic children (20 per cent). The evidence for white children is mixed and the effect is not statistically significant for black children.
- An increase in the likelihood that they live with neither parent.
  - This is especially true for black children (a 13 per cent impact) who are likely to live with a grandparent. There is a smaller impact for white children (seven per cent) and no statistically significant impact for Hispanic children.

The analysis controls for demographic variables (such as child's age), other state public assistance policies (for instance, the generosity of the State's Medicaid programme), labour market variables (such as unemployment and employment rates) and unobserved state and year effects (Bitler *et al.*, 2006:10).

Similar results were found in an earlier study by Acs and Nelson (2004), who used data from the National Survey of America's Families for 1997 and 1999 to explore the impact of welfare reform policies on living arrangements. Although using a different approach (difference-in-difference-in-differences) and hence difference controls and comparisons, they report that:

• States with a tough sanctions policy (that is, they removed the entire TANF benefit for non-compliance with work requirements) were less likely to have low-income single parent families (Acs and Nelson, 2004:281-282). However, such sanction policies did not reduce the proportion of children living with single parents. Nor did they have any clear relationship with two-parent family living arrangements or with the probability that the child lives apart from both parents.

- Family caps tended to be associated with a decline in single parenting amongst families and children (Acs and Nelson, 2004:282-284). Indeed, family caps were associated with an increase in two-parent (married) families and children living in two-parent families. However, they were not associated with a higher probability that a child lived with neither parent.
- Restrictions on the receipt of welfare by two-parent families had no significant association with living arrangements (Acs and Nelson, 2004:285-286). (The restrictions considered were whether the state imposed a work history requirement of a 100-hour test on two-parent families.)

As Acs and Nelson (2004:287) observe their analysis:

'... provide[s] some evidence that welfare policies and practices have contributed to the decline in single parenting and may, in fact, promote dual parenting and marriage.'

However, this analysis only considers one policy at a time, it ignores the 'nuances' of policy and how they interact. As Acs and Nelson (2004:287) point out, the findings for family cap policies, for instance, are likely to be a proxy marker for a set of related state policies and practices that together promote two-parent families and discourage single parent families.

## Minor parent provision and childbearing

Lopoo and DeLeire (2006) examine the minor parent provision that requires teenage mothers younger than 18 to live with a parent or legal guardian and enrol in high school in order to receive welfare benefits. They use state level natality data from the NCHS to compare the trend in fertility rates for young women aged 15 to 17 to the trend for a control group of 18 year olds. Their estimates suggest that the annual percentage decline in fertility rates following implementation of the minor parent provisions was 0.7 percentage points larger for young teens than for teens aged 18, a difference of over 22 per cent. (Lopoo and DeLeire, 2006:275).

## EITC and childbearing and single mothers

US researchers have considered whether EITC affects the fertility decisions of households – as the maximum credit available is higher for families with (two or more) children than those with none.<sup>52</sup>

Since 1991 the maximum credit for families with two or more children has been higher than that for families with one child. A smaller credit to childless recipients was introduced in 1994. These changes to EITC may be seen as creating fertility incentives. (This fertility incentive is available to both married and unmarried women since eligibility for ETIC, like WTC, is not dependent upon marital status.)

There is some evidence that the effect of EITC on actual fertility decisions is small. Baughman and Dickert-Conlin (2003:249-250) found that higher welfare benefits for unmarried mothers were correlated with higher first born birth rates (for children born between 1990 and 1999). They also found that the fertility incentive effect of EITC on first births for non-white recipients was larger than for white recipients (especially for married women).<sup>53</sup> The authors suggest that the greater effect for non white married women may be because EITC was a more important source of income for non-white women. Or if non-white women have fewer marriage options, financial incentives for additional children may matter more (Grogger and Bronars, 2001, cited by Baughman and Dickert-Conlin, 2003:250).

In a second study Baughman and Dickert-Conlin (2007) use birth certificate data for the years 1990 to 1999 and exploit the variation in state EITC programmes to test whether changing incentives in the EITC affect fertility rates – specifically whether expansion in credit influence birth rate among targeted families.

Baughman and Dickert-Conlin (2007: no page numbers) conclude that while economic theory would predict a positive fertility effect of the programme for many eligible women, their results indicate that expanding the credit produced only extremely small reductions in higher order fertility this time among white women.

The impact of EITC on female headship decisions, whilst controlling for AFDC, is considered by Dickert-Conlin and Houser (1999).<sup>54</sup> They find that the impact of EITC on female headship decisions is 'ambiguous'. Their analysis was based on data for women aged 18 to 50 from the Survey of Income and Program Participation for the period 1989 to 1995 and included controls for individual and State characteristics and incorporates individual and year effects.<sup>55</sup> Like Hoynes (1997a), their analysis is suggestive of a link between the degree of control used and the magnitude and significance of the impact of EITC on female headship decisions. When the data are pooled and individual fixed effects are not included AFDC has a positive and EITC a negative effect on female headship, and in both cases the impacts are statistically significant. But when individual fixed effects are included in the models the coefficients become insignificant.

This contrasts with the literature on welfare, where larger effects are found for white families (Moffitt (1998) cited in Baughman and Dickert-Conlin (2003:250)).

Note, in this analysis a female head is an unmarried women with children, that arises from out-of-wedlock births or divorce.

They use the maximum monthly combined AFDC and Food Stamp benefit for a family of three and the maximum combined annual Federal and state EITC for a family with two children as the independent variables of interest in the regression analysis (a linear probability model). The EITC amount is also lagged by one year because it is refunded in the following tax year.

Nonetheless, Dickert-Conlin and Houser (1999:22) maintain that EITC might only have an effect on female headship decisions for women who are likely to be eligible for the credit (as opposed to all unmarried women). As eligibility for EITC is likely to be related to income and the female headship decision they predict eligibility using wage regressions that control for individual and state variable such as age, education and female unemployment rate and assume she works for 2,000 hours. By interacting this variable with the maximum EITC entitlement variable they have a measure of the additional effect of EITC on women who were probably entitled to the credit. Using this measure (and controlling for individual effects) they find that for white women an increase in EITC of \$100 increased the probability of female headship by 0.1 per cent (or 0.02 percentage points), whilst for black women it decreased the likelihood of female headship by 1.4 per cent (or 0.6 percentage points) (Dickert-Conlin and Houser, 1999:22 and 29). Although the latter result is not 'robust', possible reasons for the different results for while and black women are: that they face different labour market opportunities; the earnings of the spouses of black women were sufficiently low that even if the women had a desire to work their likely combined income did not make them ineligible for EITC; and that the impact of EITC might vary by the route into female headship because non-martial childbearing varies by race (Dickert-Conlin and Houser, 1999:29 and 35).

Dickert-Conlin and Houser (1999:23-26) also conduct this analysis for a sub-sample of women with lower levels of educational achievement (that is, with less than high school education) because they are more likely to be in receipt of AFDC. The result for the predicted EITC eligible variable is statistically insignificant for both white and black women. In addition, in the analysis all unmarried women are counted as female heads if they have children and as non-heads if they are childless – thus it does not distinguish between cohabitating and non-cohabitating women. Nonetheless, an analysis for *single* heads gives similar results to when data for all unmarried women are used (Dickert-Conlin and Houser, 1999:32-33).

#### 4.3 Overview

As noted in Section 1.5.5, Malthus was the first to propose the view that there is a systematic interplay between economic considerations and fertility (1798, cited by Clarke and Strauss (1998)). The work of Becker is, once again, influential in the more recent literature which views decisions on single parenthood and childbearing as being affected by the benefits and costs of the options available to women. Within this framework two broad approaches can be identified (Del Bono, 2004:2). The first highlights the role of welfare systems in 'making single motherhood more economically attractive' and is particularly associated with Charles Murray (1984). The second approach focuses on men and women's economic opportunities, including the availability of marriageable men, and male and female unemployment and potential earnings.

Some commentators are concerned that welfare systems encourage non-marital births and single parenthood. Some early US studies would lend support to such a view - all of the significant studies in the Moffitt (1998b) review show that AFDC had a positive effect on single motherhood and fertility. However, overall results are mixed. Often where there is a welfare effect, it is not for the population as a whole, but for a sub-group. For instance, Schoeni and Blank (2000) find small but significant and positive effects on female headship for those with less than 12 years of schooling. Moreover, non-experimental designs that control for unobserved variables at the level of the individual find that the welfare effect is not only small but statistically insignificant.

Evidence of the impact of family caps is weak (Moffitt, 2003:350). Family caps tend to be part of a wider package of reforms and it is difficult to isolate the effect of the family cap from other policies. Moreover, no social experiments have varied the presence of a family cap holding other design features of the project constant.

That people's demographic behaviour is influenced by factors other than financial incentives in the welfare system is illustrated by those US studies that show that the family cap far from reducing non-marital births – as policy makers intended – was associated with increases in births.

Overall, the mixed nature of the findings suggests that there is no consistent and robust evidence that financial incentives in welfare systems affect childbearing and single parenthood decisions.

# 5 Implications for the UK welfare system

#### **Summary**

Identifying the policy implications of the literature reviewed is not straightforward, because there is ambiguity in the underlying theory about the direction of any welfare effect, the 'mixed' nature of the empirical findings and the key role of other factors important in influencing demographic behaviour.

Most of the reviewed research is international, and the application of the findings to the United Kingdom (UK) is limited by differences in policy and national/local context. Impact estimates cannot be simply applied to the UK.

Much of the literature is concerned with the possible effects of marriage penalties within the welfare system. However, a case for marriage penalties can be made. The argument for marriage penalties includes that couples benefit from the presence of household economies of scale, higher incomes, and from the time that a second adult can devote to a family. For society there may also be a trade-off between pursuing neutrality in the welfare system and achieving greater equity. Nonetheless, the literature includes discussions of the policy options for eliminating marriage penalties in the benefit and tax credit systems. The options are:

- Change the unit of assessment from the family/household to the individual.
- Introduce a transferable allowance for couples.
- Target families with children by, for instance, increasing the basic credit in Working Tax Credit (WTC) for couples with children.

Continued

The United States (US)-style family cap policy conflicts with fundamental principles of UK social policy including policies to reduce child poverty. However, options in the literature that may be more applicable to the UK include:

- Short-term:
  - Improve sex education in schools.
  - Improve contraceptive take up.
  - Improve access to early abortion.
- Longer term:
  - Increase employment opportunities for client groups (females and males).
  - Increase opportunities for further education for client groups (females and males).

None of the literature considered the cost-effectiveness of using the welfare system as opposed to other policies to affect family structure. The literature on other family policy instruments, such as family counselling, has not been reviewed, but it may be more cost-effective. Using the welfare system to 'nudge' family structure is also potentially in tension with other policy goals, such as tackling poverty, which are more effectively addressed by the benefit and tax credit systems.

#### 5.1 Introduction

This chapter considers the possible implications of the reviewed research for the UK. In comparison to the US it does not appear that UK policy makers have actively sought to use the tax and transfer systems to influence demographic behaviour; even though it is acknowledged that policy can affect demography (Dixon and Margo, 2006:45). Welfare policies have generally sought to achieve goals other than demographic ones. So to the extent that the UK tax and transfer system has to date affected family structure, any impact would appear to be unintended.

Identifying the implications of the review for the UK is not straightforward, because there is ambiguity in the underlying theory about the direction of any welfare effect, other factors have a significant influence on demographic behaviour and the 'mixed' nature of the empirical findings, see Section 5.2. That there is more non-UK based research than UK research also raises the issue of the extent to which findings can be transferred to the UK especially from the US, see Section 5.3. Nonetheless, implications for the UK for making the welfare system marriage neutral, see Sections 5.4.1 and 5.4.2, and for minimising non-marital births, see Section 5.4.3, are addressed ahead.

## 5.2 Theoretical and empirical ambiguity

Explicitly or implicitly public policies incorporate substantive 'theories' or models about how the factors that policy makers can influence affect the outcomes desired. Some of the policy and public debate about the impact of welfare systems on family structure appears to imply that how benefits and tax credits affect behaviour is conceptually or theoretically unambiguous: that in principle the direction of the effect can be predicted. Thus, welfare systems are seen as leading to less marriage, and more cohabitation, non-marital births and single parents. However, the literature shows that the direction of the effect is, in theory, ambiguous. It may, for instance, lead to more or less marriage, see Section 1.5.

Moreover, the underlying theory provides little guidance as to whether the size of the impact of the welfare system on family structure is small or large. In addition, whether the effect is direct (receipt of benefit causes, say, single motherhood) or there is a selection effect (some women possess characteristics that mean that they are likely to be both benefit recipients and single mothers) can be unclear (see Bradbury, 2006). Moreover, observed impacts may be less than theory would predict because the traditional Beckerian model attempts to explain whether someone will marry rather than when they will marry (Ellwood and Jencks, 2004:18-19). Specifically, the model does not predict the age at which people will marry and hence cannot explain why people are delaying the age of first marriage. Nor does it help to explain non-marital childbearing.

The conceptual model in Figure 1.1 depicts the impact of the welfare system as being moderated and mediated by other factors. How the welfare system is seen as affecting demographic behaviour is important because it influences policy options. For instance, if receipt of Income Support (IS) had a direct effect on childbearing then increasing benefit rates would lead to more births and single mothers. However, if the effect is indirect or other factors lead to both benefit receipt and single motherhood then not increasing benefit rates could undermine the women's well-being as well as having no significant effect on future childbearing.

The review suggests the exact determinants of family structure are unclear, making it difficult for researchers to inform policy makers about how to approach the design of welfare and family policy. The empirical research suggests that, in general, the impact that welfare systems have on family structure is modest, mixed (i.e. it can be both positive and negative) and other factors are influential. Implying that simply changing the welfare system with the intention of nudging family structures in a certain direction is unlikely to have any major impact. So, for instance, Geronimus (1997) argues that the assumptions upon which US policy are based, while 'well-known' are nevertheless either erroneous or presented with greater certainty than the evidence warrants. She suggests that the nature and scope of the welfare reform debate in the US resulting in Personal Responsibility and Work Reconciliation Act 1996 (PRWORA) reflected a view that poverty is the result of poor people's values and behaviours, and that this is justification for policy makers to intervene explicitly in the childbearing decisions of people on

welfare. However, Geronimus argues that the policy 'prism' through which single parenthood is viewed:

'... may ill reflect the traditions or, more importantly, the environmental contingencies and life expectancy faced by members of poor families who hope to provide for children's well-being.'

Geronimus (1997:406)

Geronimus points out that:

'... the health of poor African American women deteriorates in measurable ways as early as the mid-20s, perhaps the consequence of long-term severe socioeconomic disadvantages.'

Geronimus (1997:415)

and that this necessitates early childbearing to give poor African American mothers a greater chance of supporting their children into early adulthood. Thus:

'... welfare policy makers ... fail to distinguish between the function of providing stability, care, and economic support to children that is most often provided by married couples in the United States from the form of marriage itself.'

Geronimus (1997:418)

Geronimus (1997:406) concludes that if 'apparent consensus siphons energy and resources away from searching debate about the nature of poverty, it presents a definite social danger'.

## 5.3 Policy transfer

Much of the literature reviewed in this report is taken from the US, and this raises the question of the extent to which US findings can be transferred to the UK (see Midgley *et al.*, 2008). When assessing the applicability of findings for other countries to the UK it is crucial to locate those findings within their policy and local/national context. Arguably, important policy and other contextual differences mean that the US findings in particular cannot simply be transferred to the UK.

There are critical policy differences (such as, more generous benefits for women in the UK compared to the US) which mean that the context within which people take demographic decisions is very different. There is no UK equivalent of Aid to Families with Dependent Children (AFDC)/ Temporary Assistance for Needy Families (TANF) and Food Stamps, nor of the pro-marriage policy ambitions underpinning the former. Even policies that might appear similar, such as Earned Income Tax Credit (EITC) and WFTC/WTC, contain vital differences. For instance, EITC has 'phase-in' and 'phase-out' stages; so that as recipients' income increases so the amount of credit payable increases (phase-in) until a maximum is reached and it then reduces (phase-out). WFTC and WTC have no 'phase-in' stages; rather they incorporate a minimum working requirement of 16 hours per week. Consequently, neither the magnitude nor the statistical (in)significance of the effects observed in the US should be simply applied to the UK.

A further difficulty is that the US findings are mixed; there is no unambiguous lesson to learn. Indeed, the study by Harknett and Gennetian (2003) implies that caution is required when considering the policy transfer of research findings on the impact of welfare systems on family structure. They found that the direction of the effect of an earnings supplement on demographic behaviour differed between two sites within one country (Canada) (c.f. Section 2.2.4). The scope for the direct transfer of these findings to the UK is, therefore, limited. The context within which these studies take place is critical. Implying that policy makers should primarily draw upon UK research and not simply apply estimates from elsewhere in order to gauge the likely effects of the welfare system on demographic behaviour.

Non-UK studies may, however, help to inform UK policy makers about the likely direction of people's response to financial incentives and provide lessons on how policy-related research on this topic should be conducted, see Section 6.2. In many, but not all cases, the direction of any welfare effect on family structure is in the expected direction – even if the impact is statistically insignificant. Nonetheless, a general lesson from the US research is that actual behavioural responses to welfare system financial incentives are smaller than might be predicted. Demographic decisions are influenced by a range of complex interacting factors and people do not appear to simply react to financial incentives.

## 5.4 Welfare policy and the family

Much of the UK policy literature on welfare systems and the family is focused on the existence and impact of marriage penalties. There is also some concern about non-marital births. Accordingly, this chapter now focuses on marriage disincentives in, and policy options for, the welfare system (see Sections 5.4.1 and 5.4.2) and then policy options for reducing (non-marital) births.

## 5.4.1 Is there a case for marriage penalties?

This sub-section is focused on the case for and against marriage penalties and the next sub-section considers the options for eliminating penalties. It is recognised that there are broader moral and political issues beyond the scope of this review, namely, what is the purpose of marriage in today's society and whether government should promote marriage or other family forms. Moreover, this review has not explored the evidence on, for instance, the social benefits and costs of marriage as opposed to cohabitation.

Even in the US, where in recent years the state has adopted pro-two-parent family policies, few states have a made a concerted effort to influence family structure through the welfare system. In part this may reflect a reluctance to use the system to influence what for many may be seen as a private matter, and if pursued may further stigmatise single parenthood (Fein *et al.*, 2002:1). It may also be (tacit) recognition that other policy instruments might be more cost-effective at steering demographic behaviour than the benefit and tax credit systems. None of the reviewed studies included a cost-benefit analysis making it impossible to know the

social costs and benefits of attempting to influence family structures through the welfare system. This review has also not looked at the literature on the impacts or cost-effectiveness of some other possible polices, such as family counselling.

One criticism of the welfare system is that it is not marriage neutral – indeed, that it imposes a penalty on marriage. The broad case for abolishing marriage penalties is that marriage is a 'merit good' and so should be promoted – that is, people may not be aware of the benefits from marriage to themselves and their children. There may also be benefits that accrue to society from marriage, which might otherwise incur extra costs, say, if cohabitation rates were higher. In addition, the existence of penalties may encourage fraud in the benefit and tax system as some couples may seek to avoid financial penalties (Brewer, 2007:225; Morgan, 2007:74-79). The removal of penalties would eliminate this potential fraud.

Even evidence that the welfare system may, at most, only have modest impacts on marriage rates has not deterred some from calling for the elimination of marriage penalties. Policy can have considerable symbolic significance. Some argue for the abolition of marriage penalties because they 'sent the wrong messages' about the importance of marriage (Fein et al., 2002:1). For instance, Morgan (2007:124) states:

'By rewarding some behaviours and penalising others, tax and welfare systems affect the preference and behaviour of individuals not just through hard cash calculations but by (unavoidably) embodying and promoting certain values and assumptions.'

Invariably those making this point are pro-marriage. However, such symbolic policies risk (re-)stigmatising other family forms. They undermine the acceptance of, and being neutral towards, individual choices about family structure.

In addition, there may be a rationale for not requiring the welfare system to be marriage neutral. One rationale for marriage penalties in the welfare system is the presence of economies of scale within households – fewer resources are required to maintain a standard of living for a second adult in a household than for the first (Eissa and Hoynes, 2000b:4). Married couples benefit from sharing resources – for example, the cost of accommodation for two people living together is less than twice the cost of accommodation for a single person (Brewer, 2007:218). That this occurs is acknowledged more widely in social security through the use of equivalisation scales. However, any such economies of scale accrue from any group of people living together and imply that these other living arrangements, such as cohabitation, should be treated like marriage. That is, the economies of scale argument cannot just be used to justify penalties solely for married couples.

It could also be argued that it is more difficult for a single mother to gain paid work than for a partner in a couple. The former is likely to want part-time work, which in turn is likely to be lower paid and the marriage penalty could be viewed as compensating for this. That is, couples tend to be better-off than single mothers so they are better able to cope with lower tax credits and benefits. There is, then, an element of equity in having marriage penalties in the tax and transfer system.

Moreover, a two adult household in comparison to a single person household benefits from the time that the second person can devote to the household and this has a monetary value. For instance, one partner in a two-parent family can undertake childcare, and whilst this is unpaid work, it has an implied monetary value in that unlike the single parent situation childcare need not be purchased. Higher benefits for lone parents therefore to some extent compensate for the need to buy in services that a two-parent family can obtain through specialisation.

In addition, eradicating marriage penalties might be expensive and not cost-effective; especially as they do not appear to have a major impact on behaviour. Whether the removal of marriage penalties would be cost-effective is unknown as there appears to be no published research in this area. However, the conceptual model outlined in Chapter 1 (see Figure 1.1) posited that the welfare system could affect family structure through employment and family income. Policies that generate sustainable employment for (unskilled) men possibly have more of an effect on marriage decisions (Burstein, 2007:419) than would the removal of marriage penalties. Moreover, improving women's earnings and educational attainment/employment prospects may increase the stability of marriages (Burstein, 2007:419).

As mentioned in Section 2.1.1, achieving marriage neutrality is problematic if Government's want to target financial support on low-income couples with children and the unit of assessment for benefits and tax credits is the household rather than the individual (see also Brewer, 2007:217-220). Moffitt *et al.* (1998:272) highlight that where policy treats married and unmarried families or children in non-cohabiting and cohabiting families differently there is a trade-off between neutrality and equity. Neutrality requires that, say, people's decisions about cohabiting should not be encouraged or discouraged by public policy. However, equity considerations might lead governments to compensate children in single parent families more than those in cohabiting couples, because the latter may have additional income or lower household expenses per person (see discussion previously). But if benefit payments are lower for children in cohabiting families compared to non-cohabiting families this undermines the neutrality principle.

## 5.4.2 Reducing marriage penalties

The literature does, however, contain some proposals for reducing marriage penalties that can be used individually or in combination:

• Change the unit of assessment from the 'family' to the individual.

Marriage penalties arise in part from having the 'family' (or benefit unit) as the unit of assessment in the welfare system.<sup>56</sup> Penalties could be tackled by introducing individual (rather than household) based benefits and tax credits (Carasso and Steuerle, 2005:170-1). There would then be no financial incentive for single motherhood – the single mother and the mother in a couple would

Individualising benefits is also a policy option touched on by David Freud in his review of welfare to work (Freud, 2007:103-104).

receive the same amount of benefit/tax credit. This would imply, for example, changing the unit of assessment for WTC from the household to the individual. Thus, adults in two earner households would have separate claims for WTC. A variant of this is that total family income is allocated for tax purposes to dependants, and each 'slice' would be taxed as the income of that individual (Morgan, 2007:149).

However, this would undermine vertical equity considerations and the targeting of support on those with a low-income, as two mothers with the same nominal income might receive the same amount of credit but one lives with a partner on a high income and the other does not (see Adam *et al.*, 2002:15-16; Brewer, 2007:218; Moffitt, 2003:333). Whereas the UK has:

'... a tax and benefit system that redistributes income to couples who have a low combined income, rather than to individuals who have a low individual income.'

Brewer (2007:218)

Individualisation might also reduce tax revenues, and increase administrative and compliance costs (Alm *et al.*, 1999:203). The change in the assessment unit might also make the system even more complex to users.

In addition, individualising tax and benefits may be inappropriate if adults in the same household share their resources to some extent and benefit from economies of scale. The choice of unit assessment also has important implications for the financial autonomy of individuals in partnerships. Whether the unit for assessment for taxes and benefits for couples should be their joint or individual incomes will partly depend upon views about whether families do, or ought to, share resources.

#### Transferable allowance.

Tax penalties could be tackled by introducing a transferable tax allowance for couples (Beighton and Draper, 2007:14-18); a variant of this would be to restrict the measure to only married couples rather than those cohabiting (Brewer, 2007:217). Individuals in couples would be allowed to transfer the unused part of their personal allowance to their partner (Brewer, 2007:228-230). This proposal would benefit one-earner families with children; indeed, it would reduce the incentive for one-earner couples to become two-earner households. To contain costs to the Treasury, the ability to transfer allowances might be limited to families with children aged under a certain age (say, six) and/ or the transferred allowance to the basic rate of taxation.

#### • Target families with children.

Specific UK proposals include enhancing the tax credits for couple families to reduce the 'couple penalty' and/or introducing a tax allowance for married couples (Beighton and Draper, 2007:14-18).

One tax credit option is to increase WTC for couples with children, which could be achieved by increasing the basic credit for couples with children (Brewer, 2007:226-227) or to create a second credit for an adult within CTC or to introduce an income disregard for couples living together (Draper, 2005). Single parents would be unaffected by this proposal. It removes the penalty single parents face when they cohabit and would assist in combating child poverty. It would help one-earner families, but might discourage them from increasing their income by working more hours or moving to a job with a higher hourly rate.

As the proposal increases the credit paid to couples with children compared to couples without children, it might affect the childbearing decisions of the latter (but see Chapter 4).

An alternative is to increase WTC for two-earner couples only, for instance, by introducing an extra credit for two-earner families (Brewer, 2007:227-228). It would have similar impacts on the couple penalty as the previous proposal, but would also reduce the incentives for some couples not to increase their income.

A more radical option is to establish a universal benefit or tax credit system for families with children, so that they receive support regardless of marital status (Carasso and Steuerle, 2005:170-1). This could be seen as a variation on a 'Basic Income' scheme, although some argue that the cost of this option would be too high (Morgan, 2007:148-149).

It is possible that there is a feedback mechanism with family structure affecting labour supply decisions. Changes to the tax and transfer systems to make them more marriage/couple 'friendly' might affect work incentives (Brewer, 2007:219). In particular, the work incentives for the first earner in a couple can differ from those for a second earner (see also Buffeteau and Échevin, 2008:2-3). Thus, changes to marriage penalties to make the welfare system marriage neutral might have adverse effects on work incentives.

If union/marriage penalties are retained in the UK welfare system, then arguably the Government's should communicate more clearly the reasons for the differences in benefits for married and single households (Burstein, 2007:419).

## 5.4.3 Policy options for reducing (non-marital) childbearing

The US-style family cap policy conflicts with fundamental principles of UK social policy including policies to reduce child poverty. However, there are options in the literature that may be more applicable to the UK. These include:

#### Short-term

- Improve sex education in schools.
- Improve contraceptive take up.
- Improve access to early abortion.

#### Longer term

- Increase employment opportunities for client groups (females and males).
- Increase opportunities for further education for client groups (females and males).

However, Geronimus (1997:406) questions the merit of placing the reduction of teenage childbearing as an important goal of welfare reform:

'... welfare reformers focus narrowly on teenage childbearing and leave unexamined other important causes of persistent poverty. There is no doubt that teenage childbearing is associated with many social and public health problems. And high rates of teenage childbearing in socioeconomically disadvantaged communities likely result from the severe limits placed on the options available to the young for pursuing important goals. But none of these observations leads necessarily to the conclusion that reducing teenage childbearing itself, in the absence of other social changes, is either an attainable goal or one that would result in other social improvements. In targeting teen childbearing as a major activity for welfare reform, the basic question of cause and effect is consistently glossed over by policy makers and advocates.'

Geronimus (1997:409)

The question Geronimus argues policy makers need to address is:

'Would social problems be alleviated if the same women who become teen mothers postponed childbearing to older ages?'

Geronimus (1997:409)

The social force that determines labour market successes, according to Geronimus (1997), is inequality. In predominantly African American poor urban populations, which Geronimus (1997) suggests are some of the very populations where early childbearing gives rise to 'most concern in the general population' – the probability of premature death appears to be so high that teenagers who live in these areas cannot be confident that they will survive in to middle age. Geronimus (1997) cites figures showing that, for example, more than one-third of Harlem or Chicago south side women die by the age of 65 while less than three-quarters of Harlem or Chicago south side men survive to age 45 compared to 95 per cent of white males. In Harlem and Chicago south side, 15 year old men have less chance of surviving to the age of 45 than a typical white 15 year old American male has of living to be age 65. Geronimus (1997:423) suggests that premature death from chronic diseases, rather than the more widely publicised incidences of gang violence, are the primary reasons.

Geronimus (1997:426) suggests that if responsible parenthood includes maximising the chance that a parent will survive to see and help her child grow up, then insecurity about one's own longevity would be a serious consideration when deciding whether to defer parenthood.

The perspective that teenage childbearing represents trade-offs made in order to maximise children's well-being in hard circumstances may also explain why early childbearing has continued in extremely disadvantaged communities, despite increased access to contraception and abortion and in the face of public disapproval. It would also suggest that it may continue to exist in the face of welfare reform (Geronimus, 1997:426).

## 5.4.4 Welfare systems and modern living arrangements

This review touches on complex issues that transcend the public and private realms. In policy terms, a welfare system that was not neutral towards different living arrangements would have to trade-off this stance against vertical and/or horizontal equity. The benefit and tax credit system is probably more effectively and efficiently directed at dealing with issues of equity than it is at steering demographic choices. This is not to deny that the welfare system can have a modest effect on family structure, but that demographic behaviour is possibly better guided by other policy instruments. So whilst there is some evidence that many single parents aspire to marriage, the introduction of marriage friendly welfare policies may be less relevant – even less cost-effective - than policies that seek to help welfare recipients fulfil their pro-marriage desires.

More pragmatically, nowadays there is a wide range of family types (see Williams, 2005). If the wider context for social policy is a long-term decline in the rate of marriage, then (irrespective of whether policy makers believe that the trend can be influenced by the tax and benefit system) the case for using martial status in determining tax liability and benefit receipt is questionable. Arguably, the tax and benefit system needs to accommodate changes in family structure, with – if desired – other policy instruments steering demographic behaviour. Welfare systems are probably more suited to addressing other policy goals, such as tackling poverty and increasing the employment rate.

This does not mean that demographic behaviour should be ignored by welfare system policy makers. On the contrary, Carasso and Steuerle (2005:159) suggest that because policies are implemented piecemeal, there is '... little coordination or thought to how they affect married couples.' The implication is that the policy making process should be more holistic in considering the impacts of initiatives on family structure (see also Dixon and Margo, 2006:42-43).

## 5.5 Overview

The implications of the review are, of necessity, limited because how welfare systems affect demographic behaviour is not fully understood. In attempting to gauge the impact of welfare systems on family structure international evidence is of limited value because of key policy and local/national differences with the UK.

Nonetheless, the literature does include policy options for removing marriage penalties and reducing births. However, the case for using the welfare system to influence demographic decisions is unproven as it may not be cost-effective, as well as being at the expense of other policy goals, such as tackling work disincentives in order to combat poverty and social exclusion.

# 6 Conclusions

#### Summary

The impact of the welfare system on family structure is contested – and the empirical evidence provides mixed findings. Identifying any impact is also complicated by the analytical methods used.

There are implications arising from the review for future United Kingdom (UK) research and how any further research is carried out. There is a gap in the UK evidence base. There is a need for research that is more up-to-date in terms of policies covered, that has the potential to provide estimates of longer-term impacts and allows extensive sub-group analyses.

Researching the effects of the welfare system on family structure is difficult because establishing the causal mechanisms is problematic. Studies are complicated by the influence of other (unobserved) factors and the need for extensive sub-group analyses due to differential impacts of welfare systems on family structure.

The review finds mixed evidence that financial incentives in welfare systems affect family structure. Whilst there are studies finding significant impacts, these tend to be small and are countered by studies finding no relationship or the opposite effect. Overall, there is no consistent and robust evidence to support claims that the welfare system has a significant impact upon family structure.

## 6.1 Introduction

There are public and policy concerns about the impact of welfare systems on family structure. Even though welfare systems provide support to elderly and disabled people, the focus of this concern and of the research reviewed here is on the demographic behaviour of the non-disabled, working age population with a low-income (Hoynes, 1997b:101). There is in particular a sizeable body of United States (US) literature exploring the impact of Aid to Families with Dependent Children (AFDC) on family structure, notably on single motherhood arising from non-marital births.

This review shows that the impact of welfare systems on family structure is contested. Moffitt (1998b:67-68) conducted a major review of the US literature and shows that during the 1970s the research suggests that the welfare system had no effect on marriage and childbearing. However, during the 1980s and 1990s there was a consensus that it did have an effect (a negative effect on marriage and a positive effect on fertility); albeit the magnitude of that effect was uncertain with some claiming it was small and others that it was sizeable (Moffitt, 1998a:1 and 1998b:50). Moffitt's review also shows differences by race. A further theme emerging in his review is that in part these results seem to reflect the analytical method used, and the more the analyst controlled for unobserved heterogeneity the more likely the welfare effect was to be weaker, even insignificant (Moffitt, 1998b:71). Moffitt's 1998 review of non-experimental studies (see Appendix C) concluded that:

'A neutral reading of the evidence still leads to the conclusion that welfare has incentive effects on marriage and fertility, but uncertainty introduced by the disparities in the research findings weakens the strength of that conclusion.'

Moffitt (1998b:75)

However, in a later review Moffitt modifies his position slightly - notwithstanding uncertainty in the literature and a number of studies reporting insignificant results, he concludes welfare 'is likely to have some effect on family structure'. (Moffitt, 2003:336). However, studies that incorporate controls for unobserved individual-level variables (for instance, Hoynes (1997a) and González (2007)) find no welfare effect and together with other studies, such as the meta-analysis by Gennetian and Knox (2003), suggest that even Moffitt's tentative conclusion overstates the impact.<sup>57</sup> On balance, this review concludes that there is no strong evidence of a persistent and large impact by welfare systems on family structure; albeit individual studies can be identified where a significant and large effect is found.

Although some contend that welfare systems impact upon family structure, across countries there is no consistent empirical evidence that demographic behaviour is influenced by benefit and tax credit systems. When significant effects are found they tend to be relatively small or modest. Bitler *et al.* (2004:214) commenting on the US literature observe:

'Overall, the estimated effects of welfare are relatively small in magnitude and cannot explain the secular decline in U.S. marriage rates and rise in divorce rates since the 1960s, during which average real welfare benefits declined ...'

This comment by Bitler et al. (2004) appears to be anomalous with the traditional economic orthodoxy which would imply larger effects. This might be for methodological reasons, as the review shows it is a difficult area to analyse; for theoretical reasons, there are ambiguities with the theory (see Sections 1.5 and

An earlier version of the Hoynes' (1997a) paper is included in the Moffitt review (1988b).

5.2); for policy reasons, maybe the incentives are not large enough to have a significant effect; or for behavioural reasons, financial incentives may have little influence on individuals' demographic decisions.

It could be argued that the modest impacts on demographic behaviour reported in earlier chapters are because the incentives within the welfare system were not large enough. After all, in practise, the policies' financial incentives have been primarily designed to enhance recipients' employment prospects and earnings, not to affect family structures. However, it is difficult to identify the size of the incentive that would be required to affect demographic behaviour given the ambiguous nature and small size of the effects reported.

Moreover, the studies ignore the quality of family life. There is no attempt to capture the quality of a familial relationship (marriage or cohabitation) or a child's early years. Even if financial incentives were large enough to influence demographic behaviour there is no certainty that, for society as a whole, the quality of family life would be improved.

Much of the literature assumes that financial incentives affect people's choices. So, for instance, the theory underlying the family cap is that financial incentives and penalties are important determinants of behaviour and can be included in welfare arrangements to influence and control (poor) women's decisions to have children (Kearney, 2004:296). However, people may not respond in the way in which theories predict; people may not engage in rational behaviour in order to maximise their utility (Lewis, 2007:8). Evidence on the impact of work incentives on employment decisions can be used to support the case that the welfare system impacts upon family structure. However, decisions on demographic behaviour are arguably more complex and far-reaching than those about whether to increase hours of work or to accept a job offer. When asked, people do not necessarily identify financial matters as influencing decisions on family structure. For instance, the most common (unprompted) reason cited by clients of the Child Support Agency (CSA) for their relationship breakdown was that their partner had been seeing someone else (Wikeley et al., 2001:38); and none claimed it would make them better-off financially (Henshaw, 2006:21).

The review findings also need to be considered against a background in which families are more fluid and dynamic. As non-traditional family forms become more common they are less stigmatised, and in such a milieu people may be less responsive in their demographic behaviour to financial incentives.

## 6.2 Research implications

The review has a number of implications for future research on welfare systems and demographic behaviour. These are discussed in this section.

#### 6.2.1 Further research

The review shows that there is a need for further research on welfare systems and family structures. There are studies of the impact of the UK welfare system on family structure, but compared to the US the number of analyses is small. The varied nature of the US evidence suggests that more UK-based analyses are required to test the reliability of existing findings. Moreover, the UK evidence does not relate to the current welfare system; there is little on the impact of Working Tax Credit (WTC). There remains a lack of evidence on the effect of recent welfare-to-work programmes on family formation and childbearing (Mauldon *et al.*, 2002). Thus there is a need for further up-to-date UK studies that, over time, could provide evidence on longer-term impacts of the tax and transfer systems on family structure. Moreover, sample sizes for UK studies need to be sufficiently large to allow for detailed sub-group analyses as impacts are known to vary by sub-group (see ahead).

The reviewed (US) studies have tended to focus on the generosity of the welfare system – the level of benefits, rather than other aspects of the system, such as disregards and other support services (Moffitt, 1998b:57). However, these features might be of interest to policy makers because they might affect people's behaviour. Further work on these 'components' of the welfare system, and not just benefit amounts, might be useful.

## 6.2.2 Undertaking research on welfare systems and family structure

The studies reviewed also demonstrate that the analysis of the effect of tax and transfer systems on family structure is problematic. This is because (Carasso and Steuerle, 2005:161):

- It is difficult to establish empirically a causal link between financial incentives in the welfare system and demographic behaviour. Critically, even if, say, a couple calculate a possible marriage penalty (or bonus) before they marry this offers only limited evidence that it actually affected their decision to marry (or not). Conversely, even if people do not rationally calculate the financial implications of marrying they may still effectively be responding to incentives because they observe that unmarried couples have a higher standard of living (Carasso and Steuerle, 2005:161).
- Variables or factors other than financial incentives affect behaviour.

Specifically, an individual's social and cultural context may be a more influential determinant of behaviour than the welfare system. For example,

'Empirically identifying the effects of policy on demographic outcomes is particularly challenging: The decision to marry or cohabit is influenced by a set of difficult-to-observe factors, including social and community norms and personal tastes and preferences.'

Harknett and Gennetian (2003:451)

In the words of Gennetian and Knox (2003:9):

'Other factors such as community norms, culture or the local social policy context, including the dominant religion of the community and the stigma or acceptability of being a single parent, may also shape the influence of any change in women's economic circumstances on marital behaviour. If these important influences vary systematically for different identifiable subgroups of families, they may cause different effects on marriage for different identifiable subgroups ... However, because we do not fully understand how local context, economic opportunity structures, cultural norms and other factors affect marital decisions, these influences may lead to variations in impacts across families, sites, or studies that are difficult to explain.'

Inconsistent findings from social experiments in the US suggest that context is important – variations in findings can be attributed to the influence of political, social, cultural and economic factors operating at each site. The implication is that benefits and tax credits may have less of an impact on family structures than basic economic theory would suggest. Or, if individuals do respond to economic incentives the effect is influenced by local 'contextual' factors.

- There are complex interactions between a number of tax and welfare programmes that affect the financial impacts on households. Yet some analyses focus on one aspect rather than the combined effects of the tax and transfer systems. For instance, studies on the effects of union bonuses and penalties on marriage rates tend to ignore the effect of tax incentives on childbearing, yet these may also encourage marriage (see Rosenbaum, 2000).
- Much of the analysis reported here is focused on women's behaviour, but this
  may only provide a part of the explanation (Ellwood and Jencks, 2004:11).
  Research that explored the responses of both men and women might provide
  useful insights. The US literature in particular has focused on female headship
  rather than other forms of living arrangements that may involve single parents,
  such as cohabitation, 'sub-families', and single mothers doubling up within a
  household (typically a mother and daughter sharing a house) (Winkler, 1993).
- A related point is that not all studies carry out extensive sub-group analyses. Welfare systems might be expected to have differential impacts on demographic behaviour (Moffitt, 1998b:54). Whilst the overall effects may be small or insignificant, several studies do show that impacts vary by sub-group. The study by Moffitt et al. (1998) strongly suggests that where welfare systems treat cohabiting couples differently from married couples (for instance, different eligibility rules for stepfathers and cohabiting males unrelated to the children) then researchers need to distinguish between married and cohabiting couples. Harknett and Gennetian (2003:458) also argue that studies should distinguish between marriage and cohabitation, because the reasons for entry to, and the duration of, cohabitation may differ from that for marriage. Indeed, the effects of the welfare system on single people may be different from those on married couples (Bitler et al., 2004:232). For instance, the impact of welfare systems

on the fertility decisions of single parents are likely to differ from those on couples (Brewer et al., 2007:31). Gennetian and Knox (2003:9) go further and suggest that findings for populations mask important differences between subgroups and variations in impacts by, for example, duration of claim (recent vs. longer-term recipients), age of mother, age and number of children, prior marital status and ethic origin should be explored. For example, it might be expected that, for those already in receipt of benefits, changes to the system could affect second and subsequent fertility decisions whereas this is less likely for those flowing onto benefits. In addition, analyses of childbearing tend to focus on the birth of the first child rather than on subsequent births. Yet a full understanding of the effect of financial incentives on family structure needs to include birth order and parents' decisions about having further children (see Brewer et al., 2007:32). Similarly, welfare systems might have more of an impact on remarriage decisions than on divorce/separation decisions which may be more influenced by other factors.

- A possible reason for the absence of large effects in the reviewed literature could be that there is a considerable lag between a benefit/tax initiative and it affecting demographic behaviour. In general, the studies do not measure longer-term effects; and longer observation periods may be required especially for studies of the impacts of welfare systems on fertility (Fein et al., 2002:6). Gassman-Pines and Yoshikawa (2006:25) who find a significant and large impact (8.9 percentage points) for the New Hope project on the marriage of never-married women at year five argue that, to detect changes in demographic behaviour, longer follow-up periods are required. However, welfare and fiscal policies change over time making it difficult to establish any longer-term impact on family structure (Carasso and Steuerle, 2005:161).
- The US research in particular shows the importance of controlling for economic variables (for example, local wage rates) in the analyses (Harknett and Gennetian, 2003:456).

The review has highlighted that findings can vary by analytical method. As Moffitt (1998b:50-51) observes this is problematic because:

'Whether the differences in study findings are the result of inherent differences in different datasets or differences in the way the data are analyzed for example, in estimating techniques, definitions of variables, characteristics of the individuals examined, other influences controlled for, and so on is difficult to determine because most authors do not systematically attempt to determine why their findings differ from those of other studies.'

Nonetheless, the lesson for the UK is that adequate controls for unobserved variables need to be included in non-experimental and quasi-experimental studies.

In addition, the data requirements for studies are demanding. To investigate if welfare systems affect actual behaviour, longitudinal (panel) data are required. Data with sufficient in-group variation are required to control for unobserved variables, and to allow for changes over time, trend data are also required. Repeat cross-sectional data may be used, but will not allow fixed effects at the individual level to be included in the analysis.

## 6.3 Final thoughts

In conclusion, this review finds mixed evidence that financial incentives in welfare systems substantially affect family structure. Whilst there are studies recording significant impacts, these tend to be small and are countered by studies finding no relationship or a negative effect. Overall, there is no strong evidence to support claims that the welfare system is responsible for the breakdown of the traditional family. This is not to say that individual studies could not be highlighted to demonstrate the contrary position, but that the balance of evidence is that welfare systems have only a small impact on family structure. In particular the magnitude of the impact is not as large as might be predicted by 'traditional' economic theory. This is not to deny the existence of financial incentives, such as marriage bonuses and penalties in the welfare system, but that, on balance, other factors appear to be more important in affecting demographic behaviour. There are two caveats to this conclusion, first, it is based on a reading of selected literature; this was not a systematic review nor did it involve a meta-analysis, and secondly, the observation that other factors may be more crucial is in part based on findings from non-UK countries. The implication of, for instance, making tax and transfer policies more 'pro-marriage' is that any changes would be largely symbolic and the actual effect on behaviour would be small. Any changes would essentially be a 'statement of values' (Burstein, 2007:419).

Policies that appear to have a bigger impact on demographic behaviour than the welfare system include family planning and counselling services. (However, this review has not examined the literature on the effectiveness of these policies.) This does not mean that welfare-to-work programmes have no role to play, as promoting participants' employment and income, especially those of less skilled males, can help provide a secure financial setting that makes marriage and childbearing more feasible for low-income households (see Del Bono, 2004:17).

There are policy trade-offs in the objectives for the tax and benefit systems (Brewer, 2007; O'Donoghue and Sutherland, 1999). Being 'neutral' towards the family implies trade-offs with, for example, maintaining progressivity, treating families with similar total incomes the same, introducing complexity into the tax and benefit systems, and so on. Changing the benefit and tax system to, say, remove marriage penalties, could also have implications for work incentives for financially dependent partners and for targeting financial support to low-income families with children (see Brewer, 2007 and O'Donoghue and Sutherland, 1999). Thus, using welfare systems to promote specific demographic behaviours may involve trade-offs with other policy objectives.

## Conclusions

134

Another key conclusion, however, is the paucity of 'hard evidence' for the UK on how financial incentives affect demographic norms, attitudes and behaviour. As the impact of the welfare system on family structure is likely to remain a topic of policy and public debate, welfare-related research and evaluations should, as a matter of routine, incorporate detailed demographic variables for analysis.

## Appendix A Non-experimental methods of analysis

The main non-experimental methodologies used to identify the impact of welfare systems on demographic behaviour are as follows (Hoynes, 1997a:92-94; Moffitt, 1998b:57-59; Moffitt, 2003:336):

• Regression analyses are used to exploit cross-sectional data on differences in benefits between states to measure the welfare effect (see Moffitt, 1994 and 1998a; Schultz, 1994). These analyses model the outcome variable (usually marriage or fertility) as a function of individual characteristics, the level of State welfare benefits and, less often, other state characteristics (Hoynes, 1997a:92; Moffitt, 1998b:58). However, the estimated impact of the welfare system will be biased if there are any omitted state factors, such as social norms or religious characteristics, that are correlated with both family structure decisions or (through voter preferences) state welfare policies (Ellwood and Bane, 1985, cited in Hoynes, 1997a:91; Hoynes, 1997b:129; Kearney 2004:299; Moffitt, 1994; Ribar, 2003:1). Indeed, social norms and cultural factors are likely to affect demographic behaviour and possibly welfare policy. Hoynes (1997a:93) notes:

'... if the population in a given state believes strongly in the two-parent family, the state may not have much support for an Aid to Families with Dependent Children (AFDC) progam and, hence, it may offer low benefits. .... In general, if the unmeasured effects are positively (negatively) correlated with welfare benefits then the estimated welfare effect will overestimate (underestimate) the true effect.'

Yet these socio-cultural factors are often unrecorded in analysts' datasets, and so cannot be simply added to the statistical models (see also Ribar, 2003:12-14). The inclusion of indirect measures might reduce bias, but some bias will remain.

A possible further weakness of this approach is that the analyses do not provide information on the timing of demographic transition (Morgan, 2007:96).

On the other hand, if demographic behaviour takes a long time to respond to the welfare system it is possible that this 'comparison-of-levels' approach (unlike the approached to be discussed ahead) measures this longer term response (Moffitt, 1998b:59).

- Panel data or repeated cross-sectional data can be pooled to model variations in benefits over time using regression analysis (Moffitt, 1994 and 1998b:58). These are often called 'fixed effects' models and they capture unobserved factors common to all residents of a country, region or state, such as state divorce laws or support services and characteristics of potential spouses. The welfare effect in a state fixed effect model is captured by within state variation (or changes) in benefits over time, which is compared with changes in, say, marriage and fertility. The advantage of this approach is that the analyst does not have to specify exactly which variables are missing, only the structure of the omitted variable (Ribar, 2003:19). Moffitt (1998b:58) provides an example:
  - '... the low AFDC benefit levels and high marriage rates in most southern states may not be a reflection of a true welfare effect but may instead reflect the fact that the South is socially a relatively conservative region where social and cultural norms encourage marriage, as well as being a relatively conservative region politically where elected representatives do not legislate generous welfare benefits. ... [A] positive correlation between benefit levels and marriage (for example) would arise because there is a third variable social, cultural, and political norms that leads to them both, not because benefits affect marriage.'

It is important that the omitted variables are time-invariant and that within the data there are variations over time in the observed indicators. The method does not deal with omitted variables that are time varying – thus it is assumed that the omitted variables, for instance, social norms have not changed over time. So state fixed effects control for unobserved variables that vary across states (but not over time).

If demographic behaviours do not respond fairly quickly to changes in benefits and tax credits then datasets covering a relatively long period of time are required to capture changes. If such data are not available then this approach may only capture short-term responses to changes in benefits and tax credits (Moffitt, 1998b:59). However, if data are available for long periods of time, it might not be justifiable to assume that state effects are fixed throughout the observation period.

Compared to cross-section designs the inclusion of fixed effects, which provide increased control for unobserved variable, can reduce the significance and magnitude, even the direction, of any impact of welfare systems on family structure (see, for instance, Moffitt, 1994).

Models with state fixed effects also tend to include year fixed effects. These effects control for average changes in, say, marriage or divorce within a given year that are common to all areas (say, states).

- There may also be individual effects that are unobserved and correlated with welfare policy, especially if the data have to cover a relatively long period of time. In a pooled cross-sectional model these individual effects may be correlated with the generosity of the state's welfare system and arise through inter-state migration and changes in the composition of the sample. The use of panel data enables the identification of welfare, state and individual effects (Hoynes, 1997a:91). Analysing individual effects requires a panel dataset in order to control for changes in the composition of the sample over time. To the extent that studies do not control for individual effects, but only state effects then they may be biased through inter-state migration and sample attrition and entry. The individual effects capture time-invariant unobserved variables at the level of the individual.
- Within state studies using a determinant of benefits that does not correlate
  with family structure. These studies require recipients to be offered different
  levels of benefit or involve comparisons between those eligible and not eligible
  for welfare benefits. The problem in practice is that differences in benefit levels
  are usually associated with a demographic characteristic so finding suitable
  explanatory variables is difficult.
- There have been a small number of time-series studies that attempt to explore the contribution of the welfare system to observed falling rates of marriage and increasing rates of nonmarital childbearing (Moffitt, 1998b:55-56). A number of US researchers (for instance, Hoynes, 1997b:125-126) have observed that if financial incentives affect family formation then the observed fall in the real value of welfare benefits (especially AFDC since the mid-1960s) should be associated with declines in female headship and non-marital births, but instead they have continued to increase. Suggesting that factors other than financial incentives are operating to influence demographic behaviour. However, controlling for all the possible factors that change over time is very difficult. Even though the fixed effects analyses mentioned previously require longitudinal (at least, repeat cross-section) data, analysts have tended not to conduct time-series analyses. Indeed, they typically include dummy variables for years in regressions in order to eliminate the influence of time trends (that is, to control for year effects).

In summary, the main non-experimental methods used (primarily in the US) seek to investigate whether there is a welfare effect on family structure by using variations in benefits between individuals within states, between states and between states over time (Hoynes, 1997b:124; Moffitt, 1998b:57). However, the source of variation and the controls for unobserved heterogeneity that are used affect the reported significance and magnitude of the welfare effect on demographical behaviour. In addition, using variation arising from state and over time differences only provide information on the marginal effects of the welfare system on family formation. They do provide estimates of how the very existence

of tax and transfer programmes affect demographic behaviour. Thus these studies '... are limited in their ability to make predictions about eliminating programs.' (Hoynes, 1997b:124). This is important since policy makers might wish to know the consequences of cancelling a programme. However, such information may be gained from experimental research designs.

However, for studies of childbearing there are further analytical issues. Geronimus (1997) argues that studies employing standard multivariate techniques to control for specified, measurable background characteristics are likely to overestimate the negative consequences of teen childbearing. She suggests more accurate findings can be derived from comparing sisters, which in effect controls for factors such as general access to resources during childhood, cultural environment, neighbourhood and school system but not personal antecedents of early childhood and is likely to produce estimates of effects of teenage childbearing on subsequent economic success that are still upwardly biased but which highlight the extent to which unobserved factors associated with family background account for poor economic fortunes of teen mothers. (Geronimus, 1997:411).

Geronimus (1997) suggests that a better approach than using sisters is to compare outcomes for teenage mothers and teenagers who get pregnant but miscarry which, she suggests, effectively randomly allocates motherhood to teens. Geronimus (1997) argues that for key economic outcomes bearing directly on the question of long-term economic self-sufficiency, findings from studies using these methods challenge the assumptions upon which policy is based. These studies reduce effects estimated by more traditional methods by at least 50 per cent and in some cases find no independent effect of teenage childbearing on important economic outcomes. Geronimus and Korenman (1992, cited by Geronimus, 1997:412) reported no relationship between teen childbearing and high school graduation or subsequent family income in their analysis of sister pairs in the National Longitudinal Survey of Young Women. Using more recent data from the Panel Study of Income Dynamics, Corcoran and Kunz (1997, cited by Geronimus, 1997:412) found teen mothers to be no more likely to be welfare recipients after the age of 25 than their sisters who became mothers at older ages (Geronimus, 1997:412).

# Appendix B UK studies modelling the amount and distribution of union bonuses and penalties

The focus of this review is to explore the actual impact of welfare systems on family structure. Some analysts model the size and distribution of union bonuses and penalties but not whether they subsequently affect behaviour. This appendix outlines two such United Kingdom (UK) studies by Anderberg *et al.* (2008) and Draper (2008). They provide some information on who is likely to be affected as well as estimates of the size of any bonus or penalty. However, their findings should be considered in light of the criticisms made of this approach in Section 1.6.1.

Anderberg *et al.* (2008) use Family Resource Survey data 1995-2004 to simulate partnership or union bonuses and penalties by comparing what couples aged 20-60 would receive in welfare transfers when living together and apart.<sup>58</sup> The welfare payments included in the analysis are Family Credit (FC), Income Support (IS), tax credits, Housing Benefit and Council Tax Benefit (HB/CTB). The tax credits considered are WFTC and WTC/CTC.<sup>59</sup>

In general, Anderberg et al. (2008:3) report that:

• Reforms to IS (the increases to the then child premia in 1999 and their transfer to CTC from 2003) meant that the union penalty in IS increased (from an average of £67.90 per week in 1996 to £78.80 in 2001) and then diminished (to an average of £44 per week in 2004) (Anderberg *et al.*, 2008:6 and 11).

This study appears to be a development of earlier research by Anderberg (2007).

The Children's Tax Credit is included in WFTC.

- Tax credits generate partnership bonuses for some couples and penalties for others.
- The majority of couples were affected by partnership bonuses or penalties. An analysis of three selected years (1996, 2001 and 2004) to represent the Family Credit (FC), WFTC and WTC/CTC regimes respectively shows that in general around 90 per cent of couples faced a partnership penalty (Anderberg et al., 2008:9-11).60 Less than one in ten had a partnership bonus.61 Furthermore, under the WTC/CTC/benefit system more couples faced a partnership penalty compared with the WFTC/benefit system (an increase of six percentage points). This increase was driven by a dramatic increase in the proportion of couples facing a tax credit partnership penalty – an increase from 28 per cent in 2001 to 64 per cent in 2004. (In contrast, the proportion of couples facing a partnership penalty due to Income Support fell slightly from 28 per cent in 2001 to 26 per cent in 2004.) Similarly, under the WTC/CTC regime fewer couples faced an overall tax credit/benefit partnership bonus (a decrease of 5 percentage points). 'Hence the system in 2004 is, on average, less partnership friendly than the system in 2001.' (Anderberg et al., 2008:9). This change is mainly due to policy reform (see ahead).
- Often the amount of the bonus or penalty is large. The analysis shows that across all three selected years 20 per cent of couples had an overall partnership penalty of £60 or more per week (Anderberg et al., 2008:9). Moreover, the replacement of FC with the more generous WFTC meant than the average size of credit/ benefit partnership bonuses grew – from £7.80 per week to £16.90 per week (Anderberg et al., 2008:11). In addition, under the WTC/CTC/benefit system the average partnership penalty was larger than under the WFTC/benefit system -£46.60 per week compared to £38.10 per week – whilst average partnership bonuses were smaller - £13.90 per week under the WTC/CTC/benefit system compared to £16.90 per week for the WFTC/benefit. This reduction in overall partnership bonuses was largely due to the transfer of IS child premia to CTC. For the typical case (man working at least part-time and the woman not in paid work), under WFTC the partnership bonus was the entire amount of the couple's tax credit. However, under WTC/CTC if the couple were to separate, the man (without care of the children) could be eligible for WTC and the woman would receive CTC. Deducting this potential payment from the combined WTC/CTC they would obtain as a couple gives the smaller (net) partnership bonuses.

By year, the proportions for partnership penalties were: 1996, 87 per cent; 2001, 86 per cent; and 2004, 93 per cent (Anderberg *et al.*, 2008:11).

By year the proportions for couples who faced a bonus were: 1996, eight per cent; 2001, ten per cent; and 2004, five per cent (Anderberg *et al.*, 2008:11).

- A sub-group analysis of the WFTC/benefit system reveals that average weekly bonuses and penalties tended to be larger for:
  - couples with children compared to childless couples;
  - women who are economically inactive compared to those working part-time or full-time (although the proportion facing a penalty is lower for those not in employment (71 per cent compared to 93 per cent);
  - couples with females aged 30-50 years as younger and older women were less likely to have dependant children;
  - couples with fewer educational qualifications because they were more likely to be economically inactive (Anderberg et al., 2008:12-15).

In addition, analysis of the 1999 WFTC/IS reforms and of the introduction of WTC/CTC shows that the distribution of bonuses and penalties changed over time, partly due to these reforms and partly due to demographic changes (such as changes in the composition of the population). This analysis entails simulating what would have happened had the policy not changed (the counterfactual). Changes in partnership bonuses and penalties in this counterfactual will reflect compositional changes in the population, whilst actual changes not found in the counterfactual scenario will reflect the policy reforms (Anderberg *et al.*, 2008:15). In terms of the policy reforms (Anderberg *et al.*, 2008:16-17):

- The introduction WFTC and increases in the child premia in IS improved partnership bonuses. The reforms increased both the proportion of couples facing a bonus and the average amount of the bonus (and decreased the proportion facing a penalty, but had no affect on the amount of the average penalty). The increase in the average amount of the bonus was due to the greater generosity of WFTC compared to FC.
- The WTC/CTC reform, however, was less favourable towards partnership. The reform decreased both the proportion of couples facing a bonus and the amount of the average bonus (and increased both the proportion facing a penalty and the size of the penalty). The decrease in the size of the average bonus reflects the smaller bonus some couples received from the couple/lone parent element of the new tax credit.

In both cases, these changes were due to tax credit reform; the IS reforms did not influence the proportion of couples facing a partnership bonus or penalty.

In terms of demographic changes, an increase in the labour force participation rate of women has led to a small increase in the proportion of couples facing partnership penalties and a small reduction in the proportion facing bonuses (Anderberg *et al.*, 2008:17-19). A consequence of increasing female economic activity is that the proportion of couples confronting a penalty due to IS has decreased. However, the concomitant change is that there is an increase in the proportion of couples with a tax credit penalty.

The modelling by Anderberg *et al.* (2008) is based on a number of assumptions, including that the supposed union dissolutions do not affect the supply of labour (that is, individual earnings), that benefit take-up is 100 per cent, and that the woman would remain the primary carer after union dissolution. The analysis excluded any couple with adults who were long-term sick or disabled, retired, unemployed or self-employed. The modelling also excludes any maintenance payments. (Yet Draper (2008) suggests that maintenance payments reduce partnership penalties, so this is a significant omission.) The finding outlined above must be interpreted in light of these assumptions.

A smaller scale and less comprehensive study by Draper (2008) takes 98 families and estimates the difference in disposable incomes after housing costs for couples if they live together and apart. Separate estimates before and after maintenance payments are also made. However, these are stylised families – unlike the Anderberg et al. (2008) analysis they are not actual families (Draper, 2008:8). Moreover, the selected families are not a nationally representative sample of familial types, although Draper (2008:8) claims that they represent 'a fair cross-section of families' on average and below average incomes. However, the nature of the sample used does limit the generalisability and validity of the study. 62 Nonetheless, the estimates can be used to identify types of couples that might be affected by partnership penalties and bonuses. The estimates show that the partnership penalty is lower after housing costs are taken into account and if maintenance payments are made. The disincentives are higher for couples living in the private rented sector, one-earner couples, two-earner couples with high joint incomes but modest incomes individually, and couples on IS. There are few penalties (even bonuses) for couples living in owner occupation with a mortgage and two-earner couples with 'low' incomes.

Notwithstanding that some couples face sizable bonuses and penalties, the Anderberg *et al.* (2008) and Draper (2008) analyses do not tell us if the existence of a penalty (or bonus) is sufficient to alter behaviour; whether couples facing, say, a penalty were more likely to separate than other couples is unknown. Such an analysis would require the use of panel or pooled cross-sectional data.

## Appendix C Summary of Moffitt's (1998b) review of US non-experimental studies

There are a number of reviews of the literature on the impact of welfare systems on family structure and the most often cited is Moffitt (1998b), who reviewed 68 estimates of Aid to Families with Dependent Children (AFDC's) effect on marriage, single motherhood, female headship, fertility and abortion over the period up to 1996. A summary of his review findings is given in Table C.1. Moffitt (1998b:67-68) shows that during the 1970s the research suggests that the welfare system had no effect on marriage and childbearing. However, during the 1980s and 1990s there was a consensus that it did have an effect (a negative effect on marriage and a positive effect on fertility); albeit the magnitude of that effect was uncertain with some claiming is was small and others that it was sizeable (Moffitt, 1998a:1 and 1998b:50):

'A neutral reading of the evidence still leads to the conclusion that welfare has incentive effects on marriage and fertility, but uncertainty introduced by the disparities in the research findings weakens the strength of that conclusion.'

Moffitt (1998b:75)

Subsequently, Moffitt modifies his position slightly - notwithstanding uncertainty in the literature and a number of studies reporting insignificant results, he concludes welfare 'is likely to have some effect on family structure'. (Moffitt, 2003:336).

However, this is based on giving equal weighting to the studies regardless of the extent to which they sought to control for unobserved variables (Moffitt, 1998b:70). Yet there are more estimates for studies that did not control for unobserved heterogeneity than for those that did and, by race, the two approaches suggest different welfare impacts. That is, there is a link between the analytical approach used (or source of benefit variation) and whether a welfare effect is found by race.

For the estimates not including controls for unobserved heterogeneity slightly more studies reported significant effects (18) than insignificant effects (15) (Moffitt, 1998b:68-69). However, the studies imply a larger effect for white women (with nine significant and only two insignificant estimates) than for non-white and black women (with six significant and seven insignificant estimates). In comparison, for the smaller number of studies including controls for unobserved variables (that is, fixed effects) the estimated effect on white women is weaker (four significant and four insignificant estimates). However, the results for non-white and black women are stronger (with five significant and only one insignificant estimates).

So:

'Overall, although there is a very slight excess of significant estimates over insignificant ones across all races, it is quite small. However, the patterns differ by race and source of benefit variation .... The difference in how benefit variation affects family structure between the two races is a result of a different sorting of single mothers by state for the two races, with white single mothers tending to be concentrated in high-benefit states but black single mothers tending to be concentrated in low benefit states.'

Moffitt (2003:336)

Moffitt (1998b:71) highlights that the studies control for different State level variables. Whether labour market, demographic and/or political characteristics are controlled for varies, as does which specific variables are used. For example, labour market variables include rate of unemployment, median or average wages, and percentage employed in specific sectors. His review suggests:

'... the more variables that are controlled for in an analysis, the weaker is the estimated effect of welfare – although there is no logical reason why this need be so ...'

Moffitt (1998b:71)

Nonetheless, in substantive terms, all of the significant studies in the Moffitt (1998b) review show that AFDC had a positive effect on single motherhood/fertility and a negative effect on marriage.

Number of estimates of welfare effects on marriage and fertility in review by Moffitt (1998b) Table C.1

Type⁴		All races <sup>a</sup>			White		Nor	Non-white or Black	_~
	Insignificant	Insignificant Significant <sup>b</sup>	Mixed <sup>c</sup>	Insignificant Significant <sup>b</sup>	Significant <sup>b</sup>	Mixed <sup>c</sup>	Insignificant Significant <sup>b</sup>	Significant <sup>b</sup>	Mixed
All	8	5	<b>—</b>	8	13	5	10	12	9
Cross-state levels	9	ĸ	<del></del>	2	O	4	7	9	M
Cross-state changes <sup>e</sup>	<del>-</del>	2		4	4		<b>~</b>	5	2
Within-state	_			_			_		
Time series				_		_	_	<b>.</b>	<b>—</b>

Notes:

<sup>a</sup> Study did not report separate estimates by race.

<sup>b</sup> Reported a significant negative effect on marriage or a significant positive effect on fertility or both.

c Reported a mix of significant and insignificant estimates.

<sup>d</sup> For a discussion of the different types of model see Appendix A.

<sup>e</sup> Models that include fixed effects to control for unobserved variables.

Source: Table 4.1, Moffitt (1998b:68)

# Appendix D US studies modelling the amount and distribution of union bonuses and penalties

This review considers the impact of the welfare system on demographic behaviour. However, some United States (US) analysts model the size and distribution of union bonuses and penalties but not whether they subsequently affect behaviour. This appendix outlines selected US studies estimating marriage bonuses and penalties for Temporary Assistance for Needy Families (TANF) and Earned Income Tax Credit (EITC).

#### D.1 Temporary Assistance for Needy Families

Acs and Maag (2005) use data for cohabitating couples with children from the 2001 National Survey of America's Families and a tax-transfer simulation model to estimate the union penalties and bonuses arising from TANF, given known tax changes for 2003 and 2008.<sup>63</sup> They hypothetically 'divorce' couples (see Section 1.6.1) to calculate union bonuses and penalties. They find that most low-income cohabiting couple with children (86 per cent) do *not* receive TANF and by definition receive no penalty or bonus due to TANF (Acs and Maag, 2005:6). Nonetheless, TANF interacts with the tax system to alter the potential marriage bonuses and penalties that cohabitating couples with children face. For the 9.6 per cent of low-income cohabiting couples that faced an average tax penalty of \$1,004 in 2003, marriage would result in an average loss of \$1,800 in TANF benefits giving an overall loss of \$2,804. Under the 2008 tax changes the potential loss in TANF benefits was unchanged but increases in the tax penalty increased the combined tax and TANF loss to \$3,311. For the 65.2 per cent of low-income cohabiting couples that faced an average tax bonus of \$2,939 in 2003, the potential loss

in TANF benefits was \$2,096 leaving them with a small overall bonus of \$843. However, planned tax changes for 2008 would increase the tax bonus (the loss of TANF benefits is the same) giving an overall tax-transfer bonus of \$1,294. Thus Acs and Maag (2005:6) conclude that:

'For low-income families, the potential loss of cash welfare benefits under TANF may be a greater financial barrier to marriage than any bonus or penalty resulting from federal taxes.'

However, their analysis only applies to cohabitating couples with children and ignores other family forms, and the TANF modelling makes a number of simplifying assumptions and does not capture fully the variation in states' eligibility rules for TANF. Moreover, Acs and Maag (2005) simulate marriage bonuses and penalties, and as mentioned elsewhere in this report, such studies do not explore whether the identified financial incentives and disincentives affect actual behaviour.

#### D.2 Earned Income Tax Credit

As mentioned in Section 2.1.1, there are marriage bonuses and penalties associated with EITC. It can provide both marriage bonuses and penalties relative to cohabitation or living independently. This is because it has both phase-in and phase-out (or withdrawal) stages. For a working parent EITC increases as income rises (phase-in stage), there is then an income range where the amount of credit does not change (a 'plateau' where the maximum credit is payable), and, finally, as income rises further the amount of credit received falls (phase-out stage) until it diminishes altogether. So if a single parent marries and the couple's joint income take them above this phase-out threshold they face a marriage penalty (Carasso and Steuerle, 2005:163), and they face high marginal rates of taxation (Ellwood, 2000a:194). However, if a low earner marries a non-earner with a child they may receive a marriage bonus if they file jointly, as their combined income entitles them to a larger credit than if they claimed as two unmarried individuals (Holtzblatt and Rebelein, 2000:1108).<sup>64</sup> Similarly, a two-earner couple with children and a modest joint income could be ineligible for EITC if married, but receive it they do not marry. Cohabiting couples can also receive union penalties and bonuses. In general, EITC provides a marriage subsidy to single-earner families and a marriage penalty to two-earner households (Dickert-Conlin and Houser, 2002:27; Eissa and Hoynes, 2000:687). Given the gendered nature of employment this will tend to penalise marriage for women who work and subsidise those women who do not.

In recent years the Federal Government has sought to reduce the marriage penalties in EITC and the wider tax system. The Economic Growth and Tax Relief Act 2001 created a separate married-couple schedule for EITC, which was phased in between 2002 and 2007 (Hoffman and Seidman, 2003:84 and 95). This Act extends the 'plateau' of maximum EITC for married couples so that the phase-out stage commences at a higher income.

In the US married couples can for income tax purposes file either jointly or separately. Most couples file jointly because this reduces their income tax liability (Holtzblatt and Rebelein, 2000:1110).

In practice, US families' overall marriage bonus and penalty is affected by the interaction between the tax and transfer systems. So that, for instance, the penalty arising from the loss of TANF can in certain circumstances be offset by a marriage bonus from EITC.

In theory, there is an incentive to marry during the (generous) phase-in of the tax credit and to dissolve partnerships during the phase-out of EITC because it only counts the income of married partners (Blundell and Walker, 2001:31; Carasso and Steuerle, 2005:168).<sup>65</sup> As Scholz observes:

'In general, positive incentives to marry are provided to low- or zero-earning taxpayers with children; and positive incentives for separation (or negative incentives for marriage) are provided to couples with children when each has modest earned income.'

Scholz (1994:9)

Ellwood (2000a) makes the same point but adds that phasing-out of EITC is accompanied by the loss of Food Stamps and the phasing in of federal taxes (see also Carasso and Steuerle, 2005:168). As a consequence recipients have high effective marginal rates of taxation.

Some writers, such as Acs and Maag (2005), Dickert-Conlin and Houser (1998) and Holtzblatt and Rebelein (2000) use survey or administrative data and micro-simulation models to estimate the size of the dis/incentives in EITC for partnering. These studies tend to 'divorce' the married couples (see Section 1.6.1) when calculating marriage bonuses and penalties. Dickert-Conlin and Houser (1998) utilise hypothetical families to derive their estimates (see also Eissa and Hoynes, 2000:687; Hoffman and Seidman, 2003:84-86), whilst Holtzblatt and Rebelein (2000) focus on different family arrangements. The simulations by Dickert-Conlin and Houser (1998) suggest that marriage tax penalties are relatively large and that those facing the highest marriage bonuses in the tax system (in part due to EITC) also had the largest marriage penalties in the welfare system. Acs and Maag (2005:3-6) (see also Section D.1) find that for cohabitating couples tax changes introduced by President Bush and coming into effect in 2003 and 2008 have made the welfare system more 'marriage-friendly' – the proportion facing penalties if they were to marry has decreased. The proportion of low-income cohabitating families with children receiving a bonus in 2003 was 63 per cent and this is estimated to rise to 75 per cent in 2008. Moreover, the size of the average tax bonus and penalty was expected to increase by 2008. For low-income families the main source of marriage penalties was EITC. Penalties increase in size because cohabitating couples in 2008 became eligible for fewer and lower non-EITC tax credits and so will be less able to offset their tax liabilities.

EITC also provides incentives for cohabitation. EITC is briefly described in Section 1.4.3.

An interesting approach is used by Ellwood (2000b) which allows fertility and earnings to vary after marriage (rather than assuming that they are unchanged). Ellwood (2000b:1089-1090) uses the Panel Study of Income Dynamics for 1983 to 1991 to observe the reported income of individuals before and after marriage. He calculates their income for the last year prior to marriage and for the first year afterwards, and then estimates whether they would have been 'winners or losers' had the provisions for the rapid expansion of EITC in 1996 applied before and after marriage. Thus, he captures any changes in labour market status or childbearing following marriage, and avoids any impact that the expanded 1996 EITC regime could have had on marriage decisions.

Ellwood (2000b) estimates that there would have been more marriages where the amount of the 1996 EITC (had it been in effect) would have been lower (16 per cent, with an average loss of \$1,500) than higher (11 per cent, with an average gain of \$1,400). The majority of marriages (72 per cent) would experience no change in EITC. The main reason for the higher proportion of 'losers' is that there was a relatively high proportion of marriages where before marriage both partners worked and had a child, and in most of these cases the estimated EITC was lower post-marriage. The estimate of the proportion of winners is likely to be an underestimate as many couples were likely to have children after the first year which could increase their EITC. However, his sample, which is of new marriages, may not be representative of the wider 'stock' of marriages.

As mentioned in Section 1.6.1, studies that estimate EITC net benefits are sensitive to the assumptions made, especially about labour market status and living arrangements. For instance, studies that 'divorce' married couples and assume that the children remain with the women are likely to find that fewer families face marriage penalties than those where it is assumed that, where there are two or more children, they will be divided between the 'separated' parents.

The studies considered here show that EITC generates both marriage bonuses and penalties, and in some cases they are relatively large. The estimates do not say if the bonuses and penalties actually influenced behaviour.

### References

Aassve, A., Burgess, S., Propper, C. and Dickson, M. (2006) 'Employment, family union and childbearing decisions in Great Britain', *Journal of Royal Statistical Society A*, 169(4), pp. 781-804.

Adam, S., Brewer, M. and Reed, H. (2002) *The Benefits of Parenting: Government Financial Support for Families with Children Since 1975, IFS Commentary 91*, London: IFS.

Acs, G. (1994) The Impact of AFDC on young Women's childbearing decisions. Washington, D.C.: Urban Institute.

Acs, G. (1996) 'The impact of welfare on young mothers' subsequent childbearing decisions', *The Journal of Human Resources*, 31(4), pp. 898-915.

Acs, G. and Maag, E. (2005) Irreconcilable Differences? The conflict between marriage promotion initiatives for cohabiting couples with children and marriage penalties in tax and transfer programs, New Federalism National Survey of America's Families, Series B No. B-66, Washington: The Urban Institute.

Acs, G. and Nelson, S. (2004) 'Changes in living arrangements during the late 1990s: Do welfare policies matter?', *Journal of Policy Analysis and Management*, 23(2), pp.273-90.

Aizer, A. and McLanahan, S. (2005) *The Impact of Child Support Enforcement on Fertility, Parental Investment and Child Well Being, Cambridge*, MA: National Bureau of Economic Research.

Alm, J., Dickert-Conlin, S. and Whittington, L. (1999) 'Policy Watch The Marriage Penalty', *Journal of Economic Perspectives*, 13(3), pp 193-204.

An, C-B., Haveman, R. and Wolfe, B. (1993) 'Teen out-of-wedlock births and welfare receipt: The role of childhood events and economic circumstances', The Review of Economics and Statistics, 75(2), pp. 195-208.

Anderberg, D. (2007) *Tax Credits, Income Support and Partnership Decisions, CESifo Working Paper No. 2147*, CESSifo, Germany: Munich.

Anderberg D., Kondylis, F. and Walker, I. (2008) 'Partnership Penalties and Bonuses Created by UK Benefit Programs', CESifo Economic Studies, 54(1), pp. 1–21.

Arnato, P. (2007) 'Strengthening marriage is an appropriate social policy goal', Journal of Policy Analysis and Management, 26(4), pp. 952-956.

Bakeo, A. and Clarke, L. (2004) *The Health of Children and Young People*, National Statistics.

Baker, M., Hanna, E. and Kantarevic, J. (2003) *The Married Widow: Marriage Penalties Matter!*, *NBER Working Paper 9782*, Cambridge MA.: National Bureau of Economic Research.

Barnes, M., Lyon, N., Morris, S., Robinson, V. and Wan Yan, Y. (2005) *Family Life in Britain: Findings from the 2003 Families and Children Study (FACS)*, DWP Research Report No. 250, Leeds: CDS.

Baughman, R. and Dickert-Conlin, S. (2007) 'The earned income tax credit and fertility', *Journal of Population Economics*. Retrieved from http://www.springerlink.com/content/9590520574802806/fulltext.pdf on 5 May 2008.

Baughman, R. and Dickert-Conlin, S. (2003) 'Did expanding the ETIC promote motherhood?', The American Economic Review, 93(2), pp.247-251.

Becker, G. (1981) *A Treatise on the Family*, Cambridge, MA: Harvard University Press.

Beighton, L. and Draper, D. (2007) Supporting marriage and encouraging couples to come together: Tax and tax credit options, CARE policy paper, London: CARE.

Bergsttrom, T. (1996) 'Economics in a family way,' *Journal of Economic Literature*, 34(4), pp. 1903-1934.

Bitler, M., Gelbach, J., Hoynes, H. (2006) 'Welfare reform and children's living arrangements', *Journal of Human Resources*, 41(1), 1-27.

Bitler, M., Gelbach, J., Hoynes, H. and Zavodny, M. (2004) 'The impact of welfare reform on marriage and divorce', Demography, 41(2), pp. 213-236.

Blackburn, McK. (2000) 'Welfare effects on the marital decisions of never-married mothers', *Journal of Human Resources*, 35(1), pp. 116-142.

Blank, R. (2002) 'Evaluating welfare reform in the United States', *Journal of Economic Literature*, XL, pp. 1105-1166.

Blau, F., Kahn, L. and Waldfogel, J. (2004) 'The impact of welfare benefits on single motherhood and headship of young women. Evidence from the census,' Journal of Human Resources, 39(2), pp. 382-404.

Blaug, M. (1978) *Economic theory in retrospect*, Cambridge: Cambridge University Press.

Blekesaune, M. (2008) *Unemployment and partnership dissolution*, ISER Working Paper 2008-21, Colchester: University of Essex.

Blundell, R. and Walker, I. (2001) *Working Families' Tax Credit: A review of the evidence, issues and prospects for further research*, Inland Revenue Research Report 1, Retrieved from http://www.hmrc.gov.uk/research/wftc6rb.pdf on 19th March 2008.

Böheim, R. and Ermisch, J. (1999) *Breaking up: Financial Surprises and Partnership Dissolution*, ISER Working Paper 1999-09, Colchester: University of Essex.

Bradbury, B. (2006) *The Impact of Young Motherhood on Education, Employment and Marriage*, SPRC Discussion Paper No. 148, Sydney: Social Policy Research Centre.

Bradbury, B. and Norris, K. (2005) 'Income and separation', Journal of Sociology, 41(4), pp.425-446.

Brewer, M. (2007) 'Supporting couples with children through the tax system', in R. Chote, C. Emmerson, A. Leicester and D. Miles (eds.) The Institute for Fiscal Studies Green Budget 2007, London: Institute for Fiscal Studies.

Brewer, M., Ratcliffe, A. and Smith, S. (2007) *Does Welfare Reform Affect Fertility? Evidence from the UK*, CMPO Working Paper No. 07/177, Bristol: Centre for Market and Public Organisation.

Brewer, M. and Shaw, J. (2006) *How many lone parents are receiving tax credits?*, IFS Briefing Note 70, London: IFS.

Bumpass, L., Sweet, J. and Cherlin, A. (1990) 'The role of cohabitation in declining rates of marriage', Journal of Marriage and the Family, 53, pp. 913-927.

Burstein, N. (2007) 'Economic influences on marriage and divorce', Journal of Policy Analysis and Management, 26(2), pp. 387-429.

Buffeteau, S. and Échevin, D. (2008) *Taxation and marriage: Evidence from a natural experiment in France*, Groupe de Recherche en Économie et Développement International, Working Paper 08-01, Québec: Université de Sherbrooke.

Campbell, M. and Roberts, C. (2002) 'Child Tax Credit and the Family Comparisons', Family Policy Briefing 2, Oxford: University of Oxford Department of Social Policy and Social Work.

Camasso, M. (2004) 'Isolating the family cap effect on fertility behaviour: Evidence from New Jersey's Family Development Program experiment', Contemporary Economic Policy, 22(4), pp. 453-67.

Camasso, M., Harvey, C., Jagaqnnathan, R. and Killingsworth R. (2003) 'The use of client surveys to gauge the threat of contamination in welfare reform experiments', Journal of Public Analysis and Management 22(2), pp. 207-223.

Camasso, M., Jagaqnnathan, R. and Killingsworth R. (1998) 'New Jersey's Family Development Program, results on program impacts, experimental-control group analysis', Rutgers University, mimeo.

Carasso, A. and Steuerle, C. E. (2005) 'The Hefty Penalty on Marriage Facing Many Households with Children', The Future of Children, 15(2):157-175. Retrieved from http://www.futureofchildren.org/usr\_doc/09\_FOC\_15-2\_fall05\_Carasso-Steuerle.pdf on 1st May 2008.

Civitas (2002) The Lone-Parent Trap: How the Welfare System Discourages Marriage, London: Civitas.

Clarke, H. and McKay, S. (2008) *Exploring disability, family formation and break-up: Reviewing the evidence*, DWP Research Report No. 514, Leeds: CDS.

Clarke, G. and Strauss, R. (1998) 'Children as income-producing assets: The case of teen illegitimacy and government transfers', Southern Economic Journal, 64(4), pp. 827-56.

Corcoran, M. and Kunz, J. (1997) 'Do unmarried births among African-American teens lead to adult poverty?', Social Science Review, 71 (2), pp. 275–289.

Danziger, S., Jakubson, G., Schwartz, S. and Smolensky, E. (1982) 'Work and Welfare as Determinants of Female Poverty and Female Headship.' Quarterly Journal of Economics 97(3):519-34.

Del Bono, E. (2004) *Pre-marital fertility and labour market opportunities: evidence from the 1970 British Cohort Study*, IZA Discussion Paper No 1320, Bonn: Institute for the Study of Labor.

Dickert-Conlin, S. (1999) 'Taxes and Transfers: Their Effect on the Decision to End a Marriage', Journal of Public Economics, 73(2), pp.217-240.

Dickert-Conlin, S. and Houser, S. (1998) 'Taxes and Transfers: A New Look at the Marriage Penalty', National Tax Journal, 51(2), pp. 175-217.

Dickert-Conlin, S. and Houser, S. (1999) *ETIC, AFDC, and the Female Headship Decision*, Institute for Research on Poverty Discussion Paper 1192-99, Wisconsin: Institute for Research on Poverty, University of Wisconsin-Madison.

Dickert-Conlin, S. and Houser, S. (2002) 'EITC and Marriage', *National Tax Journal*, 55(1), pp. 25-40.

Dixon, M. and Margo, J. (2006) *Population Politics*, London: IPPR.

Draper, D. (2005) 'Tax credits: the case for structural reform', IFS conference 'Where Now for Tax Credit Policy?, Retrieved from www.ifs.org.uk/docs/draper\_taxcredits05.ppt on 1st October 2008.

Draper, D. (2008) Couple Penalty 2007/08, CARE Research Paper, London: CARE.

Dyer, W. and Failie, R. (2004) 'Do family caps reduce out-of-wedlock births? Evidence from Arkansas, Georgia, Indiana, New Jersey and Virginia,' Population Research and Policy Review, 23, pp. 441-473.

Duncan, G. and Hoffman, S. (1990) 'Welfare benefits, economic opportunities, and out-of-wedlock births among black teenage girls', Demography, 27(4), pp519-35.

Edin, K. (2002) 'Few good men: Why poor mothers don't remarry', The American Prospect, Retrieved from http://www.prospect.org/cs/articles?article=few\_good\_men on 23 July 2008.

Edin, K. and Lein, L. (1997) *Making Ends Meet: How Single Mothers Survive Welfare and Low-Wage Work*, New York: Russell Sage Foundation.

Ehrlich, I. and Kim, J. (2007) Has Social Security Influenced Family Formation and Fertility in OECD Countries? An Economic and Econometric Analysis, NBER Working Paper 12869, Cambridge, MA: National Bureau of Economic Research.

Ellwood, D. (2000a) 'Anti-poverty policy for families in the next century: From welfare to work-and worries', Journal of Economic Perspectives, 14(1), pp. 187-198.

Ellwood, D. (2000b) 'The Impact of the Earned Income Tax Credit and Social Policy Reforms on Work, Marriage, and Living Arrangements,' National Tax Journal, 53(4), Part 2, pp. 1063-1105.

Ellwood, D. and Bane, M. (1985) 'The impact of AFDC on family structure and living arrangement', in R. Ehrenberg (ed.) Research in Labour Economics, 7, pp. 137-207.

Ellwood, D. and Jencks, C. (2004) *The Spread of Single-Parent Families in the United States since 1960*, Faculty Research Working Paper No. RWP04-008, John F. Kennedy School of Government, Harvard University.

Eissa, N. and Hoynes, H. (1999) *Good News for Low-Income Families? Tax-Transfer Schemes, and Marriage*, Unpublished Manuscript, Berkeley, CA: University of Berkeley, California.

Eissa, N. and Hoynes, H. (2000) 'Explaining the fall and rise in the tax cost of marriage: the effect of tax laws and demographic trends, 1984-97', *National Tax Journal*, Part 2, 53(3), pp. 683-712.

Eissa, N. and Hoynes, H. (2000b) *Tax and Transfer Policy and Family Formation: Marriage and Cohabitation*, manuscript. Retrieved from http://www9.georgetown.edu/faculty/noe/cohabitation\_paper.pdf on 6th May 2008.

Ermisch, J. (2000) *Employment Opportunities and Pre-Marital Births in Britain*, ISER Working Paper 2000-26, Colchester: University of Essex.

Ermisch, J. (2002) *Trying Again: Repartnering after Dissolution of a Union*, ISER Working Paper 2002-19, Colchester: University of Essex.

Ermisch, J. and Francesconi, M. (2000) 'Cohabitation in Great Britain: not for long, but here to stay', *Journal of the Royal Statistical Society*, 163, Part 2, pp. 153-171.

Ermisch, J. and Francesconi, M. (2001) 'Family structure and children's achievements', *Journal of Population Economics*, 14(2), pp. 249-270.

Fairlie, R. and London, R. (1997) 'The effect of incremental benefit levels on births to AFDC recipients', *Journal of Policy Analysis and Management*, 16(4), pp. 575-97.

Fein, D. (1999) Will welfare reform influence marriage and fertility? Evidence from the ABC demonstration, Abt Associated Inc.

Fein, D., London, R. and Mauldon, J. (2002) Assessing the Effects, Research Brief Welfare Reform and Family Formation Project No. 1, Bethesda, MD: Abt Associates.

Fitzgerald, J. and Ribar, D. (2004) 'Welfare reform and female headship', Demography, 41(2), pp. 189-212.

Francesconi M. and van der Klaauw, W. (2007) 'The Socioeconomic Consequences of 'In-work' Benefit Reform for British Lone Mothers', *Journal of Human Resources*, 42(1), pp. 1-31.

Francesconi, M., Rainer, H. and van der Klaauw, W. (2007) 'The Effects of In-work Benefit Reform in Britain on Couples: Theory and Evidence', Discussion Paper No. 2980, Berlin: IZA.

Freud, D. (2007) Reducing dependency, increasing opportunity: options for the future of welfare to work, Leeds: CDS.

Gassman-Pines, A. and Yoshikawa, H. (2006) 'Five-year effects of an anti-poverty program on marriage among never-married mothers', *Journal of Policy Analysis and Management*, 25(1), pp. 11-30.

Gauthier, A. and Hatzius, J. (1997) 'Family benefits and fertility: An econometric analysis', Population Studies, 51, pp. 295-306.

Gennetian, L. (2003) The Long-Term Effects of the Minnesota Family Investment Program on Marriage and Divorce Among Two-Parent Families, New York: MDRC.

Gennetian, L. and Knox, V. (2003) *Staying Single: The Effects of Welfare Reform Policies on Marriage and Cohabitation*, The Next Generation Working Paper Series No. 13, New York: MDRC.

Gennetian, L. and Millar, C. (2004) 'How Welfare Can Affect Marriage: Evidence from an Experimental Study in Minnesota,' Review of Economics of the Household, 2, pp. 275-301.

Geronimus, A. (1997) 'Teenage childbearing and personal responsibility', Political Science Quarterly, 112(3), pp. 405-30.

Geronimus, A. and Korenman, S. (1992) 'The socioeconomic consequences of teen childbearing reconsidered', *Quarterly Journal of Economics*, 107, pp. 1187–1214.

González, L. (2005) *The determinants of the prevalence of single mothers: A cross-country analysis*, IZA Discussion Paper No. 1677, Bonn: Institute for the Study of Labor.

González, L. (2007) 'The effect of benefits on single motherhood in Europe', Labour Economics, 14(3), pp. 393-412.

Gough D. (2007) 'Weight of evidence: a framework for the appraisal of the quality and relevance of evidence', in J. Furlong and A. Oancea (Eds.) Applied and Practice-based Research, Special Edition of Research Papers in Education, 22, (2), 213-228. Retrieved from http://eppi.ioe.ac.uk/cms/Portals/0/PDF%20reviews%20and%20 summaries/David%20Gough%20Weight%20of%20Evidence%20paper%20 for%20RPE%2022-1-07%20final.doc on 7th March 2008.

Greenberg, D. and Shroder, M., (1997) *The Digest of Social Experiments*. 2nd ed. Washington, DC: Urban Institute Press.

Hantrais, L. (2004) Family policy matters: responding to family change in Europe, Bristol: The Policy Press.

Harknett, K. (2006) 'Does receiving an earnings supplement affect union formation? Estimating effects for program participants using propensity score matching,' Evaluation Review, 30(6), pp. 741-778.

Harknett, K and Gennetian, L. (2003) 'How an earnings supplement can affect union formation among low-income single mothers', Demography, 40(3), pp. 451-478.

Henshaw, S. (1998) 'Unintended pregnancy in the United States', Family Planning Perspectives 30(1), pp. 24-29. 46.

Henshaw, Sir D. (2006) Report to the Secretary of State for Work and Pensions, Recovering Child Support: Routes to Responsibility, Cm. 6894, London: TSO.

Hoffman, S. and Seidman, L. (2003) *Helping Working Families: The Earned Income Tax Credit*, Kalamazoo, Michigan: W E Upjohn Institute for Employment Research.

Holtzblatt, J. and Rebelein, R. (2000) 'Measuring the Effect of the EITC on Marriage Penalties and Bonuses', *National Tax Journal*, 53(2), pp. 1107-1133.

Horvath-Rose, A. and Peters, H. (2000) *Welfare Waivers and Nonmarital Childbearing, Joint Centre for Poverty Research Working Paper 128*, Chicago: Joint Centre for Poverty Research.

Horvath-Rose, A., Peters, H. and Sabia, J. (2008) 'Capping kids: The family cap and nonmarital childbearing', Population Research and Policy Review, 27(2), pp. 119-138.

Hotz, V. and Scholz, J. (2003) 'The Earned Income Tax Credit' in Moffitt, R. (ed.) Means-Tested Transfer Programs in the United States, Chicago: University of Chicago Press, pp.141-197.

Hoynes, H. (1997a) 'Does welfare play any role in female headship decisions?' *Journal of Public Economics*, 65(2), pp. 89-117.

Hoynes, H. (1997b) 'Work, Welfare, and Family Structure: What Have We Learned?', in Fiscal Policy: Lessons From Economic Research, A. Auerbach (ed.), Cambridge, Mass: MIT Press, pp. 101-146.

Hu, W. (2003) 'Marriage and Economic Incentives: Evidence from a Welfare Experiment', *Journal of Human Resources*, Vol. 38(4), pp. 942-963.

Jagaqnnathan, R. and Camasso, M. (2003) 'Family cap and nonmarital fertility: The racial conditioning of policy effects', *Journal of Marriage and the Family*, 65(1), pp. 52-71.

Joshi, H., Cooksey, E., Wiggins, R., McCulloch, A., Verropoulou, G. and Clarke, L. (1999) 'Diverse Family Living Situations and Child Development: a multi-level analysis comparing longitudinal evidence from Britain and the United States', *International Journal of Law, Policy and the Family*, 13(3), pp. 292-314.

Joyce, T., Kaestner, R., Korenman, S. and Henshaw, S. (2004) 'Family cap provisions and changes in births and abortions', Population Research and Policy Review, 23, pp. 475-511.

Kiernan, K. (1999) 'Cohabitation in Western Europe', Population Trends 96, London: Office for National Statistics, pp. 25-32.

Kearney, M. (2004) 'Is there an effect of incremental welfare benefits on fertility behavior? A look at the family cap', *The Journal of Human Resources*, 39(2), pp. 295-325.

Klawitter, M., Plotnick, R. and Edwards, M. (2000) 'Determinants of initial entry onto welfare by young women', *Journal of Policy Analysis and Management*, 19(4), pp. 527-546.

Lam, D. (1988) 'Marriage markets and assortative mating with household public goods: Theoretical results and empirical implications', *Journal of Human Resources*, 23(4), pp. 462-487.

Levin-Epstein, J. (2003) Lifting the lid off the family cap: States revisit problematic policy for welfare mothers, Childbearing and Reproductive Health Series Brief No 1, Washington D.C.: Centre for Law and Social Policy.

Lewis, M. (2007) States of Reason Freedom, responsibility and the governing of behaviour change Executive Summary, London: IPPR.

Lichter, D., McLaughlin, D. and Ribar, D. (1997) 'Welfare and the rise in female-headed families', The American Journal of Sociology, 103(1), pp.112-143.

Lopoo, L. and DeLeire, T. (2006) 'Did Welfare Reform Influence the Fertility of Young Teens?', *Journal of Policy Analysis and Management*, 25(2), pp. 275–298.

Loury, G. (2000) 'Preventing Subsequent Births to Welfare Recipients', in Besharov, D. and Germanis, P (eds.), Preventing Subsequent Births to Welfare Recipients, College Park, Md, School of Public Affairs.

Lundberg, S. and Plotnick, R. (1990) 'Effects of State welfare, abortion and family planning policies on premarital childbearing among white adolescents', Family Planning Perspectives, 22(6), pp.246-75.

Lundberg, S. and Plotnick, R. (1995) 'Adolescent premarital childbearing: Do economic incentives matter?', Journal of Labor Economics, 13(2), pp. 177-200.

Malthus, T. (1798) *An essay on population*. Reprinted as An essay on population: Introduction by Fogarty, M., London: J.M. Dent, 1958.

Marsh, A. and Vegeris, S. (2004) *The British Lone Parent Cohort and their Children* 1991 to 2001, DWP Research Report No. 209, Leeds: CDS.

Martin, J., Hamilton, B., Sutton, P., Ventura, S., Menacher, F. and Munson, M. (2003) *Births: Final data for 2002. National vital statistics reports*, 52(10), Hyattsville, MD: National Centre for Health Statistics.

Maynard, R., Boehnen, E., Corbett, T. and Sandefur, G. with Mosley, J. (1998) 'Changing Family Formation Behavior Through Welfare Reform', in R. Moffitt (ed.) Welfare, the Family, and Reproductive Behavior, Washington: National Academy Press, pp.134-176.

Mauldon, J., London, R., Fein, D. and Bliss, S. (2002) What Do They Think? Welfare Recipients' Attitudes Towards Marriage and Childbearing, Research Brief Welfare Reform and Family Formation Project No. 2, Bethesda, MD: Abt Associates.

McConnell, H. and Wilson, B. (2007) 'Families', in S. Smallwood and B. Wilson, (eds.) Focus on Families, Basingstoke: Palgrave Macmillan, pp.2-17.

McLanahan, S. and Sandefur, G. (1994) *Growing Up with a Single Parent: What Hurts, What Helps*, Cambridge, MA: Harvard University Press.

Mellor, J. (1998) 'The Effect of Family Planning Programs on the Fertility of Welfare Recipients'. *Journal of Human Resources*, 33, 866-95.

Meyer, B. and Rosenbaum, D. (2000) 'Making single mothers work: Recent tax and welfare policy and its effects', *National Tax Journal*, 53(4), Part 2, pp.1027-1062.

Midgley, J. with Stewart, K., Piachaud, D. and Glennerster, H. (2008) *Welfare Reform in the United States: Implications for British Social Policy*, CASE Paper 131, London: Centre for Analysis of Social Exclusion, LSE.

Moffitt, R. (1990) 'The effect of the U.S. welfare system on marital status', *Journal of Public Economics*, 41, pp.101-24.

Moffitt, R. (1992) 'Incentive effects of the US welfare system', *Journal of Economic Literature*, 30, pp. 1-61.

Moffitt, R. (1994) 'Welfare effects on female headship with area effects', The Journal of Human Resources, 29(2), Special Issue: Women's Work, Wages, and Well-Being, pp. 621-636.

Moffitt, R. (1998a) 'Introduction', in R. Moffitt (ed.) Welfare, the Family, and Reproductive Behavior, Washington: National Academy Press, pp.1-8.

Moffitt, R. (1998b) 'The Effect of Welfare on Marriage and Fertility', in R. Moffitt (ed.) Welfare, the Family, and Reproductive Behavior, Washington: National Academy Press, pp.50-97.

Moffitt, R. (2000) 'Welfare Benefits and Female Headship in U.S. Time Series' The American Economic Review, 90(2), pp. 373 -377.

Moffitt, R. (2003) 'The Temporary Assistance for Needy Families Program', in R. Moffitt, R. (ed.), Means-Tested Transfer Programs in the United States, Chicago: University of Chicago Press, pp. 291-363.

Moffitt, R., Reville, R. and Winkler, A. (1998) 'Beyond single mothers: Cohabitation and marriage in the AFDC program', Demography, 35(3), pp. 259-278.

Moffitt, R., and Ver Ploeg, M., (eds.) (2001) *Evaluating Welfare Reform: A Framework and Review of Current Work*, Washington DC: National academy Press.

Morgan, P. (2007) *The War between the State and the Family*, Hobart Paper 159, London: Institute of Economic Affairs.

Murray, C. (1984) Losing Ground, New York: Basic Books.

Nandi, A. (2008) Women's Economic Gains from Employment, Marriage and Cohabitation, ISER Working Paper 2008-21, Colchester: University of Essex.

O'Donoghue, C. and Sutherland, H. (1999) 'Accounting for the family in European income tax systems', Cambridge Journal of Economics, 23, pp.565-598.

Ono, H. (1998) 'Husbands' and Wives' Resources and Marital Dissolution', Journal of Marriage and the Family, 60(3), pp. 674-689.

Ono, H. (2001) Women's Economic Resources, Marriage, and Contexts of Gender: The Case of Japan, U.S. and Sweden, Paper presented at the 2001 annual meeting of the Population Association of America, 29-31 March, Washington, DC.

Oppenheimer, V. (1997) 'Women's employment and the gain to marriage: The specialization and trading model', Annual Review of Sociology, 23, pp. 431-453.

Pawson, R. (2006) Evidence-based Policy A realist perspective, London: Sage.

Plotnick, R. (2004) *Teenage Expectations and Desires about Family Formation in the United States*, CASE paper 90, London: CASE, LSE.

Plotnick, R., Garfinkel, I., Mclanahan, S. and Ku, I. (2006) 'The impact of child support enforcement policy on nonmarital childbearing', *Journal of Policy Analysis and Management*, 26(1), pp. 79-98.

Popay, J., Roberts, H., Petticrew, M., Arai, L., Rodgers, M., Britten, N. with Roen, K. and Duffy, S. (2006) *Guidance on the Conduct of Narrative Synthesis in Systematic Reviews*.

Presser, H. and Salsberg, L. (1975) 'Public assistance and early family formation is there a pronatalist effect', Social Problems, 23(2), pp. 226-241.

Rainer, H. (2007) 'Should we write prenuptial contracts?', European Economic Review, 51, pp. 337-363.

Ribar, D. (2003) What Do Social Scientists Know about the Benefits of Marriage? A Review of Quantitative Methodologies, Department of Economics, Washington: The George Washington University.

Ridge M., O'Flaherty, D. and Deasley, S. (2007) *Child support and work incentives:* A literature review, DWP Research Report No 402, Leeds: CDS

Roberts, P. (2007) 'Out of Order? Factors influencing the sequence of marriage and childbirth among disadvantaged Americans', CLASP Policy Brief, Couples and Marriage Series No. 9, Washington: Center for Law and Social Policy.

Roberts, P. (2008) 'The Implications of Multiple Fertility for Efforts to Promote Marriage in Programs Serving Low-Income Mothers and Fathers', CLASP Policy Brief, Couples and Marriage Series No. 11, Washington: Center for Law and Social Policy.

Rodgers, B. and Pryor, J. (1998) *Divorce and Separation: the outcomes for children*, York: JRF.

Rosenbaum, D. (2000) *Taxes, the Earned Income Tax Credit, and Marital Status, Joint Center for Poverty Research Working Paper*, Chicago: Northwestern University/ University of Chicago Joint Center for Poverty Research.

Rosenzweig, M. (1999) 'Welfare, marital prospects, and nonmarital childbearing', *Journal of Political Economy*, 107(6), Part 2, pp.S3-S32.

Rossi, P. (2001) 'New Jersey's family development program: an overview and critique of the Rutgers' evaluation', in Besharov, D. and Germanis, P. (eds.), Preventing Subsequent Births to Welfare Recipients, College Park, Md: School of Public Affairs.

Ryan, S., Manlove, J., Hofferth, S. (2006) 'State-level welfare policies and nonmarital subsequent childbearing', Population Research and Policy Review, 25, pp. 103-26.

Sabia, J. (2006) 'Blacks and the family cap: pregnancy, abortion, and spillovers', *Journal of Population Economics*, 21, pp. 111-134.

Salkind, N. and Haskins, R. (1982) 'Negative Income Tax: The Impact on Children from Low-Income Families', *Journal of Family Issues*, Vol. 3, pp. 165-180.

Svarer, M. (2004) 'Is your love in vain? Another look at premarital cohabitation and divorce', *Journal of Human Resources*, 34(2), pp. 523-535.

Schoeni, R. and Blank, R. (2000) What Has Welfare Reform Accomplished? Impacts on Welfare Participation, Employment, Income, Poverty, and Family Structure, NBER Working Paper 7627, Cambridge, MA: National Bureau of Economic Research.

Scholz, J. (1994) 'Tax policy and the working poor: The Earned Income Tax Credit', Focus, Institute for Research on Poverty, University of Wisconsin-Madison, 15(3), pp.1-12.

Schultz, P. (1994) 'Marital status and fertility in the United States. Welfare and labour market effects', *Journal of Human Resources*, 29(2), pp. 637-669.

Sherman, A. (2001) 'How Children Fare in Welfare Experiments: The Pivotal Role of Income', APPAM Fall 2001 Research Conference, Washington DC.

Smock, P., Manning, W. and Porter, M. (2005) '"Everything's There Except Money": How Money Shapes Decisions to Marry Among Cohabitors', *Journal of Marriage and Family*, 67, pp. 680-696.

Stafford, B. (2002) 'Being more certain about Random Assignment in Social Policy Evaluations', Social Policy and Society, Vol. 1, No. 4, pp275-284.

Strickland, P. (1998) Working Families Tax Credit and Family Credit, House of Commons Library Research Paper 98/46, London: House of Commons.

Terry-Humen, E., Manlove, J., and Moore, K. (2001) *Births outside of marriage: Perceptions vs Reality*, Child Trends Research Brief, Washington DC: Child Trends.

US General Accounting Office (2001) Welfare Reform: More Research Needed on TANF Family Caps and Other Policies for Reducing Out-of-Wedlock Births, (GAO-01-924), Washington, DC: GAO.

Ventry, D. (2000) 'The collision of tax and welfare politics: The political history of the Earned Income Tax Credit, 1969-99', *National Tax Journal*, 53(4), Part 2, pp. 983-1026.

Walker, I. and Zhu, Y. (2006) 'Child Support and Partnership Dissolution', *The Economic Journal*, 116, pp.C93-C116.

Weiss, Y. (1997) 'The formation and dissolution of families: Why marry? Who marries whom? And what happens in divorce?', in M. Rosenzweig and O. Stark (eds.), Handbook of Population and Family Economics, Amsterdam: Elserier Science, pp. 81-123.

Wikeley, N., Barnett, S., Brown, J., Davis, G., Diamond, I., Draper, T. and Smith, P. (2001) *National Survey of Child Support Agency Clients*, DWP Research Report No 152, Leeds: CDS.

Williams, F. (2005) Rethinking Families, London: Calouste Gulbenkian Foundation.

Willis, R. (1999) 'A theory of out-of-wedlock childbearing', *Journal of Political Economy*, 107(6), pp.S33-S64.

Wilson, B. and Smallwood, S. (2008) 'Age differences at marriage and divorce', Population Trends 132, London: Office for National Statistics, pp.17-25.

Winkler, A. (1993) 'The living arrangements of single mothers with dependent children: An added perspective', *American Journal of Economics and Sociology*, 52(1), pp.1-18.

Wiseman, M. (1993) 'Welfare Reform in the States: The Bush Legacy.' Focus, 15, pp. 18-36.

Wu, Z. and Balakrishnan, T. (1992) 'Attitudes Towards Cohabitation and Marriage in Canada,' *Journal of Comparative Family Studies*, 21, pp. 1-12.

Zimmerman, S. (1991) 'The Welfare State and Family Breakup: The Mythical Connection', Family Relations, 40(2), pp. 139-147.